Southampton

University of Southampton Research Repository

Copyright © and Moral Rights for this thesis and, where applicable, any accompanying data are retained by the author and/or other copyright owners. A copy can be downloaded for personal non-commercial research or study, without prior permission or charge. This thesis and the accompanying data cannot be reproduced or quoted extensively from without first obtaining permission in writing from the copyright holder/s. The content of the thesis and accompanying research data (where applicable) must not be changed in any way or sold commercially in any format or medium without the formal permission of the copyright holder/s.

When referring to this thesis and any accompanying data, full bibliographic details must be given, e.g.

Thesis: Katarzyna Monika Grabska (2020) " *Silicon-germanium by Rapid Melt Growth for the Silicon-On-Insulator Platform* ", University of Southampton, Optoelectronics Research Centre, PhD Thesis, pagination.

Data: Katarzyna Monika Grabska (2020) " Silicon-germanium by Rapid Melt Growth for the Silicon-On-Insulator Platform ". URI [dataset]

University of Southampton

Faculty of Engineering and Physical Sciences
Optoelectronics Research Centre

Silicon-germanium by Rapid Melt Growth for the Silicon-On-Insulator Platform

by

Katarzyna Monika Grabska

Thesis for the degree of Doctor of Philosophy

April 2020

University of Southampton

Abstract

Faculty of Engineering and Physical Sciences Optoelectronics Research Centre <u>Thesis for the degree of Doctor of Philosophy</u> Silicon-germanium by Rapid Melt Growth for Silicon-On-Insulator Platform

by

Katarzyna Monika Grabska

Silicon-germanium (Si_{1-x}Ge_x) and Ge are widely used in optoelectronics. It is fully miscible across the whole composition range, which allows for the whole spectrum of the band gap to be tuned between the values of pure Si and pure Ge. This versatility is extremely important for photonics applications such as electro-absorption modulators and photodetectors, where the operating wavelength is strongly dependent on the Si content. Traditional SiGe growth allows only a single SiGe composition to be grown across the wafer. On the contrary, Rapid Melt Growth (RMG) offers the opportunity of multiple compositions of SiGe by using just a single Ge deposition step and a single anneal step, as previously demonstrated on the silicon platform. This is made possible by engineering the structural parameters of the RMG structures to enable multiple uniform composition strips of SiGe across the wafer, each with a different, tuneable composition. Furthermore, RMG can be used for the growth of high-quality SiGe material with a smaller number of defects than other growth techniques, which can lead to the fabrication of higher performance active devices. Finally, the simplicity of the fabrication process provides a cost-effective growth method making this concept very attractive.

In this thesis, for the first time, RMG processing was successfully used to demonstrate constant composition SiGe and Ge structures (in range between 65 and 100% Ge) embedded within Silicon-on-Insulator (SOI) substrates. The structures embedded in the SOI platform enable integration to waveguides for the creation of both passive and active waveguide-based devices to form complex Photonic Integrated Circuits (PIC). This first demonstration promises to allow the development of new devices and associated applications with high performance and low cost. The findings of the research will lead to waveguide integration and work on active devices.

Table of Contents

Table o	of Co	ntentsi
Table o	of Tal	bles v
Table o	of Fig	uresvii
Researc	ch Tl	hesis: Declaration of Authorshipxvii
Acknow	vled	gements xxiii
Chapte	r 1	Introduction1
1.1	Silico	on Photonics2
1.2	Silico	on germanium applications3
1.3	Rapi	d Melt Growth for silicon germanium and germanium on Silicon-on-insulator
	plati	form4
1.4	Thes	sis outline5
Chapte	r 2	Theoretical background and literature review7
2.1	Prop	perties of Silicon, Germanium and $Si_{1-x}Ge_x$ alloys7
2.:	1.1	Energy band gap8
2.:	1.2	Lattice constant
2.2	Ge a	nd Si _{1-x} Ge _x growth on Si13
2.3	Ge a	nd Si _{1-x} Ge _x on Si growth techniques16
2.3	3.1	Molecular beam epitaxy17
2.3	3.2	Chemical Vapour Deposition17
2.3	3.3	Rapid Melt Growth
2.4. H	ydro	gen within Ge structures24
2.4	Surf	ace tension25
2.5	Sum	mary28
Chapte	r 3	Methods 29
3.1	Fabr	ication29
3.:	1.1	Deposition techniques29
	3	.1.1.1 Chemical Vapour Deposition29

	3	3.1.1.2 Thermal oxidation	
	3.1.2	Lithography techniques	34
	3	3.1.2.1 Optical lithography	
	3	3.1.2.2 Deep ultraviolet lithography	
	3	3.1.2.3 Electron beam lithography	39
	3.1.3	Etch techniques	40
	3	3.1.3.1 Dry etch	40
	3	3.1.3.2 Wet etch	41
	3.1.4	Chemical Mechanical Planarization (CMP)	
	3.1.5	Rapid Thermal Annealing	46
3.2	Cha	racterisation	48
	3.2.1	Material thickness and profile	48
	3.2.2	Profilometry and Atomic Force Microscopy	<u>50</u> 49
	3.2.3	Raman Spectroscopy	50
	3.2.4	Scanning Electron Microscopy and Focused Ion Beam	54
3.3	Sum	nmary	<u>56</u> 55
Chap	oter 4	Design of mask and experiments	57
4.1	Mot	tivation for SOI integration	57
4.2	Pro	cess flow	59
4.3	Mas	sk design	60
4.4	Sum	ımary	66
Chap	oter 5	Fabrication of SiGe wires on SOI platform	67
5.1	Forr	mation of angled Si cavities	67
	5.1.1	Dry etch	<u>69</u> 68
	5.1.2	Wet etch test	69
	5.1.3	Wet etch tests for tree-like structures	70
5.2	CMI	Ρ	75
	5.2.1	Batch 1 – process developments for 6-inch Si wafers	76
	5.2.2	Batch 2 – process developments for 6-inch wafers	

Table of Contents

5.3	Lith	ography	84
5.4	Pro	cess development summary	86
5.5	Con	stant composition SiGe/Ge wires on SOI platform	87
	5.5.1	Fabrication	87
	5.5.2	Wires with graded composition on SOI	89
	5.5.3	Branch parameters for RMG structures embedded into SOI	90
	5.5.4	Seed connection for RMG structures embedded into SOI	95
	5.5.5	Composition tuning for wavelength division multiplexing (WDM)	on SOI
		platform	<u>96</u> 95
5.6	Sum	imary	<u>102</u> 101
Chap	oter 6	Conclusions and Future Work	
6.1	Con	clusions	<u>103</u> 102
6.2	Futu	ire work	<u>104</u> 103
Арр	endix A	\ [Full process flow]	107
List	of Refe	rences	109

Table of Tables

Table of Tables

Table 1	Important properties of Si and Ge reproduced from [3]8
Table 2	The Ge content x in $Si_{1-x}Ge_x$ alloys and corresponding values of the lattice parameter [17,18]. A proportional relationship as predicted from Vegard's rule is observed for the whole range of x values
Table 3	TMAH properties for a wet etch reproduced from [24] <u>44</u> 43
Table 4	The penetration depth of light of various wavelengths into Si and Ge [4][59].
Table 5	Spot size radius for a choice of lasers and magnifications [4]52
Table 6	Parameters of sets for waveguide integration64
Table 7	The parameters for wafers from the development Batch 2. Wafer SOI1 was broken during CMP
Table 8	Results for wafer Si3 (Si test substrate) with an insulating SiO_2 layer surrounding tree-like structures, before and after CMP. This data provides information about
	PECVD SIO ₂ uniformity across the wafer and minimised wafer bow after processing
	r

v

Figure 1	The visualisation of the Moore's Law being an observation about the number of
	transistors across the integrated circuit (IC) from [3]1
Figure 2	Si and Ge are group IV elements with atomic numbers of 14 and 32 respectively, with 4 valence electrons7
Figure 3	The band diagram for semiconductors: (a) direct band gap semiconductor diagram with the conduction minimum aligned in <i>k</i> -space with the valence band maximum, (b) indirect band gap semiconductor diagram with conduction band minimum shifted in <i>k</i> -space with respect to the valence band maximum, (c) Ge band diagram; reproduced from [6]9
Figure 4	Absorption coefficient and band gap of different semiconductors for a range of wave- lengths [8]. This plot is revealing the nature of their band-structure and band gap energy
Figure 5	The band gap for the spectrum of $Si_{1,x}Ge_x$ alloys based on calculations expressed in equations 1 and 2 [12]11
Figure 6	The diagram of fcc crystal structure, where the lattice points are located in the corners of a cube and in the centre of each face; a is a lattice constant12
Figure 7	The Brillouin zone for the fcc unit cell with the characteristic symmetry points and their wave vectors [15]12
Figure 8	Basic modes of film growth: (a) Volmer–Weber mode; (b) Frank–van-der Merwe mode; (c) Stranski–Krastanov mode14
Figure 9	Transmission electron microscope cross-sections of Ge-on-Si layers for Ge epitaxial growth at different temperatures, Stransky–Krastanov growth of Ge-on-Si in an ultra-high vacuum CVD reactor at 550°C and 350°C (a, b respectively), reproduced from [24]. The thickness of the Ge layer shown here is 60 nm15
Figure 10	Cross-section atomic view of stages of the Ge-on-Si growth; a) the wetting layer, b) growth below the critical thickness, c) thickness over the critical thickness when dislocations and other defects occur
Figure 11	The phase diagram of the SiGe alloys, where L stands for liquid, crosses and circles correspond to different data sets used to construct the diagram. The top line of

- Figure 14 (a) Schematic structure and (b) optical micrograph of the sample [substrate: Si(100)] after RTA (1000° C, 1 s) reproduced from [47]......21
- Figure 16 Dependence of the SiGe composition along strips for changeable cooling rate reproduced from [50]......22
- Figure 17 The comparison of SiGe composition profiles after RMG at a range of anneal times, reproduced from [41]......23

Figure 18 Tailored RMG structure on Si platform, reproduced from [1].23

- Figure 19 The positions of the hydrogen within the fcc unit cell, a) H₂ dimer rotating inside the diamond lattice, b) H*2 defect with one hydrogen hidden within the structure, c) proposed structure for quadrihydrogen-vacancy (VH4) complex, d) proposed structure for hydrogen-saturated interstitial (IH2) complex, reproduced from [14].

Figure 22	Comparison of Ge strips shapes before and after the RMG process reproduced from [59]. The effect of lateral expansion along free edges and surface tension were investigated for a variety of strip widths
Figure 23	A diagram of a PECVD chamber with a view of the gas system, location of the RF power source and the location of the wafer. Reproduced from [9]
Figure 24	A diagram of an LEPECVD chamber with the argon plasma distribution controlled by the anode plate and the wafer's location. Reproduced from [10]32
Figure 25	SiO_2 layer thermally grown on the Si substrate against the Si level
Figure 26	The diagram of the system used for the thermal oxidation [17]
Figure 27	The correlation between resolution of the lithography and performance of the resulting microprocessors [18]
Figure 28	Lithography process for positive and negative type of photoresists. 1) The substrate (e.g. Si wafer) with photoresist on the top, 2) photomask placed above the substrate covered with photoresist and affected by the UV lamp, 3a) development of structures covered with negative resist, 4a) etching for structures covered with negative resist removal, 3b) development of structures covered with positive resist, 4b) etching with positive resist 5b) positive resist removal
Figure 29	Principle of the projection lithography [26]
Figure 30	Etch diagrams: a) angled etch, b) isotropic dry etch, c) anisotropic etch. Grey colour corresponds to the substrate e.g. Si, whereas blue colour corresponds to resist or hard mask
Figure 31	A schematic diagram of a standard ICP system used for dry etch [23]41
Figure 32	The adhesion of the resist is important for the profile of the etch and it has significant impact on further processing and performance of resulting devices [24]42
Figure 33	TMAH structure43
Figure 34	The diagram of directional etch and SEM images of etched structures [27]43
Figure 35	The process of CMP reproduced from [42]

Figure 36	H content of SiH ₄ based PECVD of SiO ₂ as a function of depth from the free surface,
	reproduced from [52]. The measurement was performed with SIMS for various
	anneal temperatures
5' 07	
Figure 37	The ellipsometry measurement, reproduced from [53]
Figure 38	Schematic diagram of a Renishaw Raman microscope [57] <u>51</u> 50
Figure 39	The energy diagram for processes occurring from the interaction of photon with
	a molecule; a) inelastic Stokes Scattering with the final energy level higher than
	the initial one, b) elastic Rayleigh scattering which implies the same energy level
	at the end and the initial level of the interaction, c) inelastic Anti-Stokes Scattering
	with the final energy level lower than the initial one [38,39]51
Figure 40	An example Raman spectrum for a SiGe structure obtained from the RMG
	processing. The asymmetry of the peaks can be minimised by decreasing the
	power of the laser and extending the time of measurement53
Figure 41	A combined SEM-FIB system with a Ga+ source where ETD, Everhart-Thornley
	detector; PtIN, platinum injection needle; SS, sample stage; CDM, continuous
	dynode electron multiplier detector; TLD, through-lens detector [62] <u>55</u> 54
Figure 42	The diagram showing secondary electrons (SE) and Backscattered electrons (BSE)
	used for SEM imaging. K, L1, L2,3 correspond to electron shells around the atomic
	nucleus [63]55
Figure 43	Material and band gap engineering of SiGe to vary the operational wavelengths
	of EAM modulators (images as an example design, based on work from the
	research group). The operational wavelength is dependent on the SiGe
	composition and devices would be incorporated into the structures as shown in
	the diagram. The composition 1 is designed for wavelength 1 based devices etc.
Figure 44	Methods of coupling light from a Si waveguide into a Ge photodetector, a)
	evanescent coupling, b) butt coupling, reproduced from [2]
Figure 45	FDTD simulation of coupling light into Ge photodetector at 1.55 $\mu\text{m},$ (a)
	evanescent coupling, (b) butt coupling, reproduced from [2]

х

- Figure 46
 Process for Rapid Melt Growth (RMG): 1) trench creation, 2) insulator deposition,

 3) seed etch, 4) blanket Ge deposition, 5) planarization of the Ge layer to wafer

 level, 6) deposition of the encapsulation layer, 7) RTA, 8) cap removal.....<u>60</u>59
- Figure 47Main types of RMG structures. (a) type 1 RMG for either single or multi- tree
structure with the waveguide space designed for main strips of each tree, (b) RMG
type 2 for test structures. Description of symbols used to describe design: SL- Seed
length, SW- seed width, SC- seed connection with the structure, L- length of the
tree-like structure, w- width of the main trench of the tree structure (with
branches), Bs- distance between branches, Bw- width of the branches, BL- length
of branches, D- distance between seed and the tree for type 1 RMG structures,
D1- distance between the strip of the tree- like structure and the end of the main
strip of structure, wa- width of the main strip for type 1 RMG, D2- distance
between tree-like structures (branches) for multiple tree structures.......6160
- Figure 48 The 6 inch mask template a) the whole template suitable for processing combining optical lithography and e-beam lithography for 34 cells (typical size for e-beam cell), b) 6 typical positions of e-beam alignment marks for a single cell, c) detailed e-beam alignment marks with x and y Vernier scales to measure alignment errors for multi lithography processing. Reproduced from [4]..<u>62</u>61

- Figure 51
 Example structures type RMG1 from set 8; (a) Ge composition vs. strips length/

 Ge composition vs. normalised length, (b)varying width of the main strip.
 6665
- Figure 52 Example type 1 RMG structures (number of trees, n=2) from set 12. Structures were designed to confirm e.g. an impact of the seed connection on the SiGe composition of main strip, the impact of the width on the composition and to confirm the importance of length of the cooling branches of tree-like structures.

Figure 53	A cross-section of a Si trench filled with Ge showing a lack of material on the sidewall
Figure 54	Dry etched Si sidewalls covered with Ge layer; (a) the slightly wider angle (approx.
	5 ° from vertical) and shallower trench compared with (b), where the difference
	between filling of the corners and vertical/ horizontal part of the trench is more
	visible
Figure 55	SEM images of the TMAH angled etched trenches filled with Ge; (a) 30 min TMAH
-	etch, (b) 45 min TMAH etch, (c) 60 min TMAH etch
Figure 56	Tree-like structures obtained during process development for batch 1- after the
	RTA step, leading to identifications of possible issues linked to either
	photolithography or/ and, described here, trench etching; (a) the rounded shape
	of not entirely filled trench, (b) structural defects visible after SiGe melt- yellow
	arrows point the Ge and Si intermixing region due to non-continuous insulating
	layer between Si and Ge, whereas blue arrows indicate issues associated to
	photolithography and trench etching71
Figure 57	The shape of transhes for various waters from Ratch 2: (1) waters SO(1, SO(2, (2)))
ligure 57	une shape of trenches for various waters from batch 2, (1) waters 5011, 5012, (2)
Figure 58	Comparison of main strip and associated branches for after the CMP step for
	process (1) SOI2, (2) SOI3 and (3) SOI4, (3a) explanation of the undercut effect
	occurring in the corner of structure (3)
Figure 59	Measurements of the diagonal between opposite branches compared with
0	minimum distance between these branches indicates significant difference and
	room for possible improvement of the final shape. Structure A is fabricated as
	mentioned described in process (3). Structure B is fabricated as mentioned
	described in process (2)
	described in process (2)
Figure 60	A cross-section of the structures fabricated with ICP/TMAH etch; a) with the 5 μm
	main strip from the Wafer SOI3, done after RMG step, b) a cross-section of the
	2.5 μm strip from Wafer Si575
Figure 61	The CMP aim for batch 1; a) the substrate covered with Ge before CMP, a
	difference be-tween trenches and surrounding layer is clearly visible, b) the
	substrate planarized to the level of embedded structures

Figure 62	The Si W3 wafer after CMP; (a) the wafer after CMP, (b) a map of the wafer with the number assigned to all cells, (c) Chip 7, (d) Chip 14, (e) Chip 28
Figure 63	The CMP process for batch 2; a) a substrate covered with Ge before CMP with clearly visible difference between levels of trenches and surrounding layer, b) the substrate after an additional photolithography and subsequently coming ICP etch to remove Ge between the trenches, c) the substrate planarized to the level of the embedded structures
Figure 64	The mask template with the location of the cells used to control the CMP process.
Figure 65	An optical microscope image of the surface of the wafer SOI 3 with tree like structures and dummy squares after CMP83
Figure 66	The layout for RMG structures; a) Mask 1 and tree-like structures and dummy squares, b) Mask 2 with seeds and another layer (to etch SiO ₂) of dummy squares, c) the final structure after the photolithography 1 and the photolithography 2.
Figure 67	Lithography mask for process (a) is designed for tailored structures to be etched into the Si layer along with dummy squares during Photolithography 1. Lithography mask (b) is used after filling trenches with Ge/SiGe and etching seeds, the structures should be planarized as described in Section 5.2. The negative resist used for this design assures that all surrounding areas are etched, and that the trenches and dummies are protected from the etch85
Figure 68	The photolithography ((a) for 1.8 sec exposure time, 40 sec development) results in rounded corners of the design with straight angles, whereas the structure obtained by e-beam (b) is in good agreement with designed structures86
Figure 69	The diagram of (A) RMG1 type structure with selected seed region and branch length, BL, and (B) RMG2 with selected branch length and distance from the seed to the strip with constant composition, D
Figure 70	The Ge composition for strips embedded in the SOI platform, (A) a graph for 3 μ m wide strips for a variety of lengths, (B) comparison of Ge composition for these strips with normalized length values next to the complete mixing model (referred as a model)

Figure 71	Complete mixing model adapted from [12]90
Figure 72	Simulations of SiGe growth using steady-state solidification model for the fast regrowth front propagation, adapted from [12]. Situations from a) to c) correspond to the increasing regrowth front propagation speed
Figure 73	Dependence of the SiGe composition profile on the branch length for RMG1 type structures for 30 um long strips with a branch length of 5, 15 and 25 μ m for structures A, B, C as shown in graph. The position of the measured strip is highlighted on each image
Figure 74	Dependence of the SiGe composition for single tree like structure (type RMG2) for different length of the cooling branches
Figure 75	Parameters of the RMG2 type structure, where: Bs - Branch Separation, w - width of the strip with constant composition, Bw - Branch width, D - distance from seed to the start of the strip with constant composition, D1 - distance from the start of the strip with constant composition to the end of the main strip, SL - seed length, Sw – seed width, Sc- seed connection, L- length of the strip with constant composition
Figure 76	Composition profile for 2 different Branch Separations (Bs), other parameters are as follows: w= 2μm, Bw= 1 μm, BL= 5 μm for diagram A and BL=25 μm for diagram B
Figure 77	The seed connection (Sc) test, (A), test for RMG2 structures with parameters as following: BL=20μm, w=3μm, Bs=4μm, Bw=1μm, D=25μm, (B) SEM image showing an example test structure set with a variance of the seed connection.
Figure 78	SEM image of structure used for obtaining tuned constant SiGe composition along the main strip. The varied parameter is distance, D1, from the seed to the strip with constant SiGe composition, distance, D, from the seed to the end of structure and Branch length (BL). Intermixing region between Si and Ge in the seed area is pointed on the image
Figure 79	SiGe composition controlled by the distance, D, between the seed and the main strip with attached branches

Figure 80	A diagram of the ratio of normalized length D1/D against Ge composition. The	he
	standard deviation for each set of measurements for each normalised length wa	as
	respectively: 0.007, 0.008 and 0.0199	

Research Thesis: Declaration of Authorship

Research Thesis: Declaration of Authorship

Print name: Katarzyna Grabska

Title of thesis: Silicon-germanium by Rapid Melt Growth for Silicon-On-Insulator Platform

I declare that this thesis and the work presented in it are my own and has been generated by me as the result of my own original research.

I confirm that:

- This work was done wholly or mainly while in candidature for a research degree at this University;
- 2. Where any part of this thesis has previously been submitted for a degree or any other qualification at this University or any other institution, this has been clearly stated;
- 3. Where I have consulted the published work of others, this is always clearly attributed;
- 4. Where I have quoted from the work of others, the source is always given. With the exception of such quotations, this thesis is entirely my own work;
- 5. I have acknowledged all main sources of help;
- 6. Where the thesis is based on work done by myself jointly with others, I have made clear exactly what was done by others and what I have contributed myself;
- 7. Parts of this work have been published as:

Patents

Gardes F.Y., **Grabska K.M**, "Optoelectronic device and method of manufacturing thereof", filed 2019.

GB Publication of United Kingdom Patent Application Number 1709213.1. OPTOELECTRONIC DEVICE AND METHOD OF MANUFACTURING THEREOF (SPAT02). Case Ref 17465/GB

US Publication of United States Patent Application Number 16/619706. OPTOELECTRONIC DEVICE AND METHOD OF MANUFACTURING THEREOF - SPAT002US. Case Ref 17465/US European (EP) Patent Application No: 18730729.3; Your Ref: SPAT002EP (RIS 17465EP); Mewburn Ref: 007594369

Chinese Patent Application Number 201880038141.9. OPTOELECTRONIC DEVICE AND METHOD OF MANUFACTURING THEREOF. Case Ref 17465/CN

Conference Contributions

- Grabska K, Ballabio A, Banakar M, Barzaghi A, Bucio TD, Isella G, Littlejohns C, Nedeljkovic M, Saito S., Gardes F, "Multi composition GeSi tuneable concentration silicon-germanium wire structures for CMOS photonics." IEEE 16th International Conference on Group IV Photonics (GFP), August 2019.
- Bucio TD, de Paoli G, Grabska KM, Ilie ST, Khokhar AZ, Lacava C, Petropoulos P, Skandalos I, Gardes FY. "Mid-index silicon nitride devices for enhanced linear and non-linear photonic functionalities." IEEE 16th International Conference on Group IV Photonics (GFP), August 2019.
- K. M. Grabska, M. Banakar, T. Dominguez Bucio, M. Nedeljkovic, A. Barzaghi, A. Ballabio, G. Isella, C. G. Littlejohns, S. Saito, and F. Y. Gardes, "Silicongermanium alloys for photonics applications", Photonics North, Quebec City, Canada, May 2019.
- L. Mastronardi, M. Banakar, A. Z. Khokhar, N. Hattasan, T. Rutirawut, T. Dominguez Bucio, K. M. Grabska, C. Littlejohns, A. Bazin, G. Z. Mashanovich, F. Y. Gardes, "56 Gbps Si/GeSi integrated EAM," SPIE/COS Photonics Asia 2018, Beijing, China, 11-13 October 2018.
- M. Banakar, L. Mastronardi, A. Z. Khokhar, T. Rutirawut, T. Domínguez Bucio, K. M. Grabska, C. G. Littlejohns, Antoine F. J. Runge, A. Bazin, A. C. Peacock, and F. Y. Gardes, "Composition engineering of GeSi Franz-Keldysh optical modulator", International Conference on Solid State Devices and Materials, Tokyo, Japan, September 2018.
- T Domínguez Bucio, SL Scholl, ST Ilie, C Lacava, K Debnath, AZ Khokhar, M Banakar, M Sotto, KM Grabska, M Clementi, D Bajoni, M Galli, S Saito, P Petropoulos, FY Gardes, "Low-Temperature NH3-Free Silicon Nitride Platforms for Integrated Photonics," 2018 IEEE 15th International Conference on Group IV Photonics (GFP), Cancun, 2018, pp. 1-2.
- L. Mastronardi, M. Banakar, A. Z. Khokhar, N. Hattasan, T. Rutirawut, T. Domínguez Bucio, K. M. Grabska, C. G. Littlejohns, A. Bazin, G. Z. Mashanovich,

and F. Y. Gardes, "56 Gb/s GeSi optical modulator", Conference on Lasers and Electro-Optics Pacific Rim, Hong Kong, July 2018.

- I Molina-Fernández, Robert Halir, A Ortega-Moñux, JG Wangüemert-Pérez, Yang Chen, Jian-Jun He, Pavel Cheben, Jens H Schmid, T Domínguez Bucio, Ali Z Khokhar, M Banakar, K Grabska, Frederic Y Gardes, "Broadband and highperformance devices for the silicon and silicon-nitride platforms," 2017 19th International Conference on Transparent Optical Networks (ICTON), Girona, 2017, pp. 1-4.
- T. Li, M. Nedeljkovic, N. Hattasan, A. Z. Khokhar, S. A. Reynolds, S. Stanković, M. Banakar, W. Cao, Z. Qu, C. G. Littlejohns, J. Soler Penades, K. Grabska, L. Mastronardi, D. J. Thomson, F. Y. Gardes, G. T. Reed, H. Wu, Z. Zhou, and G. Z. Mashanovich, "Mid-infrared Ge-on-Si electro-absorption modulator," IEEE Group IV Photonics, Berlin, Germany, 23-25 August 2017.
- F. J. Runge, Y. Franz, C. G. Littlejohns, K. Grabska, S. Mailis, F. Y. Gardes, and A. C. Peacock, "Laser-assisted material composition engineering of SiGe planar waveguides," Photonics Global Conference, Singapore, July 2017.
- K. Grabska, M. Banakar, L. Mastronardi, C. Littlejohns, T. Rutirawut, A. Zilkie, C.
 G. Littlejohns, S. Saito, F. Gardes, "SiGe compound for compact high speed electroabsorption modulators", PSCS Zepler Institute Day, Southampton, 2017.
- K. M. Grabska, M. Banakar, T. Dominguez Bucio, M. Nedeljkovic, A. Barzaghi, A. Ballabio, G. Isella, C. G. Littlejohns, S. Saito, and F. Y. Gardes, "Multi composition GeSi tuneable concentration silicon-germanium wire structures for CMOS photonics", Enlighten Conference 2019.
- K. M. Grabska, C. G. Littlejohns, S. Saito F. Y. Gardes, "CMOS photonics", Enlighten Conference 2018.
- K. M. Grabska, C. G. Littlejohns, S. Saito F. Y. Gardes, "Silicon-Germanium for photonics applications", Silicon Photonics Summer School, Ghent, Belgium, 2016.

Conference (Invited)

L. Mastronardi, M. Banakar, A. Z. Khokhar, N. Hattasan, T. Rutirawut, T. Dominguez Bucio, K. M. Grabska, C. Littlejohns, A. Bazin, G. Z. Mashanovich, F. Y. Gardes, "56 Gbps Si/GeSi integrated EAM," SPIE/COS Photonics Asia 2018, Beijing, China, 11-13 October 2018. (Invited)

- M. Nedeljkovic, T. Li, C. G. Littlejohns, N. Hattasan, A. Z. Khokhar, S. A. Reynolds, S. Stanković, M. Banakar, W. Cao, Z. Qu, J. Soler Penades, K. Grabska, L. Mastronardi, L. Crudgington, X. Chen, A. Nazir Ahmed, M. Saïd Rouifed, H. Qiu, T. Guo Xin, H. Wang, H. Wu, Z. Zhou, G. T. Reed, D. J. Thomson, F. Y. Gardes, and G. Z. Mashanovich, "Silicon-on-insulator and germanium-on-silicon free carrier modulators for mid-infrared wavelengths," Conference on Lasers and Electro-Optics Pacific Rim, Singapore, July 2017. (invited)
- M. Nedeljkovic, C. G. Littlejohns, T. Li, N. Hattasan, A. Z. Khokhar, S. A. Reynolds, S. Stanković, M. Banakar, W. Cao, Z. Qu, J. Soler Penades, K. Grabska, L. Mastronardi, L. Crudgington, X. Chen, A. N. Ahmed, M. S. Rouifed, H. Qiu, T. G. Xin, H. Wang, H. Wu, Z. Zhou, G. T. Reed, D. J. Thomson, F. Y. Gardes, and G. Z. Mashanovich, "Silicon and germanium free carrier injection modulators for the midinfrared," SPIE 2017 Optics and Optoelectronics Symposium, Prague, Czech Republic, 24 -27 April 2017. (invited)
- F. Y. Gardes, Thalía Domínguez Bucio, Lorenzo Mastronardi, Nannicha Hattasan, K. M. Grabska, Callum J. Littlejohns, Mehdi Banakar, Graham T. Reed, Goran Z. Mashanovich, Antoine F. J. Runge, Yohan Franz, Anna C. Peacock, "Group IV compounds and silicon nitride for multiplatform integrated photonics" SPIE Photonics west, San Francisco, USA, January 2017 (Invited).
- F. Y. Gardes, C. G. Littlejohns, M. Nedeljkovic, T. Bucio Dominguez, N. Hattasan, M. Banakar, L. Mastronardi, K. Grabska, G. T. Reed, G. Z. Mashanovich, A.F. Runge, Y. Franz, A. Peacock, A. Al-Attili, S. Saito, "Group IV compounds for micrometre scale device and systems" SPIE Photonics Asia, Beijing, China, October 2016 (Invited).
- F. Y. Gardes, C. G. Littlejohns, M. Nedeljkovic, T. Bucio Dominguez, N. Hattasan, M. Banakar, L. Mastronardi, K. Grabska, G. T. Reed, G. Z. Mashanovich, A.F. Runge, Y. Franz, A. Peacock, A. Al-Attili, S. Saito, "Group IV compounds for integrated photonic applications" IEEE Group IV Photonics, Shanghai, China, August 2016 (Invited).

Journals

- L. Mastronardi, M. Banakar, A. Z. Khokhar, N. Hattasan, T. Rutirawut, T. Dominguez Bucio, K. M. Grabska, C. Littlejohns, A. Bazin, G. Z. Mashanovich, and F. Y. Gardes, "High-speed Si/GeSi hetero-structure electro absorption modulator," Optics Express, vol. 26, pp. 6663-6673, 2018
- Y. Chen, T. Domínguez Bucio, A. Z. Khokhar, M. Banakar, K. Grabska, F. Y. Gardes, R. Halir, I. Molina-Fernández, P. Cheben, and J.-J. He, "Experimental demonstration of an apodized-imaging chip-fiber grating coupler for Si3N4 waveguides," Optics Letters, vol.42, pp.3566-3569, 2017
- K. Debnath, M. Clementi, T. Domínguez Bucio, A. Z.Khokhar, M. Sotto, K. M. Grabska, D. Bajoni, M. Galli, S. Saito, F. Y. Gardes, "Ultrahigh-Q photonic crystal cavities in silicon rich nitride," Optics Express, vol. 25, pp. 27334-27340, 2017

Signature:

Date: 03.04.2020

Acknowledgements

Throughout the time spent on my PhD project, I have received significant help and support from my colleagues, friends and family without whom I would not have been able to complete my journey. I would like to take this opportunity to thank all the people who helped me along the way.

I would like to express my gratitude to my supervisory team; Professor Frederic Gardes, Dr Callum Littlejohns and Professor Shinichi Saito. Special thanks must go to Professor Frederic Gardes whose insight and knowledge in the subject matter steered me through this research.

Special thanks to all members of the Silicon Photonics Group, past and present, for all their help and kind support I have received over the years. In particular, I deeply appreciate all the help received from Dr Thalia Dominguez Bucio and Dr Milos Nedeljkovic, for support in design and data analysis throughout my project. Further thanks must go to Dr Mehdi Banakar, for teaching me a lot about processing, critical thinking and for his patience. I would also like to thank Dr Callum Littlejohns for all his help with RMG processing and Dr Antulio Tarazona for extended CMP training. I would also like to extend my thanks to Dr Daniel Burt, for providing company over the endless Raman measurements and Dr Abdelrahman Attili for his assistance with wet etch processing.

Cleanroom experience was a vital part of this project and therefore I would like to thank every technician who went above and beyond in supporting me through the fabrication for this work; Mike Perry, Dr Libe Arzubiaga Totorika, Dr Kian Kiang, Tracy Anscombe and Dr Owain Clarke. I grateful for all the training and answering endless questions, especially over my first years of the PhD.

A big thank you to collaborators from VTT Technical Research Centre of Finland and Politecnico di Milano and especially Professor Giovanni Isella, Dr Andrea Ballabio and Andrea Barzaghi.

I would like to say a big thank you to all my friends who were there for me throughout my journey; I am very grateful to have you all in my life and I cannot forget the special support you provided me over this very intense time. Wahida Taskin Bhuiyan, you gave me your unconditional support and took care of me when I needed it the most. Thanks must also go to the following for all their kindness and help whilst writing my thesis; Dr Katrina Morgan, Wanvisa Talataisong, Dr Pearl John, Juliette Schuurmans, Dr Tom Bradley, Dr Jamie Raynolds, Dr Anushka Gagnaik, Dr Omar Adnan, Jon Gorecki, Dr Colin Mitchell, and Dr Greg Jasion. Thank you to Dr Ben Lowe, Dr Nikolaos Matthaiakakis, Dr Christoph Riedel and Anna Hopkins for being there for me from my first months as a PhD student, all the way up to proofreading bits of my thesis! I cannot forget to say thank you to my family for their support, especially to my mum Malgorzata, my grandparents Czeslaw and Teresa and my brother Maciek.

Finally, I would like to acknowledge the funding bodies who made this- PhD possible by offering me this award; The Engineering and Physical Sciences Research Council (EPSRC) and Rockley Photonics.

xxiv

Chapter 1 Introduction

The rapid technological development of electronic devices and communication networks witnessed in the last few decades has been driven by the increasing number of digital devices and the raising expectations for telecommunication and computing capabilities. We are currently living in the Information Age [1], where technology has grown following the predictions of Moore's law. This law describes an observation of increased (doubled) number of transistors per Integrated Circuit (IC) every two years as illustrated in Figure 1. Moreover, additional exponential formulations often considered derivatives of Moore's law have been used to describe the technological development witnessed over the past few decades in different areas [2]; like for optical fibres which have become a crucial part of telecommunication. Butters' Law expresses an exponential growth in their data transmission speed every nine months which makes the cost of transmitting data exponentially lower [2][3]. Another example of rapid development of computational technology are quantum annealers with doubled number of qubits throughout every year over ten years period as predicted with 'Rose's law'- an example of the 4-qubit is Calypso from 2005 and the 2048-qubit Washington machine in 2015 [3][4].





The new technological developments have led to an ever-increasing need for data transfer, whilst transferring information at a low cost and retaining high processing speed and transmission rate. Electronic connections based on copper limit the performance of high-speed computers and interconnects, giving space for dramatic improvement through the use of optical connections. The replacement of traditional copper interconnects has proven to be crucial for the efficiency of high-speed data communication for both long and short distances, reducing heat dissipation and losses. A big part of the research on photonics has focused on developing higher data transmission rates, higher processing speed and higher bandwidth for telecommunication and computing systems respectively. A typical target application of silicon photonics has been the data links within data centres operating at 100Gbit/s or 400Gbit/s [5]. Turning point of this research was presented in 2004 as a metal–oxide–semiconductor (MOS) capacitor structure embedded in a silicon waveguide acting as a fast Si-based modulator exceeding 1 GHz [6].This research started a discussion about new possibilities for exceeding the speed of interconnects inside the computer [7].

The next chapter will introduce silicon photonics as a platform of choice for novel optoelectronics solutions highlighting silicon-germanium (SiGe) as a material suitable for high performance integrated devices. It discusses the focus of this project on the development and integration of crystalline SiGe and germanium (Ge) material on the silicon-on-insulator (SOI) platform. The integration is based on a technique called Rapid Melt Growth (RMG)/ Liquid Phase Epitaxy (LPE) [8][9] providing a means to integrate passive and active devices for silicon photonics applications, including high speed optical modulators and detectors, across a small area at a minimal cost.

1.1 Silicon Photonics

Silicon photonics is an evolving technology leading the way for low-cost, high performance applications bringing together the advantages of both photonics and electronics through integration onto the same substrate. Commonly used traditional silicon integrated circuits (IC) industry solutions can be easily adapted for the fabrication of optical interconnects at the chip level. Additionally, the compatibility with CMOS technology allows the integration of electronics and photonics on the same substrate.

Another factor leading to the success of silicon photonics is a high refractive index ratio of vital elements like e.g. silicon waveguide and silicon dioxide cladding (refractive index ratio for Si and SiO2 is 3.5/1.443) leading to smaller device footprint, more chips per wafer and to the realisation of sub-wavelength-sized cavities for photonic crystal structures, high-efficiency grating couplers [5] etc. Silicon technology also allows carrying more data with less power, avoiding the heating problems usually faced by traditional copper-based technology [5]. In addition, Silicon's

Introduction

transparency at 1550 nm and compatibility with Ge optical modulators and photodetectors make this platform highly suitable for multiple optoelectronics applications and components [1], examples to be listed are optical transceivers [65], sensing applications [11] or LIDAR technology [12].

In 1988 the first company to commercialize silicon photonics was founded, Bookham Technology. The first product was introduced in 2000; it was an arrayed waveguide grating (AWG) multiplexer, integrated with variable optical attenuators (VOA) [13]. A great amount of research on silicon photonics has been conducted since than by international companies such as Melanox [14], HP [15], Luxtera [16] with Cisco [17], Intel [18][19], IBM [20][21], Huawei/ Caliopa [22][23] or Rockley Photonics [24]. The interest of these companies has led to the development of silicon photonics products that are commercially available, including products like transceivers and sensors [24].

1.2 Silicon germanium applications

Germanium (Ge) is an example of a well-established material in electronics, compatible with CMOS technology [11]. Ge and SiGe are highly important materials of silicon photonics especially for the receiving element of the photonic circuits, as they are materials that absorb in the entire telecommunication window from the O (1310nm) to the L (1595nm) wavelength bands. Photodetectors based on Ge materials have high responsivity up to 1550 nm [1][25][26]. Alloys of Si and Ge described as Si_{1-x}Ge_x, where the varied composition of Si is denoted by x, have been part of the mainstream CMOS production since 1989, when IBM introduced SiGe as a material of choice for their products to increase computer performance by enhancing speed, simultaneously reducing the power consumption [27].

However, although Ge and Si are compatible materials, the lattice constant difference produces a mismatch of 4.2% that causes problems for fabrication of defect free interfaces between the two materials. The lattice mismatch issues can be partially overcome by implementing buffer layers with a graded composition (from pure Si through multiple compositions of Si_{1-x}Ge_x up to pure Ge) during the growth of the materials [28][29], but the technique is complex, expensive and challenging to implement. As a result, the development of a novel technology that can be used to integrate crystalline Ge/SiGe and silicon with an improved layer quality while minimising the cost and fabrication time is still desirable.

1.3 Rapid Melt Growth for silicon germanium and germanium on Siliconon-insulator platform

Silicon-germanium $(Si_{1-x}Ge_x)$ is fully miscible across the whole composition range, which allows for the whole spectrum of both the lattice constant and the band gap to be tuned between the values of pure Si and pure Ge [1][30]. The versatility of SiGe alloys is extremely important for photonics applications such as electro-absorption modulators and photodetectors, where the operating wavelength is strongly dependent on the Si content.

Moreover, Si_{1-x}Ge_x is a material of interest for high-performance modulator devices working in the C band such as electro absorption modulators (EAM) based on the Franz-Keldysh effect [30][31]. The maintained constant composition and the quality of the Si_{1-x}Ge_x alloys are crucial to determine the energy of the band gap and therefore the operating wavelength of said modulators. Previously reported results for SiGe EAM confirm lower power consumption (calculated modulator power 44 fJ/bit), and increased speed (data rate of 56 Gb/s at 1566 nm), making these devices attractive as building blocks for silicon photonics [31].

Traditional growth methods including various kinds of chemical vapour deposition (CVD) and epitaxy (e.g. Molecular Beam Epitaxy MBE) allow the growth of only a single SiGe composition across the wafer. Therefore, the devices that can be fabricated with these layers are strongly limited to a narrowband wavelength operation, while the simultaneous modulation of multiple wavelengths on a chip is desirable. Hence, the Rapid Melt Growth (RMG) technique has been proposed as a way to achieve the growth of multiple compositions at once with only one nonselective growth step and one anneal step. RMG is a technique, where the SiGe or Ge structures are grown as single crystalline material using the Si seed as a lattice template and a source of Si diffusing into Ge tailored structures embedded into an insulator. The composition of the Si1-xGex alloys can be tuned by combining RMG with especially designed structures due to mentioned full miscibility of these two elements [8]. After the RMG process, trenches with the shape of strips have shown a graded composition with decreasing Si concentration as their length increases away from the seed. However, a constant composition can be obtained when branches are added along the strips. This modified RMG technique has allowed the creation of strips with multiple constant compositions across silicon substrates with a negligible number of defects [9]. The engineering challenges of this process are linked to the precise control of the composition along structures and removing defects within the area designated for the devices.

The silicon-on-insulator (SOI) platform has become the platform of choice for silicon photonics as the fabrication of structures embedded in the silicon overlayer of the wafers enable seamless

Introduction

waveguide integration between both passive and active devices that can be fabricated simultaneously [26].

The novelty and aim of this work are to fabricate tailored SiGe structures on the SOI platform which, contrary to the silicon platform, are suitable for the integration of passive and active photonic devices. The engineering challenges are described in detailed in Chapter 5, process development.

1.4 Thesis outline

After a brief introduction to the work follows a discussion of background theory with special emphasis on silicon photonics and notable developments in the field. The thesis aims to demonstrate the importance of SiGe and Ge based photonics and the contribution this new work makes, which involves developing the integration of the RMG based Ge and $Si_{1-x}Ge_x$ growth technology on the SOI platform.

Chapter 2 provides a theoretical background about Si, Ge and $Si_{1-x}Ge_x$ properties along with applications of these materials. Additionally, a literature review of relevant techniques suitable for the material growth is presented. This leads to justification of the RMG technique development being the scope of this work.

Chapter 3 contains an overview of methods used for fabrication and characterisation of the RMG integration work and process development. Understanding of advantages and disadvantages of those is essential to explain choices made for the fabrication of RMG structures on SOI.

Chapter 4 explains motivation of the project with a special focus on waveguide integration. Further part provides masks and experiments design for RMG tailored structures on the SOI platform.

Chapter 5 contains description of work done on the process development for the experiments necessary for SOI integration. This chapter also contains the characterisation and discussion on results obtained for RMG structures embedded within the SOI platform.

Chapter 6 summarizes the PhD thesis with discussion and conclusions on the results of all work carried out. This chapter also contains the possible applications and future work based on findings explained in previous chapters.
Chapter 2 Theoretical background and literature review

This chapter contains theoretical background regarding a subset of semiconductors (Si, SiGe and Ge), which are crucial for this thesis, in Section 2.1. Understanding the theoretical background behind the growth process at a fundamental level will allow planning of processing for optimal photonic devices and therefore Section 2.2 is describing it. Section 2.3 introduces Rapid Melt Growth (RMG) background with an overview of materials deposition and growth methods traditionally used for fabrication of silicon photonics or electronic devices. Section 2.4 describes the hydrogen role in the grown film. Finally, Section 2.5 covers the surface tension topic.

2.1 Properties of Silicon, Germanium and Si_{1-x}Ge_x alloys

Si and Ge are group IV elemental semiconductors with the electron structure as shown in Figure 2. Si is the chemical element with atomic number 14, whereas the Ge atomic number is 32, with electron configurations: $1s^2 2s^2 2p^6 3s^2 3p^2$ and $1s^2 2s^2 2p^6 3s^2 3p^6 4s^2 3d^{10} 4p^2$, respectively. The higher number of electrons and filled d-orbital result in a larger atomic radius of the Ge atom and smaller band gap than Si. Additionally, Ge possesses both higher electron and hole mobility than Si [1][2].



Figure 2 Si and Ge are group IV elements with atomic numbers of 14 and 32 respectively, with 4 valence electrons.

The most significant properties of Si and Ge for this thesis are gathered in Table 1 for a clear comparison between these two materials. The band_gap and lattice constant can be tuneable by the RMG process which will be detailed in later parts of this thesis. The difference in melting point of these two materials allows RMG to occur, which will be expanded in later parts of this Chapter.

Finally, the hole mobility and electron mobility influence applications of semiconductors and have an impact on their success in electronics, e.g. by affecting operation speed or the circuit operating frequency [1].

Property	Si	Ge
Band gap (eV)	1.12	0.66
Lattice constant (Å)	5.43	5.66
Melting point (°C)	1415	937
Hole mobility (cm ² (V· s) ⁻¹)	450	1900
Electron mobility (cm ² (V· s) ⁻¹)	1400	3900

Table 1 Important properties of Si and Ge reproduced from [3].

For the fabrication of Si, Ge or SiGe based devices, the compatibility with the CMOS production line results in a major advantage over III-V group semiconductors, traditionally used in electronics but difficult to integrate with the Si platform [4]. Additionally, the higher carrier mobilities of Ge compared to Si could be given as a major advantage; however, the lack of a stable germanium oxide (GeO₂) is the main reason for the prevalence of Si as the prominent semiconducting material (Si forms a very stable native oxide, SiO₂). As mentioned in Chapter 1, the difference in refractive index of materials and their oxides is crucial for multiple applications including the waveguide fabrication, and Si has a strong refractive index contrast with its native oxide. Refractive index of SiGe alloys is highly dependent on the composition, although it will not be described in more detail as not being the main objective of research described within this thesis.

All of these properties are supporting the suitability of Si for photonics, however this material has much more to offer, including high thermal conductivity and high optical damage threshold, where both values are around 10 times higher than for GaAs, a commonly used III-V semiconductor [5].

2.1.1 Energy band gap

The energy band gap represents the minimum energy needed for an electron to move from the valance band to the conduction band. Semiconductor materials can be divided into direct and indirect band gap materials, as is shown in

Figure 3Figure 3. In direct band gap materials, the bottom of the conduction band and the top of the valance band correspond to the same value of electron momentum. However, in indirect band

gap materials, an additional change of momentum is required to achieve the transition from the valence band to the conduction band, thus resulting in a two-step phenomenon linked to a lower probability of occurrence compared to direct band gap materials.



Figure 3 The band diagram for semiconductors: (a) direct band gap semiconductor diagram with the conduction minimum aligned in *k*-space with the valence band maximum, (b) indirect band gap semiconductor diagram with conduction band minimum shifted in *k*-space with respect to the valence band maximum, (c) Ge band diagram; reproduced from [6].

Si and Ge are indirect band gap materials, where the indirect band gap is equal to 1.11 eV for Si and 0.66 eV for Ge (at 300 K). Close to this indirect band gap, Ge possesses a direct band gap with slightly higher value of 0.8 eV. Examples of direct band gap materials like GaAs or InAs [7] are shown next to Si and Ge in Figure 4. In Figure 4, the absorption coefficient, α , of common semiconductors (including Si and Ge and those from the III-V group) is plotted as a function of Photon energy (corresponding to the wavelength).



Figure 4 Absorption coefficient and band gap of different semiconductors for a range of wavelengths [8]. This plot is revealing the nature of their band-structure and band gap energy.

As shown in Figure 4, the value of the direct band gap of Ge corresponds to the telecommunication wavelength of 1550 nm, resulting in it being a suitable material for Electro-absorption Modulators (EAM) [5] and photodetectors. The band gap of Si (mentioned 1.1 eV) [8] corresponds to the optical absorption band edge at a wavelength of 1120 nm. Therefore, for $\lambda > 1120$ nm Si is transparent (applied e.g. for Si waveguides) in contrast to the range of λ < 1120 nm, where Si is highly absorbing (used e.g. in photodetectors). The window of low-loss wavelength (when Si is transparent) is expanded up to 7000 nm making Si a material of choice also for mid-infrared (MIR) applications [9]. The addition of Ge to Si and formation of the range of Si_{1-x}Ge_x alloys, shifts the absorption edge from λ = 1120 nm to 1300 nm, or even to 1550 nm. The full miscibility of Si and Ge allows the tunability of the band gap between that of pure Si and that of pure Ge. The theoretical study of the band gap of these alloys was conducted as early as 1958 [10]. Afterwards, experimental results for low temperature band gap were fitted with analytical equations [11]. Equations 1 and 2 describe the value of band gap for varied compositions of $Si_{1-x}Ge_x$, where the critical content of Ge is x=0.85. For the Ge content x lower than 0.85 as described in Equation 1, the band gap is closer to the Si band gap value whereas with increasing value of x, the conduction-band is more like that of pure Ge and is even called "Ge-like" and described as in Equation 2 [12, 13].

$$E(x) = 1.155 - 0.43x + 0.206x^2$$
 eV (Equation 1)

$$E(x) = 2.010 - 1.270x \text{ eV}$$
 (Equation 2)

Based on equations 1 and 2, the graph for the SiGe alloy band gap can be drawn, as shown in Figure 5. The shape of a band gap curve for Ge content under x=0.85 is parabolic for unstrained SiGe, whereas for the Ge content above x=0.85 the energy band gap is linearly dependent on the composition. The band bad curve for strained SiGe with the Ge fraction x < 0.3, by contrast, is linearly dependent on the SiGe composition.



Figure 5 The band gap for the spectrum of $Si_{1-x}Ge_x$ alloys based on calculations expressed in equations 1 and 2 [12].

The valleys shown in Figure 5 correspond to the Brillouin zones for Si and Ge unit cells and will be further explained in 2.1.2.

2.1.2 Lattice constant

The lattice constant defines the distance between the units of a crystal lattice cell in threedimensional space and depending on the shape of the cell, it can have either one, two or three different values. Similarly to many III-V group materials, structures of Si and Ge are defined as the face centred cubic (fcc) unit cell, also known as a diamond structure as shown in Figure 6. For the fcc structure, all dimensions of the lattice constant are equal and described as *a*.





The fcc structure can be represented in the k-wave vector coordinates which is called the Brillouin zone. As shown in Figure 7, the main symmetry points of the Brillouin zone are relevant for the shape of a band gap curve for $Si_{1-x}Ge_x$ alloys. This representation is shown in Figure 7 with the main symmetry points defined across the Brillouin zone. The conduction-band minimum (CBM) occurs at the L-point in the Brillouin zone in Ge and near the X-point in Si [14].



Symmetry points (*u*,*v*,*w*) [*k*_{*w*},*k*_{*y*},*k*₂] Γ: (0,0,0) [0,0,0] X: (0,1/2,1/2) [0,2π/a,0] L: (1/2,1/2,1/2) [π/a,π/a,π/a] W: (1/4,3/4,1/2) [π/a,2π/a,0] U: (1/4,5/8,5/8) [π/2a,2π/a,π/2a] K: (3/8,3/4,3/8) [3π/2a,3π/2a,0]



The difference of lattice constant between Si and Ge, 0.5431 nm and 0.5658 nm respectively, causes a lattice mismatch with calculated value of 4.2% on the interface. For $Si_{1-x}Ge_x$ alloys, the lattice constant depends on the Ge composition with values obtained between those for pure Si and pure

Ge. The lattice constant of $Si_{1-x}Ge_x$ alloys can be calculated using Vegard's rule [16] described with Equation 3:

$$a_{Si1-xGex} = a_{Si} + x(a_{Ge} - a_{Si})$$
 (Equation 3)

where *a* is the lattice constant for Si, Ge or $Si_{1-x}Ge_x$ and *x* describes the Ge content.

The lattice constant parameters for varied Ge content, *x*, are included in Table 2, showing proportional trend across the whole range of the Ge content.

 Table 2
 The Ge content x in Si_{1-x}Ge_x alloys and corresponding values of the lattice parameter

 [17,18]. A proportional relationship as predicted from Vegard's rule is observed for the whole range of x values.

Ge content, <i>x</i>	Lattice constant Si _{1-x} Ge _x , <i>a</i> (nm)	
0	0.5431	
0.25	0.54825	
0.5	0.55373	
0.75	0.5596	
1	0.56575	

The lattice mismatch strongly enhances the probability of crystal defects in Ge layers grown on Si [17], which typically inhibit the resulting device properties such as dark current and responsivity. Nevertheless, a great number of high-performance active devices based on Ge have been developed using epitaxial Ge grown directly on Si, which will be explained in detail in 2.2.

2.2 Ge and Si_{1-x}Ge_x growth on Si

The lattice mismatch between Ge and Si is considered as a source of strain energy which causes deformation of the film and leads to the formation of several defects, especially on the interface between these materials [19]. As shown in Table 2, the lattice constant is varied for different SiGe composition which is also causing defects on the interface between SiGe and the Si substrate. For the simplicity of this Chapter, only growth will be described in detail on the example of pure Ge as the principle analogical to SiGe growth. Values of the SiGe lattice constant closer to the one of Si are used to minimised undesirable lattice mismatch e.g. as buffer layers which will be mention further on. Dislocations and twinning defects have an impact on the smoothness of the surface or the continuity of the layer, especially in the form of threading dislocations terminated at the

surface. Moreover, the further fabrication process e.g. wafer bonding [1] or selective etching can be interrupted if the number of defects is higher than the tolerance of the process; therefore, defects can impact the yield of the fabrication process [2]. The performance of photonics devices is also related to the number of defects including the lifetime of devices which has been shown to be extended with a decreased number of defects in the device material [1][3][20]. During the growth of Ge on Si, its lattice spacing will keep changing to conform to the Si lattice constant, however, a defect-free strained Ge on Si was demonstrated only for thickness in a range of 4-10 nm [19]. Above this value, the lattice mismatch causes increased stress within the grown layer and more defects. The defects are a mechanism for relieving the accumulated strain energy.

The fundamental modes of film growth can be distinguished as shown in Figure 8.



Figure 8 Basic modes of film growth: (a) Volmer–Weber mode; (b) Frank–van-der Merwe mode; (c) Stranski–Krastanov mode.

The Volmer-Weber mode corresponds to the situation where 3D islands of the material are formed directly on the substrate without any 2D layer underneath [20]. The Frank-van-der Merwe mode is a mode of the film growth layer by layer without island formation. This mode is typical for materials with perfectly matching lattices which is mostly relevant to homoepitaxy [21]. The Stranski-Krastanov mode corresponds to the two-step growth when 2D layers are formed up to certain critical thickness subsequently followed by formation 3D islands [22]. Firstly, to initiate the growth, a source of atoms or molecules in either liquid, solid or gaseous state is required. To build up a layer, a condition of lower chemical potential of the desired film in relation to the particles of the substrate must be conserved. The chemical potential describes how much energy is needed to join single particles to the material layer. In the situation with two materials of the same chemical potential, nothing would grow.

The strain energy can be determined by using Equation 4, where the strain energy increases proportionally to the thickness (t) of the Ge growth,

$$Estrain = \lambda At \left(\frac{\Delta a}{a}\right)^2$$
 (Equation 4)

where λ is the biaxial elastic modulus, Δa is the forced change in lattice constant, a is the unstrained lattice constant, A is the area and t is the thickness of the film.

Over the critical thickness mentioned for Ge on Si growth, the probability of dislocations formed at the interface increases. This is leading to the formation of Ge islands and the reduction of the elastic energy of the surface as shown in Figure 9. The shown growth system is typical for Ge-on-Si thin heteroepitaxy, with the transition between 2D layer growth and 3D island growth [22, 23].



Figure 9 Transmission electron microscope cross-sections of Ge-on-Si layers for Ge epitaxial growth at different temperatures, Stransky–Krastanov growth of Ge-on-Si in an ultrahigh vacuum CVD reactor at 550°C and 350°C (a, b respectively), reproduced from [24]. The thickness of the Ge layer shown here is 60 nm.

The mode of the crystal growth is dependent on the wetting mechanism and the surface strain energy [25]. The mechanism of growing Ge layers is shown in Figure 10.

The first monolayer of Ge on the top of Si (Figure 10a) and several further monolayers (Figure 10b) are characterised with the strain associated to the 4.2% lattice mismatch between Ge and Si. A wetting layer is firstly created on the surface of the substrate. As the film grows above the critical thickness, the strain energy becomes too large for the Ge to maintain its compressed form and therefore defects form to relieve the strain, supporting the growth of islands instead of smooth layers [26].





Figure 10 Cross-section atomic view of stages of the Ge-on-Si growth; a) the wetting layer, b) growth below the critical thickness, c) thickness over the critical thickness when dislocations and other defects occur.

The stress caused by the lattice mismatch between Si and Ge releases lattice dislocations which propagate throughout the Ge film in the form of threading dislocations. It is desirable to reduce the Threading Dislocation Density (TDD) in order to reduce the device dark current, the main source of noise in the resulting devices [27]. Furthermore, defects act as generation and recombination centres for carriers, and therefore acceptor-like dislocations in the Ge layer will form band barriers and reduce the responsivity of photodetectors [27]. The most widely used methods for measuring TDD are plan- view TEM and selective defect etching [12]. Reported typical values for Ge on Si are in the range from 2x10⁶ to 10¹⁰ cm⁻² [19].

2.3 Ge and Si_{1-x}Ge_x on Si growth techniques

A variety of crystalline Ge/SiGe-on-Silicon and Ge/SiGe-on-insulator (GOI/SGOI) fabrication techniques have been investigated for photonics devices in order to reduce material defects and improve the performance of the final device. The fabrication of high-quality and large-scale SGOI and GOI is also highly desirable for CMOS integrated circuits to alleviate some of the limitations of Si only based devices due to its wide band gap, especially important for wavelengths above 1.1 μ m [28]. The addition of Ge to Si allows decreased size or lower power consumption of devices. Ge also creates a bridge between silicon photonics and III-V materials [29]. The quality of an interface between layers can vary with the method used for material growth, so in this section the main SiGe and Ge growth techniques are introduced with a focus on the well-established epitaxy and Chemical Vapour Deposition (CVD) methods.

2.3.1 Molecular beam epitaxy

Molecular beam epitaxy (MBE) is one of the well-established methods widely used for depositing layers or multilayers of semiconductors including complex structures like multilayer quantum wells [30]. The MBE technique grows material films by heating a solid crystalline source which will firstly sublimate and later condense on a substrate. This technique provides a high-quality film with the highest achievable purity over other methods, but the growth rate is slower due to an ultra-high vacuum (UHV) of the order of 10⁻⁸ Pa needed to obtain growth. Moreover, SiGe growth with a constant composition is more complicated because of a melting temperature difference between Si and Ge [31]. Well controlled MBE techniques can be used for reducing the threading dislocation density (TDD) in Ge layers, which was reported as early as 1984 with a graded SiGe buffer between Si and Ge [31]. A methodology using SiGe buffer layers was developed by the Fitzgerald's group over many years to demonstrate Ge films with threading dislocations of $< 2 \times 10^6$ cm⁻² in 2006 [32]. This work contains a plethora of methods for defects, strain and roughness reduction like using a changeable temperature for specific SiGe composition growth, or Chemical Mechanical Polishing/ Planarization (CMP). MBE based Ge-on-Si growth with an additional annealing (700°C, 30 minutes) was demonstrated as another possibility for lowering TDD (1×10⁷ cm⁻² in 1989) [44]. In 2004, Liu et al. [33] developed the MBE process of Ge growth below 450°C. This technique was used shortly after for fabrication of low temperature p-Ge/n-Si photodetectors for silicon-based photonics [34]. The very low defect concentration and a layer by layer growth mechanism are suitable for the fabrication of quantum wells [30], single crystal thin films or superlattices [3]. However, the cost of set-up is higher than other techniques and the growth rate is slower than comparable growth methods [4]. Finally, in order to achieve a good quality film growth, an extremely flat substrate surface is required. Nevertheless, in spite of higher cost, MBE is commonly used especially for fabrication of III-V materials and for all the applications where the higher cost of fabrication can be balanced by beneficial properties of the resulting material [4].

The ability to grow only one defined SiGe composition across the wafer and the relatively complicated and expensive fabrication (multilayers and multiple depositions) are the main disadvantages of MBE for SiGe layer growth.

2.3.2 Chemical Vapour Deposition

Another technique of SiGe/ Ge growth is the widely used Chemical Vapour Deposition (CVD) where the combination of gases, or even one gas, are reduced or decomposed which results in film deposition on the surface of the substrate [1]. Ge-on-Si growth using CVD started as early as 1981 with GeH₄ in H₂ atmosphere [47]. Firstly, a range of high temperature processes (600-900 °C) was used to form crystalline material. Lower temperature processes (500-600 °C) were also

demonstrated but only for films with a thickness in 1-2 µm range; below 1 µm thickness islands of material were created instead of smooth layers [47]. In order to decrease the number of dislocations, a graded SiGe buffer layer, post-process annealing, and CMP methods are used for CVD based materials. SiGe layers deposited with UHV/CVD were used as a buffer multilayer SiGe structure to achieve a lower dislocation density Ge structure on the top. To decrease the number of TDD within Ge layer from 10⁷ to 10⁶ per cm², an additional CMP step was implemented for buffer layers after reaching 50% Ge content within SiGe [48]. As SiGe buffer layer processing is well established, examples of its application for development of other materials, like GaAs deposited using Metal-organic CVD (MOCVD), were also demonstrated [49]. However, buffer layers are not desirable because of decreased uniformity along the thickness of the layer which either impacts properties of the miniaturised devices or is adding additional volume.

Nowadays, many kinds of CVD can be distinguished; based on a difference in pressure (from Atmospheric Pressure CVD (APCVD), Low Pressure CVD (LPCVD) [50] to Ultrahigh Vacuum CVD (UHVCVD)), character of vapour, heating method or plasma method like Plasma Enhanced CVD (PECVD). Reduced Pressure CVD (RPCVD) is commonly used for Ge growth [51]. The PECVD and LEPECVD [52,53] will be described in Chapter 3 as techniques used for the fabrication.

2.3.3 Rapid Melt Growth

Rapid Melt Growth (RMG), also known as Liquid Phase Epitaxy (LPE), is a further technique used for SiGe/Ge-on-Si or Ge-on-SOI growth. The origins of RMG in the semiconductor industry go back to 1960 for crystal growth of III-V semiconductor materials [35][36], which were commonly used for multiple applications like solar cells, laser diodes, detectors, and LEDs [37][38][39]. RMG has become a potential method for Ge and SiGe relatively recently, from 2004 [40]. Currently, various work based on RMG can be found across several research groups for GOI and SGOI growth [41][42][43].

The RMG process requires two different materials, e.g. Si and Ge, it is called heteroepitaxy and it is possible due to the difference between melting temperatures of Si substrate and Ge (melting points: Si - 1410 °C, Ge - 938 °C) [25]. The Si substrate with Ge structures propagating onto an insulator layer (and encapsulated in an insulator) is heated above the Ge melting point and Si diffuses from the seed area into the melted Ge structures. After the temperature peak is achieved, the system is subsequently cooled and SiGe is regrown from Si seeds. The higher melting point of the Si seed assures the solidification of the high Si content SiGe melt into a crystal with a lattice close to the Si seed. This solidification happens earlier than for pure Ge which enables the recrystallization to start in the seed area with the solid/liquid regrowth front propagating along the Ge features resulting in a single crystal growth. It should be mentioned that Si diffuses from the

seed until the solidification concentration is reached for an adequate temperature as shown in Figure 11. Following the phase diagram in Figure 11, Si is preferentially absorbed at the regrowth front into the solid depleting the liquid phase of Si leading to pure Ge at the end of the strip. This way of crystallisation minimizes the lattice mismatch due to the graded concentration of the SiGe along the strips.





The temperature range of the solidification process depend on the composition of $Si_{1-x}Ge_x$ alloys and can be described with solidus and liquidus lines of the phase diagram as shown in Figure 11. The alloys are heated to the temperature above the liquidus line, when the whole volume is liquid, and subsequently cooled to the fully solid state, with temperatures below the solidus line.

The defects across Ge/SiGe strips are terminated on the interface with the insulating layers by the process known as the epitaxial necking [44,45]. This method provides a smaller defect probability along the structures because most of defects will be concentrated in the seed area. The principle of the mechanism is shown in Figure 12. The low-quality material found near the seed area is not used in later device fabrication.







An overview of the processes leading to the fabrication of RMG- based SiGe/ Ge structures is shown in Figure 13 for processing on a Si substrate.



Figure 13 The RMG process for Ge structures on a Si substrate. Step 1- deposition of an insulator (e.g. SiO₂) on a Si substrate, Step 2- a patterned layer of insulator is etched to access the Si underneath, Step 3- Ge growth by any non-selective method, Step 4 -Ge structures are etched with preserved access to the Si seed, Step 5- deposition of an insulating capping layer, Step 6: Rapid Thermal Anneal (RTA). Reproduced from [17, 45].

The first application of RMG for no defect single crystalline Ge, performed in 2004, was based on UHV/CVD used for the non-selective Ge layer deposited on the top of an insulator with etched Si seed windows and Ge features [40]. This research was followed by work from another group to show a quality dependence on increasing width of Ge strips and to investigate the Ge composition along changeable distance from the Si seed [46]. The results from this work are shown in Figure 14 with visible limitation for the quality of Ge strips; only 3 μ m wide strips and shorter than 70 μ m were fabricated without major defects [46][47].



Figure 14 (a) Schematic structure and (b) optical micrograph of the sample [substrate: Si(100)] after RTA (1000° C, 1 s) reproduced from [47].

Research on the width limitation resulted in an increase to 5 μ m width SiGe strips with Lanthanum (La) as an insulating layer [48]. However, La is not compatible with the CMOS production line creating issues with standard processing. Nevertheless, this research shows the dependence between the insulator layer and surface tension of SiGe RMG structures. Another group researched the influence of varied initial SiGe composition on the Ge final composition profile along melted structures for the SGOI growth. The outcome of their experiments confirmed the similarity between profiles along the length of strips for all initial compositions of SiGe in the melt (composition of first not selective growth/deposition) [49]. Results of this research are shown in Figure 15.





Figure 15 The dependence of the composition profile along the length of the strip, x, for a variety of initial compositions of SiGe [49].

Another stage of RMG-based SiGe processing was the introduction of the multilayer SiGe strips and presenting the dependence between the Ge composition and cooling rate (CR). The outcome of this experiment shows the impact of cooling rate and the strip length on the final composition of the strips [41][50]. Results of this research are shown in Figure 16 and Figure 17.



Figure 16 Dependence of the SiGe composition along strips for changeable cooling rate reproduced from [50].

Nevertheless, other experiments confirmed the profile independence of peak temperature anneal time [41].



Figure 17 The comparison of SiGe composition profiles after RMG at a range of anneal times, reproduced from [41].

In 2014, waveguide integrated RMG based photodetectors were demonstrated on Ge stripes [72]. A constant composition within strips of SiGe is desirable to fabricate devices so the RMG process was developed further to maintain constant composition of SiGe within designated regions [17]. This result was obtained by forming crucibles shaped as tailored structure (called "tree-like structures") with cooling branches along the SiGe strips as shown in Figure 18 for Si platform.



Figure 18 Tailored RMG structure on Si platform, reproduced from [1].

The RMG method was confirmed to be a method of choice for the growth of good quality SiGe material, with tuneable composition and, having low defect density. This is unique among the growth techniques mentioned. Crucially, controlled growth of the multiple composition of SiGe across a single structure grown on an SOI substrate is desirable and the first ever attempt is described in Chapters 4-6.

2.4. Hydrogen within Ge structures

Impurities have an impact on the continuity and quality of the layers of Ge or even SiO₂ obtained from each growth technique. A commonly occurring example of an extrinsic impurity within Ge or Si films is hydrogen, which is easily incorporated into the growing material during the growth or later processing [14]. The interstitial hydrogen is amphoteric in Si which means it can act as an acceptor (in n-type materials) or a donor (in p-type materials) of electrons. However, in the Ge lattice hydrogen is usually an acceptor [14]. Amphoteric character of hydrogen increases the chance of incorporation into layers of Ge film on insulator.

Hydrogen exists within the Ge lattice as hydrogen atoms or ions. The three forms of hydrogen can be distinguished as the following: H^* which means proton only without electron, H^0 which is a neutral hydrogen atom and H^- describing a standard hydrogen atom with an additional electron. Possible configurations of fcc unit cells with incorporated hydrogen are shown in Figure 19.





Hydrogen is confirmed to be a reason for decreased performance of Ge based devices e.g. detectors, due to increased leakage current resulting in reduced sensitivity or increased risk of breakdown of the device [1].

The hydrogen presence can cause the voids within a layer because of hydrogen out-diffusion associated to higher temperature treatment [14]. An example of a high temperature process is annealing which is used for e.g. densification of layers or Rapid Thermal Anneal (RTA) for dopant activation or RMG. Due to the presented issues linked to hydrogen within the layer, in the literature, a possible method of dehydrogenation of Ge is listed as post deposition anneal at 300°C for 30 min [14]. Hydrogen is also incorporated into SiO₂ films grown by PECVD [17], which can cause problems during subsequent thermal processes where the hydrogen can outgas into other layers. Hence it is often desirable to remove hydrogen from the SiO₂ layer, particularly for the RMG process described later. It has been shown that outgassing of hydrogen from SiO₂ can be performed at 700 °C for 30

minutes and the amount of hydrogen removed can be tested through the thickness measurement (e.g. ellipsometry) of the layer before and after annealing [17].

2.4 Surface tension

The surface tension is a cohesive force among liquid molecules tending to obtain a minimum surface area for a given volume e.g. shape of the sphere [58]. The liquid would often tend to obtain this shape after anneal/ melt. A way to avoid voids within the solidified layer is to introduce an optimized shape with lower surface tension.

As was reported in the literature, the surface tension and the contact angle (wetting angle) of the liquid Ge depends on the material below the Ge layer [41]. The possible shapes of the droplets and contact angles are shown in Figure 20. The contact angle is the angle between a surface and the drop.



Figure 20 Contact angle for a water droplet on a) normal surface, b) superhydrophobic surface and c) shape of the drops for various levels of surface tension resulting in different contact angles.

The change of the material located underneath the Ge layer can be also important for RMG processing. The decrease of the observed surface tension within the SiGe structures (resulting in the possibility of wider dimensions of the device without visible voids or rounded shape in corners) was reported by using Lanthanum oxide La₂O₃ instead of the most commonly used insulator- SiO₂ [48]. The GOI strips of different widths after the anneal process are shown in Figure 21.



Figure 21 SEM images of Ge strips formed by RMG on SiO₂ with a cap thickness of 500 nm SiO₂ [40].

The rounded shape of the material (Ge), and breaks within the wire structures were explained by the deformation of the SiO_2 capping layer happening during the viscous flow of the RMG process. The high interface energy between liquid Ge and the solid SiO_2 causes the capping layer deformation and Ge aggregation, as the cap layer is softer at higher temperatures. The increase of SiGe/Ge wire deformation was observed with increased width of the wires. The main conclusion of this research work is the width constraints due to the interaction between the liquid Ge and the insulator layers interfaces [32].

The change of the shape of structures can be observed for both the melting process and cooling time [35], shown in Figure 22. The capping shape can be also affected according to the change of the liquid structures underneath.



Figure 22 Comparison of Ge strips shapes before and after the RMG process reproduced from [59]. The effect of lateral expansion along free edges and surface tension were investigated for a variety of strip widths.

Reference [59] analyses and compares various aspects of the RMG for Ge on insulator with the thermal cap affecting structures underneath. According to the research, detailed analysis was performed for various dimensions of structures in order to determine how the surface tension affects the shape of the grown structures. However, this approach may not be fully practical for the design demonstrated later on in this thesis due to the silicon nitride layer underneath the Ge layer, which is replaced by an SiO2 layer in this work. The rounded shape of the strips is also linked to Ge volume contraction which was reported as 5.2% for the RTA melt [59]. Moreover, the curve of the free end of the strip is dependent on the ratio between the volume of the strip affected by the surface tension and the area of the free edge [42].

2.5 Summary

Si and Ge are vital elements of optoelectronics devices. The tunability of SiGe properties is highly attractive for a plethora of applications like detectors or modulators, hence the integration on SOI is required. Across the available methods of deposition, only RMG can deliver tuneable composition of SiGe alloys just after one not selective process. The SOI integration is challenging from the engineering point of view and challenges mentioned within Chapter 2 (e.g. Hydrogen content, geometry of structures and surface tension) must be considered.

Methods

Chapter 3 Methods

This chapter contains descriptions of the fundamental techniques required for the fabrication and characterisation of structures being a subject of this PhD thesis.

3.1 Fabrication

The microfabrication of multi-layered structures, including the ones described for the RMG process and referred as "tree like structures", involves multiple subsequent processes like deposition, lithography, etching, planarization or annealing. This chapter contains a short overview of all techniques involved in the fabrication of SOI integrated RMG structures with a short explanation regarding the motivations for their selection. Process development and choice of the technique is crucial for the fabrication outcome, so the understanding of basics is of high importance.

3.1.1 Deposition techniques

The deposition techniques used during the fabrication can be listed as: Plasma Enhanced Chemical Vapour Deposition (PECVD) for SiO_2 (thermal cap for annealing and the first tests of insulating layer filling trenches) and Ge (first test runs only), Low Energy Plasma Enhanced Chemical Vapour Deposition (LEPECVD) for Ge or SiGe and thermal oxidation of Si in the final fabrication.

3.1.1.1 Chemical Vapour Deposition

Chemical Vapour Deposition (CVD) is a technique where the substrate, e.g. a Si wafer, is exposed to one or more volatile precursors, which can react with each other or decompose giving a desired thin film of a material on the surface of the substrate. The quality of the deposited layer depends on multiple process properties such as temperature, gas flow rate, precursors gases, pressure and the presence or not of plasma for enhanced deposition.

In particular, CVD techniques may be used for the deposition of Ge, SiO₂ and silicon nitride (SiN) through the reaction of precursor gases [1].

The reaction for Ge can be written as a decomposition of the precursor germane GeH₄ as follows:

$$GeH_4(gas) \rightarrow Ge_{(solid)} + 2H_2(gas)$$

This process is preferably conducted in lower range of temperature (around 400 $^{\circ}$ C) to avoid the oxygen incorporated into film.

The formation of SiO_2 films can be written as the following reaction between silane SiH_4 and nitrous oxide N_2O :

$SiH_4 (gas) + 2N_2O(gas) \rightarrow SiO_2(solid) + 2H_2(gas) + 2N_2(gas)$

The decomposition of SiH₄ requires higher substrate temperature (600-700 °C) than GeH₄. Hence, this mixture of gases is the most used for SiO₂ creation. N₂O is less reactive and more selective the oxygen O₂ giving more control over oxygen content throughout the film and slowing down reaction.

The formation of silicon nitride SiN_x is also an additional type of reaction based on SiH_4 , however, ammonia (NH_3) is most often required. The reaction can be written as:

$3SiH_4(gas) + 4NH_3(gas) \rightarrow Si_3N_4(solid) + 12H_2(gas)$

The standard CVD technique can be modified in multiple ways, where the techniques PECVD and LEPECVD used within this thesis are based on using a plasma enhancement mechanism.

The main advantage of PECVD over CVD is lower deposition temperature obtained by using plasma which induces a chemical reaction and results in a faster deposition rate. Plasma is created with parallel electrodes; a grounded electrode and an RF-energized electrode placed around the chamber. The gases are transported into the chamber where Radio frequency (RF) power excites the reactants into plasma after dissociation and ionisation. The gases in this form are more reactive which increases the probability of film deposition on the substrate. After the deposition, the surface reactions are followed by the removal of by-products from the reaction chamber. It was found that coproduct gases, e.g. H₂, can be incorporated into the Si or Ge film which will impact the properties of the deposited material [2][3]. To avoid undesirable H₂ related effect in the film throughout further processing, a degassing thermal treatment leading to densification/relaxation is often used [4]. The temperature, pressure and gases flow parameters are the most important factors within the PECVD technique [5]. The substrate temperature is considered crucial for film properties such as optical (e.g. refractive index) or surface metrology like roughness or grain size [6][7] Moreover, high temperature can affect previously created features or doping profiles across the substrate, so it should be considered as a factor during process planning. In PECVD type deposition, the most common range of temperatures reported are between 200 and 400 °C, whereas standard CVD methods require temperatures from 600 up to 800 °C [4]. Also, pressure impacts rate of the film growth [8] and it has an impact on the uniformity across the wafer, where higher level of uniformity is reported for lower pressure techniques. Gas flow and gas ratio are related to the composition of the film and can be used to define deposition rate [5] along with previously mentioned parameters. The increased deposition rate is considered as a factor increasing a surface roughness of the film as demonstrated in [4][9].

Methods

A diagram of a PECVD tool is shown in Figure 23.



Figure 23 A diagram of a PECVD chamber with a view of the gas system, location of the RF power source and the location of the wafer. Reproduced from [9].

In the LEPECVD technique, a key modification is the lower energy of plasma ions compared to traditional plasma-based processing, which is applied to avoid possible damage to the crystalline structure of the deposited film by high energy ions existing for other techniques [8]. The source of electrons for LEPECVD can be e.g. a Tantalum (Ta) filament heated by high current which allows electrons to be thermally emitted. Argon is passed to the chamber where plasma is ignited by DC discharge from the heated filaments. The coils surrounding the chamber create a magnetic field which will help to focus the plasma onto the substrate. The technique is based on a low potential (+1V), high discharge current (\sim 20 –70 A) to avoid ion damage and increase mass transfer to grow epitaxial films [10][11].

A low energy, high density DC Argon (Ar) plasma is used for efficient decomposition of precursors present within the chamber. Later lower energy ion bombardment is applied for removal of hydrogen from the substrate and further preventing of absorption of the H₂ present within the chamber, which is the major issue for other CVD techniques and affects the quality of the layer.

A diagram of an LEPECVD tool with all the main elements labelled is shown in Figure 24.



Figure 24 A diagram of an LEPECVD chamber with the argon plasma distribution controlled by the anode plate and the wafer's location. Reproduced from [10].

The effective dissociation of gases SiH₄ and GeH₄ used for Ge and Si_{1-x}Ge_x growth is crucial to maintain high-quality growth at low temperature (around 500 °C for Ge films) [10][12]. The ratio of gases for the growth of Si1-xGex is also crucial to maintain the defined composition of the growth layer. The difference in H₂ desorption rate in Si and Ge is no longer an issue using this technique, as opposed to PECVD techniques. The growth rate is independent of initial Ge composition or temperature of the substrate and is often faster than for CVD or PECVD methods with a reported value of 40 Å/sec [10] The deposition system allows to grow the wide range of thicknesses within the same reactor by modifying plasma density and gas fluxes (nanometer thick layers would be deposited with lower rate of 0.4 nm/s whereas thick layers can be deposited with the rate of 10 nm/s). The capabilities of LEPECVD has found multiple applications where precision and welldefined composition of SiGe are needed, e.g. for high mobility strained Ge channels, where precisely defined quantum wells and a thick buffer are realised during a single growth [13]. The possibility of growing high aspect ratio, self- assembled Si and Ge microcrystals was also developed by LEPECVD for deeply patterned Si substrates [14]. The results obtained from LEPECVD growth are very promising, making this method desirable to solve issues observed within heteroepitaxial growth e.g. the effect of the change in shape and therefore stress can be minimised due to the low temperature used and the lattice mismatch on the obtained film can be decreased due to the wellcontrolled SiGe composition (lattice constant) [14]. Nevertheless, LEPECVD is not widely

Methods

(commercially) available. Moreover, the tool requires extended maintenance and quality control to sustain high control of the process. For LEPECVD growth, the TDD value of 1.1×10⁵cm⁻² and RMS rough-ness of 32 Å for pure Ge on Si via a graded buffer layer was reported by the LNESS group [10], which are both typically lower than the values reported using alternative growth techniques.

3.1.1.2 Thermal oxidation

As mentioned previously, SiO_2 can be obtained from various deposition methods, nonetheless thermal oxidation is preferred for some specialized cases. Furthermore, thermally grown SiO_2 has higher density, higher uniformity, higher breakdown voltage, lower roughness and slower etch rate in HF compared to CVD grown SiO_2 [15].

The thermal oxidation of Si can be described by the Deal-Grove oxidation model [15]. The initial stage of thermal oxidation requires access to bare Si substrate and then when all the Si substrate is covered with the first layer of thermal SiO₂, solid state diffusion occurs through this layer creating a thicker oxidized layer. The interface between SiO₂ and Si is stable and reproduceable. The thickness of the consumed Si to grown SiO₂ is in a 0.46 ratio, which means 2.17 μ m of SiO₂ is grown per 1 μ m of consumed Si. The thickness of the layer is well defined which allows well controlled processes even for thin layers, an example of a thermally grown SiO₂ layer on a Si substrate is shown in Figure 25 [1]. The oxidation of Si occurring without additional thermal processing is known as thermal growth of native oxide [15] and it must be considered when access to the underlaying layer is especially critical.



Figure 25 SiO₂ layer thermally grown on the Si substrate against the Si level.

Thermal oxidation is a high temperature process, which requires a tube furnace and a quartz carrier for samples/ wafers. This type of system is compatible with high temperature growth thanks to the high melting point of the quartz glass (1710°C) [16]. The temperature used for thermal oxidation of silicon is usually in the range of 1000-1200 °C or 800 °C for thin SiO₂. The quartz tube contains multiple inlets supplying gases uniformly across the tube and a wafer carrier with a slow and accurate control of the heating zones (e.g. +10 °C per minute) to control the growth and uniformity accurately [17]. Gaseous oxygen (O₂) or water/ steam reacts with the Si surface creating amorphous SiO₂. Dry oxidation is usually much slower but produces very uniform, dense layer. After the first

layer is formed, diffusion of O_2 towards the underlying Si occurs if the temperature of the furnace is high enough for the continued creation of SiO₂[1].

The reaction for dry oxidation can be written as follows:

 $Si(solid) + O_2(gas) \rightarrow SiO_2(solid)$

For wet oxidation, O₂ is replaced by steam/water and the reaction with a Si substrate is written as:

 $Si(solid) + 2H_2O(gas) \rightarrow SiO_2(solid) + 2H_2(gas)$

The system used for thermal oxidation is shown in Figure 26.



Figure 26 The diagram of the system used for the thermal oxidation [17].

Wet oxidation is chosen for higher thickness of the material, where growth rate control accuracy is not as crucial, whereas dry oxidation can be used for lower thickness for high control [17]. For the fabrication purposes, the dry oxidation was used as only thickness under 100 nm of SiO_2 was required.

3.1.2 Lithography techniques

Lithography is a vital process used in microfabrication to pattern substrates (either wafers or samples). Moreover, the accuracy of the design and all further processing heavily relies on the performance of the lithography making this process one of major areas of improvement or a possible bottleneck of the microfabrication [18]. Usually, multiple lithography steps are required in a fabrication process, with a particular emphasis on the fidelity of the pattern printed and the accuracy of the alignment between layers. After lithography, the substrate with the photosensitive material is etched (or other process such as ion implantation) and afterwards, the photoresist left on the surface can be removed allowing further processing.

For optical lithography techniques, the resolution limitation is a consequence of the Rayleigh formula written in Equation 5 [18].

Methods

$$\Delta x = \frac{k\lambda}{Na}$$
 (Equation 5)

where Δx - resolution of the system, k- process factor for specific optics, resist or process latitude, between 0.25 and 1 with typical value of 0.6, values can be confirmed experimentally for a set of known resolutions, Na- numerical aperture of the lens, defined as sine of angle of incidence, λ illumination wavelength.

The lithography methods used for this work are: photolithography – UV contact lithography (6" wafers) or DUV projection lithography (8" wafers), and electron beam lithography (6" and 8" wafers).

The visualisation of lithographic nodes and processor technology leading to 13 nm technology is shown in Figure 27 based on data from Zeiss [18].



Figure 27 The correlation between resolution of the lithography and performance of the resulting microprocessors [18].

The 436 nm technology is used in contact photolithography, whereas the 248 nm technology is corresponding to the scanner used for fabrication within the Southampton Nanofabrication Centre.

The attention of researchers and industry throughout last decade was focused on crossing the barrier of 10 nm resolution and defining new limits of CMOS processing. The e-beam lithography is able to produce devices with resolution in range of a few nanometres, however, the optical lithography with lower cost is a tool of choice when the resolution is not critical.

The current capabilities of lithography allow fabrication with sub-10 nm resolution, mainly thanks to the application of immersion systems, double patterning technology, reticle enhancement techniques or, finally, applying extremely short wavelengths (13.5 nm) in techniques called Extreme ultraviolet lithography (EUVL) [19]. Also, next-generation lithography (NGL) based on even shorter

wavelengths for the X-ray lithography (XRL) (1 nm) [20] and deep X-ray lithography (DXRL) (0.1 nm) are under development.

3.1.2.1 Optical lithography

Ultraviolet (UV) lithography is a method where photosensitive polymer materials are spun onto a substrate and exposed to a wavelength of 436 nm (emitted by a Mercury Hg lamp) through a patterned chrome mask deposited on either fused silica (QZ) or soda-lime (SL) glass [21].

To perform successful lithography the following steps are most often necessary:

- 1) Substrate heating or surface preparation
- 2) Spin coating
- 3) Pre-exposure bake (softbake)
- 4) Mask alignment
- 5) Exposure
- 6) (Optional) Post-exposure bake
- 7) Development
- 8) Hard bake before next processes step

The preparation of the substrate involves baking or priming with chemicals. Thermal treatment in an oven or on a hot plate is mostly desirable to improve the adhesion to the substrate by dehydration and cracking the -OH bonds on the surface_[22] Dehydration can be also performed with acetone and isopropanol, whereas improving adhesion requires chemicals like HMDS (Hexamethyl disilazane vapor) as an example of an adhesion promoter.

The prepared substrate can be covered with light sensitive polymers known as photoresists using a spin coating or spray technique; the thickness of the resist is a result of the resist viscosity and spin process settings.

Due to the possibility of having the solvent still present in the coated material, the substrate should be baked again either in an oven or on a hot plate with parameters (time of baking and temperature of baking) adjusted to the kind of resist and its thickness [22].

The UV exposure is a vital part of the optical lithography where light is initiating a plethora of chemical reactions across various photoresist types [15][23]. The structures are either exposed or covered by the chrome mask, depending on the type of resist or the field of the mask (dark or light field for varied coverage of the mask). To sum up, after UV exposure, positive photoresist becomes

Methods

soluble in the developer when exposed; while negative photoresist is soluble in the developer when unexposed.

Some resists need to be post-exposure baked to finish the photopolymerization processing initiated during the exposure [15]. This is the case for AZ2070 (as an example) where post exposure bake is a requirement, which is often the case for negative resists. Photoresist is then developed using a chemical which is different for each resist.

A schematic of a lithography process with subsequent etching for either positive (e.g. S1813) or negative (e.g. AZ2070) resist is shown in Figure 28 [15].



^{Figure 28 Lithography process for positive and negative type of photoresists. 1) The substrate (e.g. Si wafer) with photoresist on the top, 2) photomask placed above the substrate covered with photoresist and affected by the UV lamp, 3a) development of structures covered with negative resist, 4a) etching for structures covered with negative resist, 5a) negative resist removal, 3b) development of structures covered with positive resist 5b) positive resist removal.}

One of the advantages of contact photolithography is the low cost of processing and accessibility, however, the pattern resolution and alignment between layers are limited to 0.5 or 1 μ m [21][24] respectively. Given these limitations, predominantly the feature size limitation, contact lithography is not suitable to integrate photonic waveguides and either e-beam lithography or deep UV projection lithography must be used.

3.1.2.2 Deep ultraviolet lithography

Deep UV lithography (DUV lithography) is an optical projection type lithography technique where the pattern on a reticle/mask is projected to the wafer surface through an optical system. The DUV lithography technique is called an industry "workhorse" [25] and is used as a high-throughput technology suitable for large area and high accuracy volume production where traditional optical lithography does not meet expectations for minimum critical dimensions and alignment tolerances.

The wavelengths used for DUV lithography are 248 nm obtained using a Krypton Fluoride KrF laser or 193 nm obtained using an Argon Fluoride ArF laser. A diagram showing an example of a projection lithography system is shown in Figure 29.



Projection Lithography

Figure 29 Principle of the projection lithography [26].

The DUV lithography available in the Southampton Nanofabrication Centre is based on the 248 nm KrF laser and allows to obtain features with resolution down to 150 nm [27].

Methods

3.1.2.3 Electron beam lithography

Electron beam lithography (EBL) is a powerful technique for creating nanostructures beyond the capabilities of conventional photolithography with possible resolutions of a few nanometres [28][29]. The Jeol JBX 9300 e-beam writing system in the Southampton Nanofabrication Centre (SNC) is capable to achieve linewidths of 8 nm with a minimum beam diameter of 2 nm [24].

The technique doesn't require physical optical masking, and patterns are exposed on the polymer by moving an electron beam over the sample following a digital template. The electron source is commonly created through hot Tungsten/ Zirconium (W/ZrO2) filament used to obtain field electron emission, which are later focused either through electric or magnetic lenses. The preparation of the sample is similar to optical lithography and contains preparation of the surface to improve adhesion (e.g. additional chemicals before the resist) and to remove any unwanted residuals from the surface by heating. The prepared substrate is spin coated with either positive (ZEP or PMMA) or negative resist (AZ® nLOF 2000 Series), afterwards espacer acting as a conducting layer to remove the electrons and prevent charge build-up can be added. The resist is exposed to the electron beam interacting with the polymer (resist). The exposed areas of the resist are affected by the chain scission which is the degradation of the main chain of the polymer. As a result, the solubility of the resist changes within specific regions across the wafer enabling selective removal of unexposed regions using a developer [30]. The sample is scanned across the features with the electron beam patterning sample accordingly to the design delivered usually in the CAD file (.gds file) and processed by the Beamer software to deliver patterning. Due to easily changeable design which is replacing traditional lithographic mask, the cost and time between fabrication steps (mask order etc) can be decreased when the process can be conducted in a chip scale. The flexibility of the design is especially convenient for multiple variations of single processing operations on the design like during tests. After the exposure the substrate is developed with designated chemicals and ready for further steps like etching. The espacer is removed with water only before the development.

The e-beam technique provides very high resolution; however, limitations of this technique are also significant. The cost of the maintenance is high, and time of the exposure can be long for more detailed structures with time of exposure in several orders of magnitude longer on a large-area substrate compared to optical lithography [8]. Dimensions and accuracy of the pattern transfer from a file onto substrate compared to optical lithography [31]. Dimensions and accuracy of the pattern transfer from a file onto a substrate depends on either ebeam or resist/ development (thickness, time of development) parameters. Almost two decades ago, the optimized e-beam processing for standard PMMA resist was used to fabricate lines under 5 nm [32] [33]. Higher speed

(energy) electrons are needed to maintain higher resolution, whereas for lower speed (energies of electrons), energy in range 0-5 eV, the resolutions drop down to 100 nm [34]. The foundation for the energy and resolution dependence is the number of interactions causing scattering between electrons and the resist [35].

3.1.3 Etch techniques

Etching techniques are used for selective removal of Si, oxide/nitride/metal layers after they are exposed using one of the patterning lithography techniques. The "masking" material used to protect the features should not be significantly etched during the process to avoid affecting the shape of the pattern and to enable the required etch depth to be reached. Usually, the masking material is photoresist remaining after the lithography on the top of the wafer or a hard mask adjusted to the etched material and not sensitive to the etchant. After the etching, the mask is usually removed using either O₂ plasma ashing or another adequate etching technique (tailored resist stripper). Two major types of etching are dry etch (e.g. inductively coupled plasma ICP or Reactive Ion Etching RIE) and wet etch where a liquid phase etching solution is used. Depending on the design and technique used, it is possible to obtain different etch profiles as shown in Figure 30.



Figure 30 Etch diagrams: a) angled etch, b) isotropic dry etch, c) anisotropic etch. Grey colour corresponds to the substrate e.g. Si, whereas blue colour corresponds to resist or hard mask.

3.1.3.1 Dry etch

The Inductively coupled plasma etch (ICP) dry etch technique was mostly used in this project, however RIE etch was also investigated.

3.1.3.1.1 Inductively coupled plasma etch

The ICP technique can used for selective etch of Si, SiO₂ or Ge, usually with the assistance of either the resist layer on the top of the substrate or a hard mask supported by suitable recipes. ICP etch is a dry etching technique relying on RF currents (typical value 13.56 MHz) induced in electrodes (coil shaped) surrounding the chamber as shown in Figure 31. A strong RF current induces a

Methods

magnetic field, which subsequently induces an alternating electric current within the chamber, dissociating the gases into a plasma of ions and reactive neutral elements (highly reactive but not yet in a form of ions) [2].



Figure 31 A schematic diagram of a standard ICP system used for dry etch [23].

Due to separated RF and ICP generators, the independent control of ion energy and ion density allows flexibility of the process. The ions obtained as a result of the plasma's creation take part in physical etching whereas the reactive neutral elements take part in chemical etching. A common chemical reaction for dry etch of SiO_2 by gaseous CF₄ can be written as following:

$$SiO_2(solid) + CF_4(gas) \rightarrow SiF_4(gas) + CO_2(gas)$$

3.1.3.2 Wet etch

Wet etching is used in microfabrication to remove layers or to etch structures within layers chemically, resulting in a very selective process (between materials), but an isotropic etch profile. Commonly, wet etching is conducted with diluted hydrofluoric acid HF for SiO_2 or tetramethylammonium hydroxide TMAH for Si.

The choice of the etching method is based on the material to be etched, available hard mask or resist, the depth and the desirable profile of the trench within the material. All these profiles can be desirable or avoided based on the purpose of the processing.

The example mechanism of wet etch can be described using reactions like the below reaction between SiO_2 and diluted HF as written:

 $SiO_2(solid) + 6HF(liquid) \rightarrow H_2SiF_6(liquid) + 2H_2O(liquid)$

The etch rate for 20:1 HF is varied between 30 - 80 nm / min for thermal or PECVD SiO₂ respectively; (based on the etch rate measurement maintained in the cleanroom facilities, depending on the precise concentration of the etchant). The chemical doesn't attack the resist even after several hours and high concentration like 50% [24].

The issues during the wet etch may be linked to the lack of adhesion of the resist to the surface of the substrate which is shown in Figure 32 for poor adhesion and for sufficient adhesion between layers.





When etching of the entire SiO_2 layer is required, which is the case for the removal of the capping layer or the hard mask, for example, the HF etch is used as a standard operation without any photoresist on the top of the wafer.

The Si etch is often conducted by dry etch, however, the angled etch or undercut is desirable for some applications e.g. suspended waveguides or membranes [25,26]. Strongly alkaline aqueous solvents like KOH, NaOH or TMAH are suitable for angled Si etching. In Figure 33, the structure of TMAH with OH- anion is illustrating the alkaline character of this component.
Methods



Figure 33 TMAH structure.

The mechanism of the TMAH etch for Si can be described using the general reaction for alkaline solvents as written in the following reaction:

$$\begin{aligned} Si(solid) + & 2OH^{-}(liquid) + & 2H_2O(liquid) \rightarrow Si(OH)_4(liquid) + & H_2(gas) \\ & \rightarrow & SiO_2(OH)_2^{2^-}(liquid) + & 2H_2(gas) \end{aligned}$$

The directionally TMAH etched structures within Si substrate are shown in Figure 34.



Figure 34 The diagram of directional etch and SEM images of etched structures [27].

As shown in Table 3, the TMAH is characterized with a good etch ratio for Si structures oriented (100/111) and (110/111) needed to obtain angled trench formation with atomically flat surfaces. Lower etch rates of SiO₂ and Si₃N₄ compared to Si allow the use of these materials as a reliable hard mask. However, due to this low etch rate, a short HF dip is always needed before the TMAH based processing as any native SiO₂ could form a masking layer preventing Si etching. Time of the HF dip should be adjusted to the expected thickness of the layer and confirmed etch rate of the HF solution, for the fabrication described in this thesis the time was set to 5 s in HF 20:1.

Table 3 TMAH properties for a wet etch reproduced from [24].

	Etchant	Etch rate	ratio for Si	Etch rate (absolute) [nm/min]				
	ТМАН	(100)/(111)	(110)/(111)	Si (100)	Si_3N_4	SiO ₂		
(25%,80°C)		37	68	0.3-1	<0.1	0.2		

3.1.4 Chemical Mechanical Planarization (CMP)

Chemical Mechanical Planarization or Polishing (CMP) are processes which either smooth or planarize the surface of the substrate. In the age of constant demand for decreased size of features across the chip and demands for novel lithography techniques, global planarization is a necessity [40]. CMP is widely used for III-V compound semiconductors and silicon photonics (SiO₂, SiNx and SiGe/ Ge layers).

CMP combines simultaneous chemical etch and mechanical abrasion by abrasive particles to secure advantages and minimise undesirable effects of these processes separately. The surface of the polished substrate (chip or wafer) is affected simultaneously by high pressure and chemicals which allows effective planarization from a starting point of varied thicknesses of the material across the sample. The regions with thicker material on the substrate can be subjected to higher pressure of the polishing pad enabling higher planarization rate comparing to thinner layer or material within trenches on the same substrate [41]. This mechanism embraces the uniform planarization of the material on the top of the wafer.

The principles of the CMP process are shown in Figure 35.





Figure 35 The process of CMP reproduced from [42].

The performance of CMP can be defined by the uniformity and the level of control of processing. Uniformity of the wafer can be considered on a multiple level including: WID (within die), WIW (within wafer) and WTW (wafer to wafer) planarity/ nonuniformity.

The uniformity and the rate of material removal depends on multiple conditions with special importance of the specification of the polishing system and the design.

Parameters of the processing like down force, platen and carrier speeds are confirmed to have an impact on CMP uniformity and removal rate [31]. The membrane installed between the rotation carrier and the wafer is responsible for applied down force on the back of the wafer [13]. Increased down force and faster platen or carrier speed increase the removal rate [14].

Parameters of the processing like down force, platen and carrier speeds are confirmed to have an impact on CMP uniformity and removal rate [31]. The membrane installed between the rotation carrier and the wafer is responsible for applied down force on the back of the wafer [43]. Increased down force and faster platen or carrier speed increase the removal rate [44].

Parameters of the substrate like design (pattern density) [45] and substrate dimensional characteristics, including substrate total thickness variation (TTV) and wafer shape (bow and warp) [34][46] should also be considered.

A plethora of applications of CMP rely on the selectivity of the slurry. Highly selective slurries are required to achieve tuneable selectivity for systems without a slurry sensitive stop layer.

On the contrary, the situation with required near-perfect planarity requires slurry with no selectivity to secure smoothness within the same level of various materials. The complexity of the polished system might also require multi-step processes with more than one type of slurry used for each step [44].

Besides selectivity, other properties of slurry are also of high importance. The concentration and additional components like H_2O_2 were researched for Ge designated slurry [32] [33]. The removal mechanism for CMP for Ge depends on the pH of the system due to the chemistry behind Ge creating chelate complexes with varied solubility. The polishing highly relies on this principle and can be described with various reactions for a whole range of pH [47].

The initial step is an oxidation of Ge on the top of the substrate and it can be described with following reaction:

$$Ge(solid) + 2H_2O_2(liquid) \rightarrow GeO_2(liquid) + 2H_2O(liquid)$$

The processing in acidic and neutral pH < 8 is resulting in creation of non-soluble in water Ge hydroxides like Germanic acid:

$$GeO_2(liquid) + 2H_2O(liquid) \leftrightarrow Ge(OH)_4(liquid)$$

For pH in range between 8 and 11 the dissociation of GeO_2 occurs more effectively as written in the following reaction:

$$GeO_2(liquid) + 2H_2O(liquid) \leftrightarrow Ge(OH)_3^-(liquid) + H^+(liquid)$$

The highest removal rate was reported for pH > 11 [48] when highly soluble Ge based anions are formed as written in the following reaction:

$$GeO_2(liquid) + 2H_2O(liquid) \leftrightarrow GeO_2(OH)_2^{2-}(liquid) + 2H^+(liquid)$$

The most commonly and commercially available slurries for Ge CMP are G1000 and W2000, used usually either for low or high pattern density respectively [49].

It is worth mentioning that a decreased TDD for Ge layers grown by UHVCVD was demonstrated for buffered SiGe systems after planarization [50]. The thicker buffers are linked to higher stress observed on the top of the layer; hence the planarization is used to remove the threads in pileups to relieve the strain introduced in the subsequent growth and minimize the nucleation of new threading dislocations [50].

3.1.5 Rapid Thermal Annealing

Rapid Thermal Annealing (RTA) is a heat treatment used for multiple processes including RMG, densification or/and dehydrogenation of layers, and dopant activation [4]. A Rapid Thermal annealer contains a chamber with heating tungsten W filament lamps mounted usually on the top of the chamber, above a stage with quartz holders to support the wafer. The maximum rate of heating up the wafer substrate is 100°C/s, which is controlled either by the thermocouple in direct contact with the backside of the wafer for lower temperatures or by measurement of the infra-red

Methods

emission by a pyrometer at higher temperatures. The temperature range and accuracy are specified for each type of thermocouple [51] and depend mostly on the metals used for the junction within the thermocouple. The heating zone is created above the wafer, whereas the walls of the chamber are kept at a constant temperature using process cooling water flowing around the chamber. The parameters of the process like temperature or lamp power are programable by a PID (Proportional, Integral and Derivative) loop control system based on collecting data from the measurement device, e.g. the temperature or power sensors, and constantly adjusting the response.

The RTA system is used for samples with tailored Ge or SiGe structures placed on the Si carrier wafer. It's worth mentioning that the relatively small volume and different absorption properties of the Ge surrounded by insulator might cause higher temperature within the Ge structures than measured either by pyrometer or by thermocouple attached to the back of for Si wafer. Moreover, especially for the measurements done by the thermocouple, an additional thickness of the carrier wafer is a reason for a considerably different temperature between that measured and that within the top Si layer on the SOI sample or Ge based structure. The difference in thermal conductivity between Si and SOI samples can also be considered as significant and enough for a temperature variance between platforms for a given lamp power. The range of anneal temperatures possible for both platforms is between 400 and 1200°C with varied time in the range 1 second and 30 minutes [4]. An ambient gas (e.g. hydrogen or nitrogen) can be pumped into the chamber to replace air. The gas is supporting the RMG process by adding an additional pressure to ensure encapsulation by the insulator cap on top of the melted structures to prevent these structures from balling up [4]. Gases used during RTA impact properties of the surface like e.g. roughness with advantage of H_2 over N_2 [4] [51]. Nevertheless, an N_2 flow was used for this project because H_2 is not supplied to the tool and the surface of the Ge is not exposed anyway.

RTA can also be used for an efficient removal of H incorporated within SiO_2 layers as confirmed in research on SiH_4 based SiO_2 films in a few temperature ranges as demonstrated in [52]. The efficiency of the hydrogen removal depends on the time and the temperature of annealing. Hydrogen concentration as a function of the depth from the free surface can be measured by secondary ion mass spectroscopy (SIMS) for a range of anneal temperature as shown in Figure 36.







Hydrogen removal matters for the RMG process when crystalline SiGe or Ge structures are formed in a temperature high enough to allow gas to leave the SiO₂ or Ge layers. If the dehydrogenation processing is not performed before annealing then the gas escaping these layers can create voids and decrease the consistency of the resulting structures. The bow of the wafer (caused by adding highly strained layers to one surface of the wafer) influences CMP or wafer bonding, so it is important to optimize the annealing process in order to remove a sufficient amount of hydrogen and at the same time avoid the negative effect of thermal processing on the wafer's bow. Annealing at 700 °C for 30 minutes was used for SiO₂ layer densification and H removal resulting in a thickness decrease of 5% for a 50 nm thick SiO₂ film [4].

3.2 Characterisation

Analysis techniques used for the material characterisation can be listed as follows: Scanning Electron Microscopy (SEM), Focused Ion Beam (FIB), Atomic Force Microscopy (AFM), Raman spectroscopy, reflectometry, profilometry and ellipsometry.

3.2.1 Material thickness and profile

The thickness control of layers is crucial for design and fabrication, especially for steps such as deposition and etching. The techniques used to measure layer thickness during this project were spectroscopic reflectometry and spectroscopic ellipsometry; both methods are non-destructive and contact-free, which is desirable for further processing.

Methods

One of methods used to determine not only thickness but also optical (e.g. complex refractive index) and dielectric properties is Spectroscopic Ellipsometry (SE), where the range of measured thicknesses is usually in the range of 1 nm to a few μ m [53]. Ellipsometry can be used to measure most material types: dielectrics, semiconductors, metals, superconductors, organics, biological coatings, and composites of materials [54]. The technique is highly popular due to its sensitivity and a chance to obtain quick *in-situ* measurements.

The basis of spectroscopic ellipsometry measurement is the reflection and change in the polarisation of the reflected light after contact with the sample due to the changeable Fresnel reflection coefficient with the wavelength. A diagram describing the principle of the measurement is presented in Figure 37, where p-polarized light has an electric field direction parallel to the plane of incidence of a device, and s-polarized light has the electric field oriented perpendicular to that plane.





The polarization change ρ is represented as reflection parameters like an amplitude ratio, Ψ , and the phase difference, Δ , and can be express as equation 6.

$$\rho = tan(\Psi)e^{i\Delta}$$
 (Equation 6)

For fabricated structures, the obtained response was compared with the model library and the fitting set of layer parameters is calculated by the software of the tool M-2000 Ellipsometer from J. A. Woollam with targeted accuracy of the tool even as good as 0.01 % [53]. The accuracy of ellipsometry is enhanced by measuring the ratio of parameters instead of their absolute values.

Ellipsometry has limitations due to contamination or oxides present within layers of materials; the surface roughness should also be considered as a factor disrupting measurements.

Systems based on Ge with underlaying SiO_2 can be inspected on both Si and SOI substrates using Ellipsometry. However, the thickness of the Si substrate will not be defined after the measurement,

which is important when trying to measure etch depth into the Si substrate. This drawback implies the necessity of profilometry as a supplementing tool for measurements.

Reflectometry can be used for measurements of transparent or semi-transparent layers up to three layers which is also a limitation for measurements of non-transparent layers like Ge. The reflectometry is designated for smooth surfaces, however models considering roughness can be applied to improve an accuracy of the measurement.

3.2.2 Profilometry and Atomic Force Microscopy

As mentioned, profilometry is a technique complementing ellipsometry or used when other techniques are not enough to define the topography of the wafer, thicknesses of layers or differences within the thickness of the same layer (e.g. trenches etched within the Si wafer). Profilometry can be either light or stylus based. The stylus-based tools are usually slower and physically touch the surface. The probe touching the surface can damage the sample or introduce contamination. The result of measurements can be a single point, a line scan or even a full three-dimensional map.

Another technique used to define topography of a sample is Atomic Force Microscopy (AFM), especially useful to define surface roughness after processes like etching, deposition or CMP. The advantage of this technique is a much higher resolution in the z-direction. How-ever, only small areas can be mapped.

3.2.3 Raman Spectroscopy

Raman Spectroscopy is a non-destructive technique widely used for nanomaterial characterisation through a variety of disciplines from biology to optoelectronics [55]. The data obtained from this technique allow the determination of multiple properties of a material including: chemical composition, structure, phase, polymorphism, impurities or strain [56]. Based on the characteristic Raman spectrum, it is possible to identify the material with the help of extensive Raman spectra libraries.

This technique relies on a laser light source to irradiate a sample and generate Raman scattered light, which is detected as a Raman spectrum by a CCD camera. The resulting spectrum provides a characteristic pattern used to identify substances and confirm their properties.

An example of a Raman spectroscopy setup is presented in Figure 38; moreover, this specific tool was also used for measurements of $Si_{1:x}Ge_x$ composition carried within this project.

```
Methods
```



Figure 38 Schematic diagram of a Renishaw Raman microscope [57].

The working principle of Raman spectroscopy is the interaction between the laser photons and molecular vibrations, phonons or other excitations in the measured sample. The change of the energy between vibrational states is presented in Figure 39 for both elastic and inelastic scattering.



Figure 39 The energy diagram for processes occurring from the interaction of photon with a molecule; a) inelastic Stokes Scattering with the final energy level higher than the initial one, b) elastic Rayleigh scattering which implies the same energy level at the end and the initial level of the interaction, c) inelastic Anti-Stokes Scattering with the final energy level lower than the initial one [38,39].

The change in energy corresponds to the shift of the energy which can be presented by the change in wavelength as shown in Equation 7.

$$\Delta \omega = \frac{1}{\lambda 0} - \frac{1}{\lambda 1}$$
 (Equation 7)

I

where $\Delta \omega$ is the Raman shift, $\lambda 0$ is the incident wavelength and $\lambda 1$ is the scattered wavelength.

The penetration depth of the Raman measurement is highly dependent on the wavelength of the laser, wavelengths of 532 nm, 633 nm, and 785 nm can be used to probe varying depths of material, including Ge and Si, which is shown in Table 4.

Table 4 The penetration depth of light of various wavelengths into Si and Ge [4][59].

	Laser wavelength [nm]							
	532	633	785					
Penetration depth in Si (nm]	980	2650	8930					
Penetration depth in Ge [nm]	17	53	179					

The numerical aperture of the lens used during the measurement and various lenses result in different spot size diameters as shown in Table 5. It should be mentioned that these values are valid for straightforward optical microscopy and are only an approximation for more complex Raman microscope.

Table 5Spot size radius for a choice of lasers and magnifications [4].

Magnification	Numerical	Laser 532 nm Laser 633 nm Laser 78			
	aperture	S	opot size radius [μm]	
x5	0.12	2.71	3.22	4.00	
x10	0.4	0.81	0.97	1.20	
x20	0.75	0.43	0.51	0.64	

Both these factors, penetration depth and the size of the spot, influence the Raman measurement and accuracy of the data especially when the measured structures are either on the top of the substrate or embedded into the substrate with a different material surrounding the structure. Sizes of the structures relevant for this thesis are varied in the range between 100 nm to 10 μ m, and with the range of alloy composition varied between pure Si and pure Ge. Application of the Raman Spectroscopy technique for the composition measurement of Si, Ge and Si_{1x}Ge_x is based on the ratio between Ge-Ge and Si-Ge peaks and requires well defined, sharp peaks, which were achievable with the use of a high-density grating (3000 lines/mm) designated for the 532 nm laser

Methods

I

used for this project. The power of the laser was set to minimum value to avoid heating which can melt material and change the local composition.

The Ge and $Si_{1:x}Ge_x$ composition can be indirectly defined via Raman spectroscopy as peaks characteristic for Si-Si, Ge-Ge and Si-Ge can be observed on the Raman spectrum. An example spectrum is shown in Figure 40.



Figure 40 An example Raman spectrum for a SiGe structure obtained from the RMG processing. The asymmetry of the peaks can be minimised by decreasing the power of the laser and extending the time of measurement.

The Raman shift of single crystal Ge is measured to be 300.7 cm⁻¹, whereas for singe crystal Si it is 520.2 cm⁻¹; a full width at half maximum (FWHM) of 5.3 cm⁻¹ is reported for Ge and 4.6 cm⁻¹ for Si [59]. The Si-Ge peak can be observed between 390 cm⁻¹ to 410 cm⁻¹ [4]. The strain and the temperature can affect the position of peaks, but for the composition's calculations described in this thesis are based on equation 8 relating only to the intensity of Ge-Ge and Si-Ge peaks. This methodology of defining the Si_{1-x}Ge_x composition is based on using the Mooney equation [60] (equation 8):

$$\frac{(I(siGe))}{(I(GeGe))} = \frac{(2(1-x))}{kx}$$
 [Equation 8]

Where, x is Ge concentration to be calculated, k is a constant (1.2), I(SiGe) describes the integrated peak intensity for Si-Ge, and I(GeGe) describes the integrated peak intensity for Ge-Ge obtained from the spectrum, as shown in Figure 33. The constant k can be found from measurements of known composition samples and depends on the excitation wavelength.

The MATLAB software was used to fit Lorentzian curves to each peak visible on the Raman spectrum. The Raman peaks for alloys usually deviate from the Lorentzian fitting, hence areas of

the peaks were integrated for both cases, Lorentzian fitting and raw data, to determine errors. According to the Money equation, the error in the final composition can be given as a propagated error for the equation for the ratio of peaks. The error given for each peak was defined as a percentage difference between the area under the peak for raw data and the area under the peak fitted with Lorentzian curve. All the errors were estimated to be within 5% for the data (up to ± 2.5 % for each data point), for the clarity of the graphs they are not included.

3.2.4 Scanning Electron Microscopy and Focused Ion Beam

The characterisation of materials in μ m or nm scale require specialized tools to provide information needed for further processing. In nanofabrication, both top view and cross-section imaging are of high importance.

Commonly used Scanning Electron Microscope (SEM) is a type of microscope producing images by using a high energy electron beam to interact with atoms of the inspected material. The gathered data can be used to determine materials thicknesses (especially for cross-sectioned material or side view), topography of the surface and composition. The technique used to deliver cross-section is Focused Ion Beam (FIB) based on a liquid metal ion source (e.g. gallium (Ga) ion source) which is accelerated and focused by an electric field [61]. Many powerful systems rely on the collaboration between FIB and SEM, where the ion-beam system is used to mill the material and afterwards SEM is operated to image the facet of the cut. An example of such a combined SEM-FIB system is shown in Figure 41.





Figure 41 A combined SEM-FIB system with a Ga+ source where ETD, Everhart-Thornley detector; PtIN, platinum injection needle; SS, sample stage; CDM, continuous dynode electron multiplier detector; TLD, through-lens detector [62].

The topology of samples can be imaged by SEM using highly accelerated electrons, which interact in an inelastic way with electrons of the specimen's atoms. The response of the sample is collected and processed by suitable detector and presented as an image on the screen of a computer. Two main interactions of electrons in SEM technique are shown in Figure 42.



Figure 42 The diagram showing secondary electrons (SE) and Backscattered electrons (BSE) used for SEM imaging. K, L1, L2,3 correspond to electron shells around the atomic nucleus [63].

Secondary Electrons (SEs) will be emitted after inelastic interactions with the K shell of the atom whereas Backscattered Electrons (BSEs) will be elastically scattered with energy depending on the atomic number (Z) of the interacting atom. These kinds of interacting electrons play a key-role for SEM imaging; firstly, SE are especially beneficial for an inspection of the topography of the sample, secondly BSE can deliver information about composition of the specimen as the number of scattered electrons increase proportionally with Z number. The resolution depends on the excitation volume which results in high SE resolution reaching even the range of ~ 10 nm [62].

3.3 Summary

All described methods are used either for fabrication or characterisation of RMG samples. Some of methods (e.g. PECVD for first insulation layer) were used only on the stage of process development, however, leading to the solutions used for the final design. The fabrication was conducted in the wafer scale up to the RTA processing, annealing and characterisation (Raman spectroscopy and SEM) were conducted in the sample scale.

Design of mask and experiments

Chapter 4 Design of mask and experiments

This chapter provides a detailed description of the motivation behind the aimed SOI integration of Rapid Melt Growth (RMG) structures and an explanation of the design of the experiment and the process flow. The masks described here were used for process development and finally for obtaining the results described in Chapter 5.

4.1 Motivation for SOI integration

Rapid Melt Growth (RMG) on the SOI platform is a method of delivering multiple compositions of SiGe strips that can be used to fabricate devices e.g. modulators and detectors across the same chip or even within a single structure. The RMG process requires only a single Ge or SiGe deposition and a single anneal step. From an engineering point of view, said integration provides a way to minimize footprint while improving the performance of devices built in the RMG structures. The miniaturization is linked to the ability to create multiple types of devices within the same layer or even the same tailored structure. At the same time, the low number of defects results in a better device performance. The advantage of the RMG technique is also dramatically decreased cost of

fabrication when one anneal step can replace multiple deposition/epitaxy steps, which is the current solution for multiple SiGe composition across one wafer.

As explained in detail in Chapter 2, the properties of SiGe alloys are related to the Ge composition. The material engineering of SiGe structures is linked to the tunability of the band gap and the lattice constant where tailored RMG structures are engineered to tune the SiGe composition along a central structure. A typical application of SiGe material engineering is shown in Figure 43.



Figure 43 Material and band gap engineering of SiGe to vary the operational wavelengths of EAM modulators (images as an example design, based on work from the research group). The operational wavelength is dependent on the SiGe composition and devices would be incorporated into the structures as shown in the diagram. The composition 1 is designed for wavelength 1 based devices etc.

Devices like modulators or detectors built on RMG structures require waveguide integration. The main methods of coupling light into the photonics devices, e.g. detectors, are butt coupling or vertical (evanescent) coupling [1]. For evanescent coupling, the device is placed on the top of a waveguide, whereas for butt coupling, the device is in contact with a waveguide's facet. For many waveguide integrated applications, the waveguide material is Si and the material of photodetector or modulator is Ge or SiGe. Both, evanescent and butt coupling for Ge structures are shown in Figure 44.





Figure 44 Methods of coupling light from a Si waveguide into a Ge photodetector, a) evanescent coupling, b) butt coupling, reproduced from [2].

A comparison of butt coupling and evanescent coupling for Ge structures is provided with finite difference time domain (FDTD) simulations, which results are presented in Figure 45.





The FDTD simulations compare the light absorption within the Ge photodetector. The length required to absorb 95% of light in the evanescent coupled structure is 17 μ m, whereas the same absorption was achieved within only 7 μ m for the butt-coupled structure. Moreover, the thickness of the Ge structures used for evanescent coupling was 500 nm. In order to decrease the length of the device, a thicker layer would be required. A comparison of properties of butt or evanescent coupling were also provided by Silvaco and the results of the simulations are available in [3].

Although the evanescent coupling is easier to fabricate without an additional etch and careful alignment, which are required for butt coupling, the larger dimensions typically lead to a larger capacitance, and therefore, a slower device [1]. Surface planarity is also an advantage of butt-coupled structures, enabling more complex multi-layered systems.

4.2 Process flow

The fabrication of RMG structures on the SOI platform required modifications of the RMG process on the top of the Si platform which was described in detail in Chapter 2. The RMG structures suitable

for butt coupled integration within an SOI platform should be embedded within the Si overlayer and positioned onto the insulating layer to form a direct connection between the Si and the tailored structures areas. The final structures embedded into the SOI wafers were developed throughout multiple test runs. The final integration process of the RMG technique to form SiGe within an SOI cavity is shown in Figure 46.



Figure 46 Process for Rapid Melt Growth (RMG): 1) trench creation, 2) insulator deposition, 3) seed etch, 4) blanket Ge deposition, 5) planarization of the Ge layer to wafer level, 6) deposition of the encapsulation layer, 7) RTA, 8) cap removal.

The design with Ge/SiGe filled trenches should be planarized to the Si level which can be achieved with the Chemical Mechanical Polishing (CMP) technique.

The full process flow is attached in Appendix 1, Table 1.

4.3 Mask design

The design of masks for the RMG processing was based on the research of tailored "tree-like structures" on the Si platform [1]. Tree like structures are designated to be integrated with waveguides and used as a base material for active device like e.g. recently reported Franz-Keldysh modulator [2][3]. The active area of the devices reported within this research was within 1-1.5 μ m width and 40-55 μ m length. Consequently, the stripes of the proposed SiGe tailored structures on SOI platform are designed to allow integration of waveguides and active devices in size as these reported (1-3 μ m wide and 60 μ m long strips alongside with other structures).

Design of mask and experiments

Masks included strips and two main types of tailored ("tree-like") structures on SOI RMG as shown in Figure 47. The strips are needed to research gradually changeable SiGe composition on the SOI platform, whereas tailored structures with branches are required to achieve a uniform composition and for research on similarities to Si platform and further waveguide integration. The type 1 RMG and type 2 RMG structures can be distinguished based on the position of the strip of the tree-like structure (with added branches). For the type 1 RMG, an additional strip is added, leading from the seed towards tree-like structures. This design is dedicated for the waveguide integration through the strips led by the part with added branches (with desirably constant composition). The seed region was shifted outside the strip area to avoid complications of the waveguide's design while the seed contains all the defects so it cannot be used as a part of devices. The main strip of type 1 RMG can be linked to changeable amount of tailored structures to accommodate multiple SiGe compositions and consequently be a base for multiple active devices place.



Figure 47 Main types of RMG structures. (a) type 1 RMG for either single or multi- tree structure with the waveguide space designed for main strips of each tree, (b) RMG type 2 for test structures. Description of symbols used to describe design: SL- Seed length, SWseed width, SC- seed connection with the structure, L- length of the tree-like structure, w- width of the main trench of the tree structure (with branches), Bs- distance

between branches, Bw- width of the branches, BL- length of branches, D- distance between seed and the tree for type 1 RMG structures, D1- distance between the strip of the tree- like structure and the end of the main strip of structure, wa- width of the main strip for type 1 RMG, D2- distance between tree-like structures (branches) for multiple tree structures.

The design of the geometry of the structures included within this mask was based on the results obtained for the analogous structures on Si described in the literature [4, 5, 6] and former test development conducted within this project. The initial parameters were based on the Si results as a starting point. The added branches are responsible for cooling of the structures, which was described in detail in Chapter 2.

The masks presented in Figure 48 and Figure 49 were designed for the 6-inch platform for both Si and SOI wafers, where Si wafers are acting as test wafers for most of the processing. The general layout for 6-inch masks for e-beam and photolithography processing is shown in Figure 48. All masks are compatible with each other which allows integration of waveguides fabricated with e-beam lithography on layers previously patterned with photolithography.



Figure 48 The 6 inch mask template a) the whole template suitable for processing combining optical lithography and e-beam lithography for 34 cells (typical size for e-beam cell), b)

Design of mask and experiments

6 typical positions of e-beam alignment marks for a single cell, c) detailed e-beam alignment marks with x and y Vernier scales to measure alignment errors for multi lithography processing. Reproduced from [4].

The first step of the process was performed with either photolithography or e-beam lithography to pattern the tailored RMG structures and alignment mark layers. This design decreases the number of steps due to excluding additional mask with alignment marks only. E-beam lithography was followed by three photolithography; for dummy squares within the same layer as the RMG structures, Si seeds and openings of dummy squares and further Ge etch with protective layer of squares covering previously formed openings. The process flow is presented in Appendix 1 for the sake of clarity. These layers could have been fabricated with photolithography due to less critical shape and dimensions. Dummy squares are included across all the layers to support uniformity of processes such as: Ge/SiGe growth and CMP. The last layer with waveguides was designed for e-beam lithography.

The template of a cell is shown in Figure 41. All cells from the mask were identical which was preferable when uniformity across the wafer is of high importance for processes. Deposition, etch or especially CMP depend either on the uniformity of the layer but also on the position of the cell. Additionally, the Rapid Thermal Anneal (RTA) used to melt RMG structures was carried at the chip level to allow multiple parameters (temperature, anneal peak time, etc.) to be researched.



Figure 49 The template of the cell divided into section with various test structures for RMG processing. Sets 1-6, 13, 14, 16 are designed for waveguide integration, Set 7- variety of strips with access to Si seed, Set 8 – strips with branches, Sets 9-15 are designated for material tests for tailored parameters of trees.

Cells 1-6 were designed for waveguide integration by focused grating couplers which are used to ensure more efficient usage of the space across each cell. Nevertheless, the grating couplers are suitable only for a narrow wavelength range.

The parameters of cells 1-6 are presented in Table 6. Across all cells the type 1 RMG was used.

Table 6Parameters of sets for waveguide integration.

Cell	nA	D, D1 [µm]	W [μm]	aw [µm]	Bw [μm]	Bs [µm]	Comments
1	1	30,30,30;30,90,150	3	3	3	3	Repeated from cell 10
2	1	30,30,30;30,90,150	1	2	1	2	
3	1	30,30,30;30,90,150	2	2	2	2	
4	1	30,30,30;30,90,150	2	2	2	2	L={20,40,50,60,70} μm
5	1	30,30,30;30,90,150	1	2	1	2	Repeated from cell 2 (for Raman)
6	2	30,40,50; D2=30	2	2	1	2	L={20,40,50,60,70} μm
7	4	50, D2=60	3	1	1	3	

Design of mask and experiments

13	1	30,30,30;30,90,150	3	3	3	3	Repeated from cell 1 (for Raman measurements)

Cell 7 was designed to research the composition and maximum width of strips embedded into SOI to compare with previous work done on Si platform, where the maximum width of strips not affected by surface tension was limited to 5µm [1][4].

The example of sets included within Cell 7 is shown in Figure 50.



Figure 50 The example of data from the design of set 7; (a) Ge composition vs. seed connection, (b)_Ge composition vs. strips length/ Ge composition vs. normalised length.

Set 8 is designed for the comparison of type 2 RMG structures as shown in Figure 51.



Figure 51 Example structures type RMG1 from set 8; (a) Ge composition vs. strips length/ Ge composition vs. normalised length, (b)varying width of the main strip.

The designed experiments for Ge composition vs. strips length/ Ge composition vs. normalised length and Ge composition vs. seed connection were also done parallel to structures from set 7, where the branches were added with identical parameters of width, length and seed connection as these ones for basic strips. This is meant to confirm the cooling effect of the branches added to the strips for RMG structures on the SOI platform. Besides this, more parameters were compared as shown in Figure 52._The example sets designed to gathered data are also shown in Figure 52 for structures with 2 trees on one main strip.



Figure 52 Example type 1 RMG structures (number of trees, n=2) from set 12. Structures were designed to confirm e.g. an impact of the seed connection on the SiGe composition of main strip, the impact of the width on the composition and to confirm the importance of length of the cooling branches of tree-like structures.

All the sets of the structures were designed to collect the data for specific parameters like branch dimensions, seed connection or distance from the seed for structures embedded into SOI platform. Results are presented in Chapter 5.

4.4 Summary

Masks and process flow described within Chapter 4 were used for the final fabrication after process development runs. The plethora of designs (variations of dimensions) was included to research the RMG process on the SOI platform. The possible difficulties of the fabrication were identified

Fabrication of SiGe wires on SOI platform

(predicted) and resolved within the process development described in detail in Chapter 5; the final process flow was based on gathered solutions for undesirable effects.

Chapter 5 Fabrication of SiGe wires on SOI platform

This chapter describes process development leading to formation of constant composition SiGe wires on the SOI platform. One of the most crucial findings of this process development work has been linked to the importance of the shape of the trench filled with SiGe/ Ge on an insulating layer as well as the quality of the deposited Ge layer (from the growth and subsequent chemical mechanical polishing). Within the development of the RMG structures on SOI most issues were identified and solved without a substantial change of mask design.

5.1 Formation of angled Si cavities

The sidewalls of the cavities designed to be filled with Ge/SiGe on insulator were initially straight as shown in Figure 53. The process flow for structures embedded into the Si layer requires the planarization of the layer to the level of the insulator (the red line in Figure 53 indicates this level).



Figure 53 A cross-section of a Si trench filled with Ge showing a lack of material on the sidewall.

The lack of Ge in the bottom corners of the trench is clearly visible from the cross section of the trench. After planarization of these structures, the amount of Ge would not be enough to fill a whole trench after melting (this links to issues with the surface tension describe in Chapter 2 as the surface tension will cause the formation of rounded ends of Ge/SiGe structures or create rounded voids not filled with Ge/SiGe. Additionally, an insulating cap can collapse into voids not filled with material. Each of these issues might cause further complications and epitaxy breakdown resulting in a non-uniform composition across the strip.

One of the issues identified during the first process iteration was the lack of insulator between the Ge/SiGe and substrate layers on the bottom of cavities, especially in corners. This kind of discontinuity within the layer usually causes interaction between Ge and Si during the anneal step, which is undesirable outside of the seed area.

The solution was found by using a wet angled Si etch to shape trenches and subsequently prevent the lack of materials in corners. This modification resulted in more homogenous Ge layer and SiO₂ layer underneath enabling a continuous isolation/encapsulation of the SiGe strips or tailored structures. A comparison of dry etching (vertical sidewalls) and wet etching (angled sidewalls) is presented in detail below.

Fabrication of SiGe wires on SOI platform

To compare dry and wet etch results and Ge layer quality after both etch techniques, 500 nm Ge was deposited with a PECVD tool. The structures etched were strips obtained using a standard photolithography mask containing a variety of strip widths, lengths etc.

5.1.1 Dry etch

The dry etch was performed in an ICP tool using the following etch recipe:

- Pressure: 15 mT
- Temperature: 15 °C
- RF Power: 50 W
- ICP Power: 800 W
- SF₆ Flow: 25 sccm, C₄F₈ Flow: 45 sccm

The sidewall angle of the structure after this etch is around 90° with slightly wider angle for Figure 54a than Figure 54b which has an impact on the Ge layer thickness which is different for horizontal and vertical parts of the trenches.



Figure 54Dry etched Si sidewalls covered with Ge layer; (a) the slightly wider angle (approx. 5°
from vertical) and shallower trench compared with (b), where the difference between
filling of the corners and vertical/ horizontal part of the trench is more visible.

Though, it is possible to obtain the angled sidewalls by using a dry etch, the uniformity across the substrate and repeatability are not reliable enough for subsequent RMG processing.

5.1.2 Wet etch test

Uniformity across the wafer and repeatable results for etching were confirmed with TMAH angled Si etching. As previously for ICP etch, the wet etch process was carried out following deposition of

the SiO₂ and Ge layers. Due to the high selectivity of TMAH based solvents between Si and SiO₂, an HF 20:1 dip (5 seconds) for each sample was necessary before the TMAH etch. Wet etching was performed with 25% TMAH at room temperature (no additional heating) for 3 etch times: 30, 40 and 60 minutes. This concentration of TMAH at room temperature enables a relatively slow etch (theoretical etch rate around 8.8- 9 nm per minute for 54 °angle). Variation of the etch rate and sidewall angle are possible by varying the TMAH concentration.

For samples prepared for this test, the time needed to etch throughout the 400 nm Si overlayer (all samples were obtained from SOI wafer with 2 μ m buried oxide (BOX) and 400 nm Si overlayer) was calculated to be 45 minutes. After analysis of results, the etch rate was confirmed to be 9 nm/minute at room temperature and 45 minutes is exactly the time needed to etch 400 nm Si. The sample exposed for 60 minutes in TMAH was etched fully without a further undesirable undercut effect; the 30 minutes etched sample confirmed the possibility of control of the process in case a specific thickness of Si overlayer is required for further processing. The sidewall angle and etch rate are both accurate and repeatable enough for the requirements of this experiment. Images of the etched trenches are shown in Figure 55.



Figure 55 SEM images of the TMAH angled etched trenches filled with Ge; (a) 30 min TMAH etch, (b) 45 min TMAH etch, (c) 60 min TMAH etch.

5.1.3 Wet etch tests for tree-like structures

As mentioned previously, the lack of SiO₂ in the cavity corners can be indirectly identified during the Raman measurements of the RMG Ge layers or be visible by microscopy characterisation of the structures after RTA. The Si and Ge leakage from within the structures can be identified through FIB cross- section as intermixing between the Ge and Si in the corners as well as unexpected fluctuations of SiGe composition across the structures calculated after Raman analysis. Examples of non-continuous layers are shown in Figure 56.

Fabrication of SiGe wires on SOI platform



Figure 56 Tree-like structures obtained during process development for batch 1- after the RTA step, leading to identifications of possible issues linked to either photolithography or/and, described here, trench etching; (a) the rounded shape of not entirely filled trench,
(b) structural defects visible after SiGe melt- yellow arrows point the Ge and Si intermixing region due to non-continuous insulating layer between Si and Ge, whereas blue arrows indicate issues associated to photolithography and trench etching.

In Figure 56a, the border between Ge and Si seems continuous without the obvious leakage like the one visible in Figure 56b. It might be a proof of achieving the full continuity of the insulating layer or just a specific place without the intermixing issue as the insulating layer can be non-uniform across the structure and might not show signs of leakage along the entire length.

The TMAH etch was first tested on strips where the orientation was adjusted to directional (anisotropic) character of the etchant. The structures with corners (including tailored structures with branches) can also be affected by the directional etch.

The test structure obtained for test in Figure 56b was fabricated using photolithography, masking the SiO_2 layer for etching after photolithography, PECVD insulating layer to prevent intermixing between Si and Ge and TMAH angled etch for trenches. To resolve the addressed issues, the following development processes were necessary:

- Photolithography optimization to avoid rounded corners of the structures,
- The etch recipe tests to avoid the change of designed straight angles across tree-like structures. The TMAH etch is anisotropic which partially explains the shape of 'corners' visible in Figure 56b,
- Preventing issues on the interface between Si and Ge due to a non-continuous underlaying PECVD SiO₂ layer

Based on the results obtained from the initial stage of process development, further work on the creation of angled sidewalls was performed and will be referred as Process development Batch 2. Firstly, it should be mentioned that the thickness of the insulating layer covering the trenches was increased to 50 nm which doubled the value of an analogous layer from wafers in Batch 1. This change ensured improved coverage of corners, even for partially filled structures, and also prevented over-etching during the Ge deposition (which required HF etching prior to deposition to remove native SiO₂ from the seed area). Modifications of the angled etch for trenches are detailed in Table 7.

 Table 7
 The parameters for wafers from the development Batch 2. Wafer SOI1 was broken during CMP.

PROCESS	WAFER	ICP etch depth	TMAH etch depth [nm]	PECVD SiO ₂ thickness within the trench [nm]	PECVD SiO ₂ thickness around structures [nm]
1	SOI1, SOI2,	-	300	50	50
2 SOI3, Si5		300	<100	50	50
3	SOI4	-	350	50	50+70=130

The idea of process for wafers from Batch 2 is also presented in Figure 57. The main difference was related to the creation of the trench and modifying the insulator layer. The hard mask caused issues during wet etch and additionally, it had an influence on directional etch. The trench creation was researched for the design based on either TMAH directional etch or combination of wet and dry etch.

Fabrication of SiGe wires on SOI platform



Figure 57 The shape of trenches for various wafers from Batch 2; (1) wafers SOI1, SOI2, (2) wafers SOI3, Si5, (3) SOI4.

The results for test development batch 2 obtained from the fabrication described in Table 7 and in Figure 57 are shown in Figure 58.



Figure 58 Comparison of main strip and associated branches for after the CMP step for process (1) SOI2, (2) SOI3 and (3) SOI4, (3a) explanation of the undercut effect occurring in the corner of structure (3).

The general summary of structures fabricated with either TMAH only or mixed TMAH/ICP ensured an impact of the etch on the shape of the corners and the number of voids in the Ge layer. The voids visible in Figure 58 for process (3) are caused by undercut occurring through the subsequent HF and TMAH etching as explained in Figure 58 in diagram (3a). The HF dip necessary before the TMAH etch should be conducted carefully (not longer than 1 s dip and in-depth water cleaning afterwards) to avoid the presented effect. The way to prevent any impact of residuals from

undercut SiO_2 layers is full removal of this layer after etch and deposition of the new SiO_2 layer across the whole wafer. In Figure 59, the importance of the undercut effect and rounded shape of structures corners is explained further.



Figure 59 Measurements of the diagonal between opposite branches compared with minimum distance between these branches indicates significant difference and room for possible improvement of the final shape. Structure A is fabricated as mentioned described in process (3). Structure B is fabricated as mentioned described in process (2).

From Figure 59, one can observe that there is a significant size difference of more than 25% (1 μ m) between achieved and theoretical minimal size of crossing for widths of measured branches for Wafer SOI 3. It is worth adding, that if the gap between fins is too big, it can be a reason for surface

Fabrication of SiGe wires on SOI platform

tension appearing later on (during RMG) and affecting Ge structures, as was mentioned in the theoretical background chapter of this thesis. Further process development should be focused on bringing the achieved shape and size of the structures closer to the original design.

The coverage of the corners of the trench filled with Ge is visible in Figure 60. This result can be compared with the results initially presented Figure 53, where voids in corners were clearly visible.



Figure 60 A cross-section of the structures fabricated with ICP/TMAH etch; a) with the 5 μ m main strip from the Wafer SOI3, done after RMG step, b) a cross-section of the 2.5 μ m strip from Wafer Si5.

Based on the results from the development test performed for wafers from Batch 2, optimal processing for further creation of angled trenches and tree-like structures was chosen to be a combination of TMAH and ICP etch. The combined ICP/ TMAH etch for the Si layer led to the closest to the straight angle corner (from the top perspective). The ICP/TMAH etch depth ratio was 3:1 (300 nm to 100 nm Si) whilst the angle of Si etch was measured after FIB cross-section as 120°. The optimized etch brings the shape of the obtained structures closer to the originally designed shape and minimizes the number of voids within corners. These conditions can be successfully used for the Si etch for further fabrication. However, the limitations of optical lithography will most likely be visible as rounded corners or changed size compared to the design (depending on the development, exposure and resist used). Section 5.3 below focuses on the process development of a lithography technique suitable for the tree-like structures embedded into an SOI platform.

5.2 CMP

The RMG process for SiGe tree-like structures embedded into SOI wafers requires planarization to the level of the top of the Si, filled with either Ge or SiGe. The initial CMP process and training on

Formatted: Don't keep with next

CMP technique was conducted in collaboration with VTT in Espoo, Finland, to enable further continuation of processing at the University of Southampton facility. The processing for CMP on RMG structures consists of three stages: Batch 1 with only Si wafers was leading up as a test before the work with SOI wafers, Batch 2 with Si and SOI wafers (training on CMP) and Batch 3 processing in the cleanroom of University of Southampton.

To conduct CMP work on tree-like structures the following factors need to be considered:

- The presence and size of structures and spacing between branches (compare to Chapter 4 with mask description- Bs, Bw),
- RMG structures embedded within Si layer require precise control of the CMP stop point. The thicknesses of insulating layers and structures are within the range 100-600 nm which might be easily overpolished if removal rate (RR) of CMP will not be slowed down from standard reported values for pure slurry [1],
- the necessity of multilayer dummy structures design for uniform distribution of features across the wafer.

5.2.1 Batch 1 – process developments for 6-inch Si wafers

The process development batch was meant to gather data about CMP for tree-like structures and dummy structures surrounding them which will be used for the SOI batches. The general aim for this CMP test is shown in Figure 61.



Figure 61 The CMP aim for batch 1; a) the substrate covered with Ge before CMP, a difference be-tween trenches and surrounding layer is clearly visible, b) the substrate planarized to the level of embedded structures.

The surface of each test wafer was covered with a uniform Ge layer with the thickness in range 400-600 nm. The stop layer for the Ge CMP was the PECVD SiO_2 layer (underneath the Ge layer, Fabrication of SiGe wires on SOI platform

surrounding tree-like structures), with a thickness aimed to be 100 nm. The aim of the CMP process was to remove the Ge from the areas not in the trenches and stop polishing/ planarization on the SiO₂ stop layer, leaving Ge only in the trenches. Due to the relatively large size of the features (in range from 1 μ m for trees to 50 μ m for dummy squares) and a clear contrast between Ge and SiO₂/ Si surrounding structures, the effect of CMP is visible via optical microscopy examination. The SiO₂ thickness around the Ge structures was investigated by Reflectometer FilmTek 2000M, although a reliable value was only achieved when the layer was smooth enough. The Ge-based structure profiles have been measured by "Dektak" Surface Profilometer after each CMP step to test for proper planarization. The thickness of Ge was determined based on indirect measurement from profilometer and height differences between the SiO₂ layer and Ge trench/feature (depending on the stage of CMP). The wafer bow was measured on an ADE 9500 UltraGage to test for stress in the Ge layer.

Polishing of Ge covered substrates (wafers or samples) can be performed using a variety of slurries, including Cabot W2000 or G1000. The slurry used for this thesis for Ge CMP was Cabot W2000, designed originally as a metal slurry for Tungsten (W), diluted 1 to 1 with DI water and with 0.5% H_2O_2 added. Afterwards, the wafers were cleaned carefully with DI water only. This slurry is recommended as a slurry for blanket wafers due to a high polishing rate (200-600 nm/min) but it can also be used for patterned wafers if diluted with deionised (DI) water [2]. Alternatively, the Cabot slurry G1000 is intended for low density patterned designs [3]. Therefore, mentioned low percentage addition (<5%) of H_2O_2 to the slurry/ DI water increases the rate of Ge CMP through enhanced GeO₂ creation [4][5].

Parameters of the CMP process are listed as the following:

- Downforce = 3 psi
- Carrier Speed = 79 increased to 149 RPM after the first wafer
- Table Speed = 83 increased to 151 RPM after the first wafer
- Slurry Flow = 120 ml/min
- Slurry Cabot W2000 diluted 1:1 with DI water and H₂O₂ added to the slurry

Severe erosion seemed to occur within Ge structures of the first wafer during CMP with lower carrier and table speeds, so value of both was increased for the following wafers to avoid this effect.

Cells across the wafer can be polished with different rates due to the wafer bow, with greater pressure and faster polishing on the edges than in the middle of the wafer [6]. To maintain control over the Ge removal rate, the structures from across the wafers (cells 7, 14 and 28 as shown in Figure 62) were measured with profilometry and reflectometry before and between the steps of

the CMP (1 to 3 steps per wafer). A map of the wafer with results for various locations are shown in Figure 62.

Results are in good agreement with the prediction as the polishing rate is slower in the middle of the wafer (cell 14) and this is proven by visible Ge left between structures and dummy squares, whereas for more distant cells (cell 7 and 28) Ge deposited between trenches was removed. The underlaying PECVD SiO₂ layer was also partly polished with a measured SiO₂ thickness of 10 nm less than at the beginning of the CMP process. The non-uniform polish rate across the wafer is an issue as the yield of fully polished cells is lower than expected.



Figure 62 The Si W3 wafer after CMP; (a) the wafer after CMP, (b) a map of the wafer with the number assigned to all cells, (c) Chip 7, (d) Chip 14, (e) Chip 28.

The wafer bow was measured before and after CMP, showing a consistent trend of decreasing bow value for all wafers after CMP (bow in range 29-40 μ m before CMP and 4-11 μ m after CMP) due to the fact that the blanket Ge layer has been broken, therefore relieving the stress. The pressure applied during processing minimised the wafer bow whilst the polishing was taking place.

Between CMP steps, Ge seems to peel off (delaminate) in many places, especially in the dicing lanes which is likely to be connected to H_2O_2 added to the slurry which etches the Ge layer.

It was found that the polishing rate for dummy squares is lower than for tree-like structures (dummy squares and Ge covered spacing between them) due to a significantly bigger area than the tree-like structures. An improvement of this design was based on these results and other work on pattern density dependence reported in [7].
Based on the profilometer measurements, structures were polished within an average height difference of approximately 10 nm between the middle and the edge of the main strip of the structure (measured for structures from each cell) and more than 50 nm between average measurements for structures from cells 7 and 28. A non-uniform pattern density for the polished structures can lead to an effect called dishing which has been reported to be affected by pattern size and trench width [8].

To successfully perform the RMG, Ge or SiGe-filled trenches should be fully defined without this material left around the trench. At the same time, the structures should be still surrounded by the underlaying insulator to assure stopping CMP before over_polishing, which is defined by reaching Si. As the removal rate is faster on the edge of the wafer, the time of planarization should be optimized to stop it when there is no Ge left between the structures in the middle of the wafer and the SiO₂ layer underneath the structures is still present for cells located on the edge of the wafer.

Overall, based on the first CMP results, the SiO_2 thickness should be either increased to protect Ge within the trenches and the underlying layers during CMP, or the polishing rate should be decreased for a better control of the process.

Non-uniform polishing across the wafer can cause loss of test die across the wafer as they are not suitable for subsequent RTA because Ge left between structures can negatively affect further processing during anneal and waveguide integration for the following reasons. Firstly, Ge randomly left on the SiO₂ (without access to Si seeds) can create spherical voids during the anneal process due to surface tension between the Ge and SiO₂. Secondly, waveguides are designed between structures within the Si layer, so remaining Ge areas located on top of the Si must be removed before the waveguide etching step.

5.2.2 Batch 2 – process developments for 6-inch wafers

In light of findings from process development of Batch 1, a second batch was fabricated with a modified process as shown in Figure 63. The main modification was the introduction of an ICP etching step for the removal of the Ge surrounding the tree-like structures prior to the CMP step. In this way there is significantly less Ge to be removed in the CMP step, resulting in a more controllable and repeatable process without the risk of unwanted remaining Ge.



Figure 63 The CMP process for batch 2; a) a substrate covered with Ge before CMP with clearly visible difference between levels of trenches and surrounding layer, b) the substrate after an additional photolithography and subsequently coming ICP etch to remove Ge between the trenches, c) the substrate planarized to the level of the embedded structures.

The profiles of structures across the wafers were controlled before and between the CMP steps (the total number of steps varies between 2 and 5 for each wafer). The location of cells controlled for Batch 2 is shown in Figure 64. The location of cells was changed comparing to Batch 1 in order to increase the distance between cells and improve the uniformity measurement across the wafer. Analogously to Batch 1, the characterisation of wafers was based on measurements of SiO₂ thickness next to the Ge structures (by Reflectometer) and the structures' profiles ('Dektak' Profilometer).







During the CMP processing, SOI wafer number 1 was broken and was not processed further. However, the rest of wafers were planarized until the Ge structures were level with the SiO₂.

Parameters of the process are listed as follows:

- Downforce = 3 psi
- Carrier Speed = 47 RPM
- Table Speed = 53 RPM
- Slurry Flow = 120 ml/min
- Slurry Cabot W2000 diluted 1:1 with DI

Parameters were adjusted based on results for Batch 1. The slower speed (for carrier and table) was used to slow down the CMP process and allow to control the process even with an increased pattern density- after adding Ge etch mask, compared to Batch 1.

The slurry used for the Ge CMP was Cabot W2000 diluted 1+1 with DI water. H_2O_2 has been excluded due to delamination of Ge reported for CMP of Batch 1. Additionally, H_2O_2 also increases the polishing rate of Ge [9][10] and a high polishing rate is not needed for the thin layers in batch 2 with a low Ge coverage due to the new patterned design, which was meant to increase the polishing rate [7].

The initial and final thicknesses of SiO_2 across the example wafer are shown in Table 8. A significant amount of PECVD SiO_2 was polished away across the whole wafer, with even no SiO_2 left in some areas. Without the SiO_2 layer (stopping layer), the Si layer would be polished away along with the Ge structures, which would inhibit the performance of the waveguides which are to be etched as

the final processing step. The selectivity of the CMP between Ge and PECVD SiO_2 is around 2:1 (meaning that 200 nm of Ge was polished at the same time as 100 nm of PECVD SiO_2).

 Table 8
 Results for wafer Si3 (Si test substrate) with an insulating SiO₂ layer surrounding treelike structures, before and after CMP. This data provides information about PECVD SiO₂ uniformity across the wafer and minimised wafer bow after processing.

WAFER	CMP TIME [s]	CHIP-1 SiO₂ [nm]	CHIP-2 SiO₂ [nm]	CHIP-3 SiO₂ [nm]	Wafer bow [µm]
Si3	0	94	96	105	15
Si3	420	43	14	0	8

The initial values of the wafer bow were much lower for Si wafers which had been processed with a Ge ICP etch compared to Si wafers covered with blanket Ge (Batch 1). The bow measured for Batch 2 was found to be around 15 μ m compared with around 30 μ m for Batch 1. The final values were comparable for both Batches.

The bow measured for SOI wafers was much higher than for Si wafers, even between 30 and 50 μ m for each SOI wafer. The possible reasons for the high bow for SOI wafers have been presented in the literature [10], but it is apparent that the bow of a virgin SOI wafer is high according to the wafer specification from the wafer manufacturer.

Profilometer measurements were carried out to identify a profile variation within the tailored structures from wafers from Batch 1 and Batch 2. For Batch 1, dishing was commonly observed, while for Batch 2, the shape was convex. Structures (trenches) and surrounding them area from Batch 1 were evenly covered with the layer of SiGe or Ge, whereas for Batch 2, areas surrounding trenches were etched to the level of SiO₂. The accuracy of photolithography allows around 1 μ m of margin around structures so overgrowth of the design was set to 1 μ m from any direction. The reason of the profile after CMP to change was a difference with access to the edges of structures for both batches. The area of spacing between structures, designated to be uncovered, is much bigger than area of branches or strips. This led to a difference in the polishing rate, higher for smaller regions like tailored structures (either stripes or branches). The slower planarization of area surrounding trenches with simultaneous faster planarization of branches is causing mentioned earlier dishing along structures and material left between structures. Waveguides are designed to be fabricated exactly between dummy squares, where material is not removed if additional mask for etch is not added. To sum up, process of Batch 2 was significantly more adjusted to the design's needs as Ge had been removed from all undesired areas, including spacing between the branches

Fabrication of SiGe wires on SOI platform

of the tree-like structures. At the same time material within trenches were not over_polished. An example of Wafer from Batch 2 is shown in Figure 65.



Figure 65 An optical microscope image of the surface of the wafer SOI 3 with tree like structures and dummy squares after CMP.

The result shown in Figure 65 can be compared with results for Batch 1 presented in Figure 62, where Ge between dummy squares and branches is still clearly visible due to not fully polished wafers.

A summary of the findings from the CMP process development performed on Batch 1 and Batch 2 are listed below:

- the uncovered PECVD SiO₂ can be problematic as it is too fragile to be a stop layer for Ge CMP. A denser thermal oxide layer could be used instead, although this consumes Si and would result in a thinning of the top Si overlayer, which will affect the waveguide and other devices performance,
- the size of dummy squares should be adjusted to match the size of the structures to be polished,
- parameters of the CMP tool (force applied on the back of the wafer, carrier speed) and slurry (concentration, other substances added to commercially available slurry) should be adjusted carefully when processing requires precision due to thin Ge and insulator layers,

- dummy squares and tailored structures should be fabricated in the same way; fabrication within 3 layers to allow uniform growth, planarization and RTA,
- the additional mask just before CMP to remove the Ge around structures is found to be beneficial for CMP performance.

5.3 Lithography

RMG processing requires minimum two layers to fabricate; structures on insulator and seeds. To adapt the process for the SOI platform, the trenches were etched first into the Si layer together with the dummy squares. After this step, an insulator layer was deposited, and it was followed by subsequent lithography for seed etching together with selectively removing the insulator from the dummy squares. After Ge deposition, the final lithography was designed to protect the Ge structures so that an etch step can remove the unwanted Ge before the CMP step.

Lithography was used for 3 steps of processing as shown in Figure 66 and Figure 67. The design was adjusted to account for the alignment accuracy of the photolithography with a tolerance of 1 μ m for the second and third layers to cover the underlaying structures.



Figure 66 The layout for RMG structures; a) Mask 1 and tree-like structures and dummy squares,
b) Mask 2 with seeds and another layer (to etch SiO₂) of dummy squares, c) the final structure after the photolithography 1 and the photolithography 2.

As mentioned in Section 5.1, process development of lithography can lead to general improvements of the shape of the structures patterned onto the substrate. Process development for photolithography is based on parameters like exposure time, using either contact or non-contact masks, or development time.

The findings from CMP processing led to another mask as shown in Figure 67. The mask was used before CMP process to remove material surrounding structures and dummy squares. From process development point of view, it is important to consider accuracy of the photolithography in range of 1μ m and rounding shape of squares after typical for photolithography processes. The size and shape of structures after photolithography can be modified by just mentioned parameters considered for process development.



Figure 67 Lithography mask for process (a) is designed for tailored structures to be etched into the Si layer along with dummy squares during Photolithography 1. Lithography mask (b) is used after filling trenches with Ge/SiGe and etching seeds, the structures should be planarized as described in Section 5.2. The negative resist used for this design assures that all surrounding areas are etched, and that the trenches and dummies are protected from the etch.

Finally, the shape of the tailored structures was challenging to achieve using contact photolithography processing. The dimensions of the branches or strips, and the spacing between cooling branches, were within the range of 1-5 μ m, challenging abilities of contact photolithography. As shown in Figure 68, the shape of the branches varied depending on the lithography technique used. Multiple photolithography processes with varied exposure time, development time or substrate contact with the mask (either contact or non-contact lithography) were examined with the final effect shown in Figure 68a.



Figure 68 The photolithography ((a) for 1.8 sec exposure time, 40 sec development) results in rounded corners of the design with straight angles, whereas the structure obtained by e-beam (b) is in good agreement with designed structures.

The e-beam technique was ensuring the most accurate transfer from the design into the fabricated structures, based on the described process development for photolithography and e-beam lithography. The shape of the corners had an impact on the results from other processes like wet directional etching of either Si or SiO_2 , which were especially visible after finishing the RMG processing.

Photolithography is significantly cheaper and faster than e-beam lithography, however, the technique compromised the shape of tailored structures in significant way. The SOI integration demanded RMG structures to be embedded into Si layer and patterns obtained after lithography had a direct impact on the result. Therefore, the e-beam lithography was required to meet design expectations.

5.4 Process development summary

As a summary of the process development, the following findings can be listed:

 all processes used for SOI integration of RMG structures impact the final effect in a significant way. The rounded shape obtained after photolithography was enhanced by directional wet etch causing increased difference from the designed shape. Even if rounded corners didn't have such a high impact on the tailored structures on Si, the results on SOI show less tolerance when the tailored structures were etched into the Si layer instead of being on the top of the Si layer.

- The CMP process required modification of the mask and the design to achieve the final
 effect of cleared areas around structures and not over polished structures. The dimensions
 and thicknesses were challenging for CMP where the slurry is usually designated for
 polishing more material in the short time.
- E-beam was the method of choice for defining the tailored structures. However, time and cost of processing were not ideal and left a room for improvements.

5.5 Constant composition SiGe/Ge wires on SOI platform

Based on the process development described in detail in the earlier sections of Chapter 5, a range of structures were fabricated to demonstrate RMG integration to the SOI platform. The modifications of the processing leading to constant composition SiGe/ Ge wires embedded in the silicon overlayer are presented in this chapter.

5.5.1 Fabrication

The fabrication was carried out on 6" SOI wafers (400 nm Si, 2μ m buried SiO₂) and Si test wafers. For the most effective photonics devices the RMG SiGe/Ge layers must be embedded within the Si layer to enable butt coupling to Si waveguides. To achieve this, the first step was the deposition of a 50 nm thick SiO₂ hard mask using plasma enhanced chemical vapour deposition (PECVD). The hard mask was then patterned using electron-beam lithography to deliver the desirable shapes for the cavities of the proposed structures. The cavities within the Si layer were etched using combined dry and wet etching techniques to obtain an etching depth varied between 100 and 300 nm with an angle of 54.3°, as a result of the TMAH etch. To avoid issues caused by under etching of the silicon, the hard mask was fully removed after this step. The insulating SiO₂ layer was than replaced across the whole wafer using dry oxidation of the Si layer. This step could be done by PECVD, however, this process would demand additional actions like thermal dehydrogenation and densification, as shown in the previous chapters. After the insulator formation on the whole wafer, the openings for the Si seeds were etched through the oxide. To achieve this, the substrates were patterned by contact photolithography and wet etched with HF etchant (diluted solution, 20:1). This step was used to expose the Si underneath and on the sidewalls of the trench-seed connection in the seed region as shown in Figure 69.







A non-selective SiGe deposition was performed by using low-energy plasma-enhanced chemical vapour deposition (LEPECVD), in collaboration with LNESS group, to maintain high quality and low hydrogen content within the material. This step was preluded by an HF dip to ensure the removal of the native SiO₂ within the seed area, with the parallel formation of Si-H bonds which prevent surface oxidation in the time between the HF dip and the SiGe deposition. Wafers with a SiGe film covering the whole area of the substrate were then ready for planarization. To improve the performance of the CMP, the film was patterned with UV contact lithography and dry etched with an inductively coupled plasma (ICP) tool to remove the unwanted SiGe around the trenches. The level of the SiGe material was planarized to the level of the insulating layer by Chemical Mechanical Polishing (CMP) with dilute 1:1 W2000 slurry. As mentioned in Chapter 4, the fabricated design also contained dummy squares filled with SiGe, which are helpful to sustain pattern uniformity during the planarization process. The size of the dummy squares was 1x1 µm (for the trench etch layer) so that they were comparable to the size of the device trenches. The CMP processes was followed by PECVD of a 1 μm thick SiO_2 insulating cap. This material was deposited on top of the wafer to encapsulate the structures before the rapid thermal annealing (RTA) process. Afterwards, RTA was performed to finalize the creation of the RMG structures on the SOI platform. The temperature of the RTA was firstly stabilised at approximately 500°C before ramping up to the desired peak temperature in the range between the Si and Ge melting points (chosen to be 950 °C according to the thermocouple attached to the back of the wafer) at a rate of approximately 100°C/s. The time at the peak temperature was 1 s, before letting the wafer cool naturally (by switching off the RTA lamps). The resulting structures were characterised with Raman spectroscopy to confirm the composition across the structures. After Raman spectroscopy measurements, the top insulating layer was removed using a combination of dry etching (ICP) and 20:1 HF wet etching in order to facilitate the material characterization with scanning electron microscope (SEM).

5.5.2 Wires with graded composition on SOI

Conventional RMG is performed on simple straight strips emanating from a Si seed. Results for a variety of 3 μ m wide strips with 10 μ m overlap to the seed area (Sc) are shown in Figure 70.



Figure 70 The Ge composition for strips embedded in the SOI platform, (A) a graph for 3 μm wide strips for a variety of lengths, (B) comparison of Ge composition for these strips with normalized length values next to the complete mixing model (referred as a model).

The anneal temperature within the structure (due to inaccurate measurement of thermocouple) might be determined locally based on the phase diagram for SiGe alloys and finding the temperature for the highest Si content in the structure (seed area). The 30 % Si content is related to around 1030 °C based on the phase diagram presented in Chapter 2.

As can be seen in Figure 70, the composition of RMG straight strips is graded along their length. This is a result of the separation between the solidus and liquidus on the SiGe phase diagram (details presented in Chapter 2) resulting in preferential Si rich solid formation. This situation occurs especially when cooling of structures is not immediate. The data from Figure 70A can be normalised against the strip length, as seen in Figure 70B. The graded composition of the SiGe stripes was also described by the complete mixing model [11], which is implemented into Figure 70B. This model assumes a relatively slow regrowth front propagation speed (slow in the sense that the regrowth front propagation speed is slower than the diffusion constant of Ge in SiGe/Si) enabling the Ge rich liquid at the regrowth front to fully diffuse into the melt [12]. The diagram of the complete mixing model is shown in Figure 71 to illustrate the principle of the mechanism.



Figure 71 Complete mixing model adapted from [12].

The composition is changeable and according to the complete mixing model it should be dependent on the solidified fraction of the normalized length of the strip. Based on the difference of the area under the peak for raw data and Lorentzian fit, the errors for the composition were found to be within 2-4% of the data, which means the data on the graph is in a good agreement within this range.

The data from Figure 70B corresponds to a maximum of only 2% difference between each strip length after reaching 70% of the total length. This data can be compared with the research for strips on the Si platform in the literature [12] . Even if the trend is similar, the high composition (above 95% of Ge) is reached much earlier for the Si platform (around 0.5 of normalized length) than for the SOI platform presented in Figure 70 (around 0.7 normalized length). The anneal temperature presented here for the SOI platform was higher than the literature value for the Si substrate (1030 °C compared with 1008°C, when the anneal temperature is calculated in the same way for both substrates) and the amount of Si diffused further into the strip is higher due to the higher initial Si content. This finding shows that the trend for RMG strips embedded into the SOI platform is analogous as for the strips on the top of the Si substrate.

5.5.3 Branch parameters for RMG structures embedded into SOI

As mentioned in Chapter 4, the cooling branches of the RMG structures embedded into the SOI were expected to support faster cooling along the main strip, in turn leading to a faster regrowth front propagation speed, according to the steady-state solidification model [12]. The simulation done based on this model is shown in Figure 72 for varied regrowth front propagation speed.



Figure 72 Simulations of SiGe growth using steady-state solidification model for the fast regrowth front propagation, adapted from [12]. Situations from a) to c) correspond to the increasing regrowth front propagation speed.

This model assumes that the regrowth front propagation speed is faster than the Ge diffusion into the Ge/SiGe melt, meaning that the Ge rich region at the regrowth front is not able diffuse into the melt. The faster propagation speed, the flatten the composition profile of the structures.

Within this project, the additional area of the added cooling branches was used for faster heat dissipation, subsequently leading to constant composition in the main strip of branches as shown in Figure 73. To confirm this hypothesis, parameters such as the branch length/ area, branch separation (BS) and the distance from the seed (D) were investigated when gathering results. Additionally, the potential applications for building active devices like Franz-Keldysh modulators were also considered to ensure appropriate structure geometry with a constant SiGe composition profile along a 50 μ m long strip with a width greater than 1 μ m. All the presented results were gathered from samples with deposited Si_{0.015}Ge_{0.0985}.

The correlation between the uniformity of the composition profile and the branch length was confirmed for both types of RMG structures presented in Chapter 4 (see Figure 69 for a clarification

of the two RMG types). Matching trends for both types of structures are highly important for further waveguide integration, where RMG2 type structures are preferable for easier integration of waveguides and active devices. This is related to the elimination of the seed from the area designated for device fabrication.

Based on the methodology described in Chapter 3, Raman spectroscopy was used to measure the spectra along the structures and subsequently to analyse it and to calculate the SiGe composition.

Results obtained from the RMG structures show the relationship between the branch length and the uniformity of the composition profile, as shown in Figure 73 and Figure 74.



Figure 73 Dependence of the SiGe composition profile on the branch length for RMG1 type structures for 30 um long strips with a branch length of 5, 15 and 25 μ m for structures A, B, C as shown in graph. The position of the measured strip is highlighted on each image.



Figure 74 Dependence of the SiGe composition for single tree like structure (type RMG2) for different length of the cooling branches.

The length of the branches is confirmed to be crucial for the gradient of the composition profile, with a length of between 20 and 25 μ m proving sufficient to achieve a uniform SiGe composition, which is shown in Figure 73 and Figure 74 for RMG1 and RMG2 respectively. The width of all branches (1 μ m) was the same for fair comparison of the composition profile. The width of the strip with constant composition was 2 μ m for structures presented in Figure 73 and 1 μ m for structures presented in Figure 74. All parameters can be found in Figure 75 for the sake of clarity.



Figure 75 Parameters of the RMG2 type structure, where: Bs - Branch Separation, w - width of the strip with constant composition, Bw - Branch width, D - distance from seed to the start of the strip with constant composition, D1 - distance from the start of the strip with constant composition to the end of the main strip, SL - seed length, Sw – seed width, Sc- seed connection, L- length of the strip with constant composition.

A significant difference can be observed for a branch length in range 5-25 μ m. The ratio of cooling branch area to the main strip area was found to be crucial for the composition profile. Structures RMG1 shown in Figure 74 had ratios as follows: 1.66 for 5 μ m long branches, 5 for 15 μ m long branches and 8.33 for 25 μ m long branches. For RMG2 structures, as presented in Figure 74, the ratio of cooling surface to the main strip were 2.4 for 5x1 μ m cooling branches and 9.6 for 20x1 μ m cooling branches. This implies that a ratio > 8 must be used in order to achieve a uniform composition in the main strip, which is of high importance for the heat dissipation from the main strip and in turn the cooling rate of the system. The area of cooling branches is of high importance for the heat dissipation because it leads to faster cooling by contact with the cooler surroundings. This is linked to a faster regrowth front propagation speed and a more uniform composition profile along the main strip. Minimising the length of the branches whilst achieving a constant composition profile is important to limit the footprint of the structure to enable densely packed devices and therefore a reduction in the cost per device.

Besides the length of the branches (BL), the separation between the branches (Bs) was also considered as a factor for the design of the tailored structures. For the sake of clarity, the parameters mentioned here are shown previously in Figure 75.

The results of varying the branch separation are shown in Figure 75. As above, structures had the branch length varied between 5 and 25 μ m. The separation (Bs) was either 3 or 6 μ m, for branches with the same width (Bw) and the same width of the main strip (w).





These results are showing the dependence between the SiGe composition uniformity and the geometrical parameters of the tailored structures with a focus on the branch area and main strip ratio. To obtain enough cooling and subsequently constant SiGe composition along the main strip,

a greater surface area of cooling features is needed to maintain fast heat dissipation. In Figure 76, two separations were compared for two lengths of branches. The change in the composition profile was negligible for the shortest branches of 5 μ m because the impact of this length of branch is insufficient to achieve fast cooling. On the contrary, the composition profile of the main strip with 25 µm long branches was much more affected by the change in separation. A changeable separation of the cooling branches corresponds to a change in ratio between the cooling branch area and the main strip. The calculated ratios are as follows: 0.75 and 1.25 for 5 μm long branches and 3.75 and 6.25 for 25 μ m long branches, when the branch separation was 6 μ m and 3 μ m respectively. A uniform composition was achieved only for the ratio 6.25 for 25 μm long branches and 3 μm separation. Additionally, for the narrow branch separation, the minimum Ge content is higher than for a wider separation due to a bigger total area/ volume and limited diffusion from the Si seed. However, the composition profile is flatter for smaller branch separation, there is a limit (value to be confirmed) when the branches would stop help with heat dissipation and a whole structure would behave as a strip with graded composition. The length of the main strip with branches added was limited to 37 μm (after this length no branches were added) which explains the changeable composition after this length for the results shown in Figure 76b.

5.5.4 Seed connection for RMG structures embedded into SOI

The overlap between the SiGe strip and the seed (seed connection) was also tested as a parameter relevant for the tailored structures. The range of the contact length was varied when other parameters stayed constant. Results for the RMG2 structure with a uniform Ge composition profile are shown in Figure 77.



Figure 77 The seed connection (Sc) test, (A), test for RMG2 structures with parameters as following: BL=20μm, w=3μm, Bs=4μm, Bw=1μm, D=25μm, (B) SEM image showing an example test structure set with a variance of the seed connection.

Based on the seed connection test for SOI samples, there was no significant difference (above 2%) between the composition of the strip with constant Ge composition. This implies that Si diffusion from the seed is not limited by the overlap between the seed and the SiGe strip₂ which is expected due to the high diffusion coefficient of Si in liquid Ge [13]. Nevertheless, this test was conducted only for one anneal temperature and more research is required for a variety of anneal temperatures e.g. higher temperatures to obtain higher Si content within the strip.

5.5.5 Composition tuning for wavelength division multiplexing (WDM) on SOI platform

The RMG process on SOI can be utilised for a plethora of active devices e.g. modulators operating at a wide range of wavelengths aimed at exploiting wavelength division multiplexing (WDM) to dramatically increase the bandwidth of an optical link. To fabricate electro-absorption modulators suitable for exploiting WDM, the SiGe composition of each device should be tuned in a predictable manner, preferably as a function of the distance from the seed. In this way, the SiGe composition of each modulator can be tuned so that the operating wavelength of each modulator is matched to the wavelength of each channel in the WDM device. The experimental design to test this tuneability was based on single tailored structures (RMG2 type) with varied distance D of the "tree-like" structures from the seed, whilst maintaining consistent branch dimension parameters (set to be BL=20 μ m long branches, Bw=1 μ m, Bs=2 μ m , w= 3 μ m , and L=50 μ m for the main strip to achieve constant SiGe composition). An example of the fabricated structures is shown in Figure 78 with all the parameters mentioned.



Fabrication of SiGe wires on SOI platform

Figure 78 SEM image of structure used for obtaining tuned constant SiGe composition along the main strip. The varied parameter is distance, D1, from the seed to the strip with constant SiGe composition, distance, D, from the seed to the end of structure and Branch length (BL). Intermixing region between Si and Ge in the seed area is pointed on the image.

The structures shown in Figure 78 were used to research the relationship between the SiGe composition and the distance from the seed area. The results obtained for this experiment are presented in Figure 79.



Figure 79 SiGe composition controlled by the distance, D, between the seed and the main strip with attached branches.

From Figure 79, the uniform profiles for various SiGe compositions can be seen. The Ge content increased with distance from the Si seed as expected, based also on results presented in Figure 71 for strips embedded into SOI platform. All added structures had the same dimensions to obtain the same ratio between the area of the cooling branches and the area of the main strip. The tuneability was based on a change of the ratio between distance D (distance between the seed and the main strip designed as a base for subsequent active device fabrication) and distance D1 (the remaining length of the strip emanating from the seed). Based on this, the ratio of D and D1 was found and described as proportional and it is shown in Figure 80. The length of the main strip, L, (50 μ m) was designed as a suitable length for analogical active device fabrication as mentioned in Chapter 4.3.

Fabrication of SiGe wires on SOI platform



Figure 80 A diagram of the ratio of normalized length D1/D against Ge composition. The standard deviation for each set of measurements for each normalised length was respectively: 0.007, 0.008 and 0.01.

The first attempt to utilise the SOI platform for WDM applications is shown in Figure 81 for a tailored structure with 4 "tree-like" structures (Figure 81a).



Figure 81 The tailored structure with 4 added "tree-like" structures as shown in (A) with the composition profile for branch 1, branch 2 and branch 3 as shown in (B). The distance, D, between seed and the first structure is 60 μm, distances between branches 1-2 and branches 2-3 are also set to 60 μm, Bs=2 μm, Bw=1 μm, w=1 μm. The total length of the strip of the structure is 264 μm.

Branches 2 and 3 are characterised with constant, high Ge content, which is promising for further development of this technique on the SOI platform for WDM applications in the C-band (where compositions in the range 98-100% are required) with devices of minimized area [14]. The fourth "tree-like" structure was not characterised due to a disconnection in the seed strip leading to incomplete RMG. Consequently, more process development on structures with multiple "tree-like" structures is required to obtain a uniform profile across all the structures.

In order to compare the effect of geometry on RMG structures embedded in Si, with those embedded in SOI, results from a Si test wafer are also presented in Figure 82. The Si substrate was processed in the same way as the SOI, with the "tree-like" structures embedded into the substrate. The structures had the same parameters for distances, D (25 and 80 μ m) and D1 (20 μ m), as structures from Figure 79 for the SOI platform.

Fabrication of SiGe wires on SOI platform



Figure 82 Results for a RMG2 type structures embedded onto the Si platform, Branch Separation,
 BS, was kept constant as 3 μm, other parameters are as follows: (A) Bw=1 μm, w=1 μm, (B) Bw=2 μm, w=2 μm, (C) Bw=3 μm, w=3 μm.

The ratio between the cooling branch area and the main strip area was as follows: 6.4, 8 and 9.6. for A, B and C respectively. Also, in this case a ratio of 8 is sufficient to result in uniform composition. Overall, the lowest deviation was shown for the set B with parameters: $Bw=2 \mu m$, $w=2 \mu m$.

A high Ge content is desirable for EAMs operating in the C-band and this data proves that it is possible to tune the Ge content within the desired range (98-100%), at least on the silicon platform. However, since the silicon based devices were processed in the same way as the SOI based devices, and all other trends have been comparable between Si and SOI, it points to the fact that it will be possible to achieve the same control on the SOI platform with further process optimisation and a lower annealing temperature.

5.6 Summary

The results presented in Figure 79 demonstrates the proof of principle for tuneability of RMG structures on the SOI platform. This result is crucial for the compatibility of the process developed in this thesis for the integration of active RMG devices with waveguides on the SOI platform, paving the way for a new era of low-cost devices like photodetectors or Franz-Keldysh modulators operating at a range of wavelengths, exploiting the significant benefits of WDM.

Within this work, the ability to not only achieve a uniform SiGe composition on the SOI platform using a RMG technique has been proven, but crucially the ability to tune the uniform composition across the wafer without modifying the processing steps has also been demonstrated. This proves the ability to tune the composition of the SiGe structures on the SOI platform using only a single deposition and single anneal step, a result that has never been reported for the SOI platform. Research on the effect of cooling branches has only been demonstrated previously for the Si platform [13][15]. This first demonstration on the SOI platform promises to allow the development of new devices with high performance and low cost. An example application can be EA modulators operating between 1500 nm and 1600 nm, which demand a Si content between 2% and 0.5%, respectively. Another potential applications like quantum wells for modulators or transistors at 1310 nm.

Chapter 6 Conclusions and Future Work

6.1 Conclusions

Both SiGe and Ge are highly important elements of photonics circuits, operating in a wide range of near-infrared and mid-infrared wavelengths. A variety of Ge and SiGe applications require systems with multiple Ge and SiGe based devices to be built across one chip; reported examples of highspeed active devices such as high-speed photodetectors, electro-absorption modulators or infrared waveguides are suitable for Ge-based design. The plethora of epitaxial growth methods for Ge and SiGe on Si are discussed in the literature review. However, there are a number of areas leaving room for improvement in the SiGe/Ge growth process including the fact that the presence of a lattice mismatch between Ge and Si results in high dislocation densities in the SiGe/Ge layer, and the ability to grow only a single composition on each wafer. Applications requiring multiple compositions of SiGe or pure Ge like wavelength division multiplexing (WDM) are not achievable with the commonly used growth methods. A single growth step allowing multiple compositions across a single wafer is a desirable solution for a high number of applications. This project is oriented with a special focus on finding opportunities for fabrication of devices operating within the C-band wavelength window (1530-1565 nm), and especially 1550 nm. This wavelength is suitable for active devices like Franz-Keldysh electro-absorption modulator (EAM). The composition of the structures to meet this aim is found to be around 98.5-99.5% along with a device length of approximately 50 µm.

This project focused on the development of SiGe and Ge structures obtained from the rapid melt growth (RMG) technique on the SOI platform, which is required for active device integration with waveguides. The RMG technique uses similar principles on SOI and Si platform. RMG allows the overgrowth of SiGe or Ge on an insulator layer from a Si seed. For the SOI platform, the formation of structures/ trenches is patterned into the Si overlayer, and then filled with Ge or SiGe, which is a critical step for the success of the subsequent processing. The trenches embedded into the Si and subsequently covered with an insulating layer should enable the Ge to fill the trench without voids and with connection to the Si only in the designated seed area. The design within the SOI platform demands that the surface is planarized with Chemical Mechanical Polishing/ Planarization (CMP). The substrate planarized to the level of the trenches filled with Ge is covered with additional insulating layer before Rapid Thermal Anneal (RTA) when the Ge will be melted. During the melt Si diffuses from the area into the Ge or SiGe structure, creating SiGe melt. The difference between melting temperatures for Ge and Si allows Si to stay in the solid state and to be used as a crystalline template for the epitaxial regrowth of SiGe.

The integration of the RMG structures into the SOI platform has required extensive process development work across many of the fabrication steps including trench etching, Ge or SiGe deposition and CMP.

RMG enables defect-free structures which is not achievable by any other technique of epitaxial growth. Defects associated to the lattice mismatch are trapped within the Si and Ge/SiGe interface due to the epitaxial necking mechanism. This part of the structure is designed to not be used during waveguide integration or fabrication of active devices. The initial Ge/SiGe deposition is unselective with multiple available options e.g. PECVD or LEPECVD, however, the hydrogen content within the layer and other impurities play a role as a factor for further processing. The number of defects obtained from this initial growth step do not influence the final quality of the structure due to the Ge melt step at high temperature (above Ge melting point, 938 °C). The characterisation of the RMG structures is based on the Raman spectroscopy technique, which compares the intensity ratio of the Ge-Ge and Si-Ge peaks obtained from each measurement.

Tailored structures with varying area of cooling branches and varied distance from the seed were researched on the SOI platform to compare the composition across analogical structures on the Si platform. The impact of increasing the area of the cooling branches was confirmed for the SOI platform in fairy similar way as for the research on Si [1], with the most effective cooling, and therefore uniform composition profile, for 25 µm long branches. Also, the constant composition structures (in range 80-100% Ge) were presented on SOI platform. The controllability of the composition of the compound by design was demonstrated and further work on refining the process and structures would potentially demonstrate the suitability for EAM's for WDM applications.

RMG on SOI allows fabrication of multiple compositions of SiGe across the wafer after only a single deposition and a single anneal step, which is unique across all epitaxial growth methods. Moreover, the tuneability achieved by RMG can be controlled by the design of the structures and also uses standard, commonly accessible fabrication steps. This method creates a low-cost and accessible solution, removing the limitations of only being able to achieve one SiGe composition across the wafer, and the high SiGe/Ge layer defect densities resulting from the Si and Ge lattice mismatch.

6.2 Future work

The work on RMG growth of SiGe and Ge structures within the SOI platform delivers multiple area of interest for further research. Possible applications of an SGOI platform are listed below:

• Optimisation of RMG structures within the SOI layer for further miniaturisation of devices

Conclusions and Future Work

- Expansion of the composition towards lower content of Ge (region below 85%): The
 properties of SiGe alloys depend on the composition. The SiGe alloys from Si-like
 composition region, under 85%, can be successfully applied as a component of modulators
 e.g. as a strain layer [2][3]. The Ge content as low as the range 15-40% could therefore be
 researched for photonics applications.
- Waveguide integrated structures on SOI platform: Waveguide integration within the SOI platform is required for tests of the optical properties of the material e.g. optical losses, band gap. This step is leading the way to a plethora of active devices like detectors or modulators placed within waveguides.
- SiGe devices like electro-absorption modulator (EAM) based on Franz-Keldysh effect: These
 are ideal devices in the age of increased demand for high-performance devices, where low
 power consumption and high speed are crucial. The high quality of the RMG material is a
 chance to improve speed or footprint of modulators [4] if built using the demonstrated
 RMG method with the added benefit to decrease the number of fabrication steps and cost.
- Fabricate a wavelength division multiplexed (WDM) optical link: The multi composition SiGe regions across one structure allows fabrication of an array of modulators working at slightly different wavelengths to be realised. Moreover, the high-quality pure Ge strips can also be realised as a material for photodetectors, allowing realisation of a full WDM optical link.
- Multilayer devices using Si seed: The ability to create RMG structures from Si seeds is a chance to fabricate multilayer devices using either Ge or SiGe. The ability to create photonics or electronics devices across multiple layers is a chance to maximize the number of devices per area of the chip.
- Improvement of Ge photodetectors on Si and SOI platform: Low cost photodetectors are important for silicon photonics. The material quality was proven to be highly important for the performance of the resulting device [5]. Further work on Ge photodetectors fabricated by RMG process is a chance to optimize the performance of devices.

Appendix A [Full process flow]

Step No.	Process Name	Process Description	
1	SiO ₂ Deposition #1	PECVD 50 nm / Standard Recipe	
2	E-Beam Lithography #1	Trees trenches lithography	
4	SiO ₂ Etch #1	ICP etch 50 nm	
5	Si Etch #1	ICP etch 100 nm	
6	Plasma Clean #1	O ₂ Plasma Standard Clean	
7	Photolithography #1	Dummy squares/ S1813	
8	SiO ₂ Etch #2	ICP etch 50 nm	
9	Si Etch #2	ICP etch 100 nm	
10	Plasma Clean #2	O2 Plasma Standard Clean	
11	HF Dip #1	HF Dip 5 seconds 20:1	
12	Si Etch #3	TMAH Angled Si Etch 200 nm	
13	HF etch	SiO ₂ etch all layer	
14	SiO ₂ Deposition #2	PECVD 60 nm / Standard Recipe	
15	Photolithography #2	Seed Windows / S1813	
16	SiO ₂ Etch #3	HF Etch 60 nm 20:1 (oxide removed only in the trench)	
17	Plasma Clean #3	O ₂ Plasma Standard Clean	
18	HF Dip #2	HF Dip 5 seconds 20:1	
19	SiGe Deposition #1	LEPECVD 300-400 nm SiGe/ it might be replaced by Ge LEPECVD	
20	Photolithography #3	Ge etch / \$1813	
21	Ge Etch #1	ICP etch for Ge around structures	
22	Plasma Clean #4	O ₂ Plasma Standard Clean	
23	CMP #1	CMP / Ge above the trench removal	
24	Wafer Clean #1		
25	SiO ₂ Deposition #3	PECVD / protective cap 1000 nm	
26	Anneal #1	RTA / chip scale	
27	SiO ₂ Etch #4	ICP SiO ₂ down to 70 nm of SiO ₂	
28	SiO ₂ Etch #5	HF Etch All 20:1 SiO ₂ left each sample separately	
29	Characterisation #1	Raman / Optical Measurements	

List of References

Chapter 1 Introduction

[1] A. P. Knights and K. Doylend J., "Silicon photonics-recent advances in device development," in Advances in Information Optics and Photonics, 2008.

[2] D. Padua, Encyclopedia of Parallel Computing, 2011.

[3] M. Roser and H. Ritchie, "https://ourworldindata.org/technological-progress," 2020.

[4] R. Tanburn, E. Okada, and N. Dattani, "Reducing multi-qubit interactions in adiabatic quantum computation without adding auxiliary qubits. Part 2: The 'split-reduc' method and its application to quantum determination of Ramsey numbers," Aug. 2015.

[5] D. Thomson, A. Zilkie, J. E. Bowers, T. Komljenovic, G. T. Reed, L. Vivien, D. Marris-Morini, E. Cassan, L. Virot, J.-M. Fédéli, J.-M. Hartmann, J. H. Schmid, D.-X. Xu, F. Boeuf, P. O'Brien, G. Z. Mashanovich, and M. Nedeljkovic, "Roadmap on silicon photonics," J. Opt. (United Kingdom), 2016, doi: 10.1088/2040-8978/18/7/073003.

[6] A. Liu, R. Jones, L. Liao, D. Samara-Rubio, D. Rubin, O. Cohen, R. Nicolaescu, M. Paniccia, "A high-speed silicon optical modulator based on a metal-oxide-semiconductor capacitor," Nature, 2004, doi: 10.1038/nature02310.

[7] G. T. Reed and A. P. Knights, "Silicon Photonics: An Introduction," 2004.

[8] C. G. Littlejohns, M. Nedeljkovic, C. F. Mallinson, J. F. Watts, G. Z. Mashanovich, G. T. Reed and F. Y. Gardes, "Next Generation Device Grade Silicon-Germanium on Insulator," Sci. Rep., 2015, doi: 10.1038/srep08288.

C. G. Littlejohns, T. Dominguez Bucio, M. Nedeljkovic, H. Wang, G. Z. Mashanovich,
 G. T. Reed and F. Y. Gardes., "Towards a fully functional integrated photonic-electronic platform via a single SiGe growth step," Sci. Rep., 2016, doi: 10.1038/srep19425.

[10] R. Baets, A. Z. Subramanian, S. Clemmen, B. Kuyken, P. Bienstman, N. Le Thomas, G. Roelkens, D. Van Thourhout, P. Helin, and S. Severi, "Silicon photonics: Silicon nitride versus silicon-on-insulator," in 2016 Optical Fiber Communications Conference and Exhibition, OFC 2016, 2016, doi: 10.1364/ofc.2016.th3j.1.

[11] K. Wada and L. C. Kimerling, "Photonics and electronics with germanium," 2015.

[12] R. Soref, "Mid-infrared photonics in silicon and germanium," Nature Photonics. 2010, doi: 10.1038/nphoton.2010.171.

[13] I.E. Day, S.W. Roberts, R. O'Carroll, A. Knights, P. Sharp, G.F. Hopper, B.J. Luff, and M. Asghari, "Single-chip variable optical attenuator and multiplexer subsystem integration," in Conference on Optical Fiber Communication, Technical Digest Series, 2002, doi: 10.1109/ofc.2002.1036211.

[14] "https://www.mellanox.com/" accessed on 03/04/2020.

[15] T. Morgan, "https://www.enterpriseai.news/2014/06/12/hp-puts-memristorsheart-new-machine/," 2014 accessed on 03/04/2020.

[16] M. P. Mack, A. Ayazi, Y. Chi, A. Dahl, P. De Dobbelaere, S. Denton, S. Gloeckner, K. Yan Hon, S. Hovey, Y. Liang, G. Masini, A. Mekis, M. Peterson, T. Pinguet, S. Sahni, J. Schramm, M. Sharp, C. Sohn, K. Stechschulte, P. Sun, G. Vastola, L. Verslegers and R. Zhou, "Luxtera's Silicon Photonics Platform for Transceiver Manufacturing," 2017, doi: 10.7567/ssdm.2014.b-6-1. accessed on 03/04/2020.

[17] "https://www.cisco.com" accessed on 03/04/2020.

[18] C. V. Poulton, A. Yaacobi, D. B. Cole, M. J. Byrd, M. Raval, D. Vermeulen, and M. R. Watts, "Coherent solid-state LIDAR with silicon photonic optical phased arrays," Opt. Lett., 2017, doi: 10.1364/ol.42.004091.

[19] Y. Kang, Z. Huang, Y. Saado, J. Campbell, A. Pauchard, J. Bowers, and M.J. Paniccia., "High performance Ge/Si avalanche photodiodes development in intel," in Optics InfoBase Conference Papers, 2011, doi: 10.1364/ofc.2011.owz1.

[20] Y. A. Vlasov, "Silicon photonics for next generation computing systems," 2009, doi: 10.1109/ecoc.2008.4729553.

[21] Y. A. Vlasov and S. J. McNab, "Losses in single-mode silicon-on-insulator strip waveguides and bends," Opt. Express, 2004, doi: 10.1364/opex.12.001622.

[22] D. M. Gill, C. Xiong, J. Proesel, J. Rosenberg, J. Ellis-monaghan, J. Orcutt, M. Khater, J. Ellis-Monaghan, A. Stricker, E. Kiewra, Y. Martin, Y. Vlasov, W. Haensch, and W. Green, "Demonstration of error free operation up to 32 Gb/s from a CMOS integrated monolithic nano-photonic transmitter," in CLEO: Science and Innovations, CLEO-SI 2015, 2015, doi: 10.1364/CLEO_SI.2015.STu4F.3.

[23] "http://caliopa.com/" accessed on 03/04/2020.

[24] A. Rickman, "The commercialization of silicon photonics," Nature Photonics. 2014, doi: 10.1038/nphoton.2014.175.

[25] M. Piels and J. E. Bowers, "Photodetectors for silicon photonic integrated circuits," in Photodetectors: Materials, Devices and Applications, 2016.

[26] C. G. Littlejohns, "Silicon - germanium for photonics applications," University of Southampton, 2015.

[27] "https://www-03.ibm.com/press/us/en/pressrelease/7819.wss.".

[28] M. T. Currie, S. B. Samavedam, T. A. Langdo, C. W. Leitz, and E. A. Fitzgerald, "Controlling threading dislocation densities in Ge on Si using graded SiGe layers and chemical-mechanical polishing," Appl. Phys. Lett., 1998, doi: 10.1063/1.121162.

[29] E. Y. Chang, T. H. Yang, G. Luo, and C. Y. Chang, "A GeSi-buffer structure for growth of high-quality GaAs epitaxial layers on a Si substrate," J. Electron. Mater., 2005, doi: 10.1007/s11664-005-0175-5.

[30] Y. Shiraki and N. Usami, Silicon-Germanium (SiGe) Nanostructures. 2011.

[31] L. Mastronardi, M. Banakar, A.Z. Khokhar, N. Hattasan, T. Rutirawut, T. Domínguez Bucio, K. M. Grabska, C. Littlejohns, A. Bazin, G. Mashanovich, and F.Y. Gardes, "High-speed Si/GeSi hetero-structure Electro Absorption Modulator," Opt. Express, 2018, doi: 10.1364/oe.26.006663.

Chapter 2 Theoretical background and literature review

[1] F. Gamiz, N. Rodriguez, C. Navarro, C. Marquez, S. Cristoloveanu, "Functional Nanomaterials and Devices for Electronics, Sensors and Energy Harvesting," 2014.

[2] R. Pillarisetty, "Academic and industry research progress in germanium nanodevices," Nature. 2011, doi: 10.1038/nature10678.

[3] B. V. Van Zeghbroeck, "Chapter 4: p-n Junctions," Princ. Semicond. Devices Heterojunctions, 2010.

[4] J. A. Del Alamo, "Nanometre-scale electronics with III-V compound semiconductors," Nature. 2011, doi: 10.1038/nature10677.

[5] B. Jalali and S. Fathpour, "Silicon photonics," Journal of Lightwave Technology. 2006, doi: 10.1109/JLT.2006.885782.

[6] D. A. B. Miller, "Germanium quantum wells for high-performance modulators in silicon photonics," Photonics Spectra, 2007.

[7] S. R. Das, "Advances in InSb and InAs Nanowire Based Nanoelectronic Field Effect Transistors," in Nanoelectronics, 2019.

[8] H. J. Xiang, B. Huang, E. Kan, S. H. Wei, and X. G. Gong, "Towards direct-gap silicon phases by the inverse band structure design approach," Phys. Rev. Lett., 2013, doi: 10.1103/PhysRevLett.110.118702.

 B. Jalali, V. Raghunathan, R. Shori, S. Fathpour, D. Dimitropoulos, and O. Stafsudd, "Prospects for silicon mid-IR Raman lasers," IEEE J. Sel. Top. Quantum Electron., 2006, doi: 10.1109/JSTQE.2006.885340.

[10] R. Braunstein, A. R. Moore, and F. Herman, "Intrinsic optical absorption in germanium-silicon alloys," Phys. Rev., 1958, doi: 10.1103/PhysRev.109.695.

[11] J. Weber and M. I. Alonso, "Near-band-gap photoluminescence of Si-Ge alloys," Phys. Rev. B, 1989, doi: 10.1103/PhysRevB.40.5683.

[12] Y. Shiraki and N. Usami, Silicon-germanium (SiGe) nanostructures: production, properties and applications in electronics. 2011.

[13] A. P. Knights and K. Doylend J., "Silicon photonics-recent advances in device development," in Advances in Information Optics and Photonics, 2008.

[14] K. Wada and L. C. Kimerling, Photonics and electronics with germanium. 2015.

[15] F. Haj Hassan, "Electronic models," 2015.

[16] Y. M. Haddara, P. Ashburn, and D. M. Bagnall, "Silicon-germanium: properties, growth and applications," in Springer Handbooks, 2017.

[17] C. G. Littlejohns, "Silicon - germanium for photonics applications," University of Southampton, 2015.

[18] P. A. Gabrys, S. E. Seo, M. X. Wang, E. Oh, R. J. Macfarlane, and C. A. Mirkin, "Lattice Mismatch in Crystalline Nanoparticle Thin Films," Nano Lett., 2018, doi: 10.1021/acs.nanolett.7b04737.

[19] R. People and J. C. Bean, "Calculation of critical layer thickness versus lattice mismatch for Ge xSi1-x/Si strained-layer heterostructures," Appl. Phys. Lett., 1985, doi: 10.1063/1.96206.

[20] V. M. Kaganer, B. Jenichen, R. Shayduk, W. Braun, and H. Riechert, "Kinetic optimum of volmer-weber growth," Phys. Rev. Lett., 2009, doi: 10.1103/PhysRevLett.102.016103.

[21] C. Claeys and E. Simoen, "Extended Defects in Germanium: Fundamental and Technological Aspects," Springer Sci. Bus. Media, p. 158, 2008.

[22] A. Baskaran and P. Smereka, "Mechanisms of Stranski-Krastanov growth," J. Appl. Phys., 2012, doi: 10.1063/1.3679068.

[23] D. J. Eaglesham and M. Cerullo, "Dislocation-free Stranski-Krastanow growth of Ge on Si(100)," Phys. Rev. Lett., 1990, doi: 10.1103/PhysRevLett.64.1943.

[24] J. Michel, J. Liu, and L. C. Kimerling, "High-performance Ge-on-Si photodetectors," Nat. Photonics, 2010, doi: 10.1038/nphoton.2010.157.

[25] A. Dobbie, "Investigation of the Electrical Properties of Si1-xGex channel pMOSFETs with High-k Dielectrics," University of Warwick, 2007.

[26] A. Nayfeh, "Heteroepitaxial growth of relaxed germanium on silicon. Ph.D. dissertation," Stanford University, 2006.

[27] J. Liu, M. Beals, A. Pomerene, S. Bernardis, R. Sun, J. Cheng, L. C. Kimerling & J. Michele, "Waveguide-integrated, ultralow-energy GeSi electro-absorption modulators," Nat. Photonics, 2008, doi: 10.1038/nphoton.2008.99.

[28] K. Yu, F. Yang, H. Cong, L. Zhou, Q. Liu, Lichun Zhang, B. Cheng, C. Xue, Y. Zuo, Ch. Li "Fabrication of high-hole-mobility germanium-on-insulator wafers through an easy method," J. Alloys Compd., 2018, doi: 10.1016/j.jallcom.2018.02.178.

[29] M. Cooke, "Germanium and silicon photonics," Compd. Adv. Silicon, 2016.

[30] Y. Huo, H. Lin, R. Chen, Y. Rong, T. I. Kamins, and J. S. Harris, "MBE growth of tensilestrained Ge quantum wells and quantum dots," Front. Optoelectron., 2012, doi: 10.1007/s12200-012-0193-x.

[31] S. Luryi, A. Kastalsky, and J. C. Bean, "New infrared Detector on a Silicon Chip," IEEE Trans. Electron Devices, 1984, doi: 10.1109/T-ED.1984.21676.

 [32] D. M. Isaacson, C. L. Dohrman, and E. A. Fitzgerald, "Deviations from ideal nucleationlimited relaxation in high-Ge content compositionally graded SiGe/Si," J. Vac. Sci. Technol.
 B Microelectron. Nanom. Struct., 2006, doi: 10.1116/1.2366584.

[33] J. L. Liu, S. Tong, Y. H. Luo, J. Wan, and K. L. Wang, "Ge films grown on Si substrates by molecular-beam epitaxy below 450°C," J. Appl. Phys., 2004, doi: 10.1063/1.1738530.

[34] P. R. Bandaru, S. Sahni, E. Yablonovitch, J. Liu, H.-J. Kim, and Y.-H. Xie, "Fabrication and characterization of low temperature," Mater. Sci. Eng. B, vol. 113, no. 1, pp. 79–84, Oct. 2004, doi: 10.1016/J.MSEB.2004.07.007.

[35] P. Capper, S. Irvine, and T. Joyce, "Epitaxial Crystal Growth: Methods and Materials," in Springer Handbook of Electronic and Photonic Materials, 2006.

[36] H. Nelson, "Epitaxial growth from the liquid state and its application to the fabrication of tunnel and laser diodes," RCA Rev., vol. 24, no. 4, pp. 603–615, 1963.

[37] H. H. Wieder, A. R. Clawson, and G. E. McWilliams, "InxGa1-xAsyP1-y/InP heterojunction photodiodes," Appl. Phys. Lett., vol. 31, pp. 468–470, 1977.

[38] R. H. Saul and D. D. Roccasecca, "Vapor-doped multislice LPE for efficient GaP green LED's," J. Electrochem. Soc., 1973, doi: 10.1149/1.2403644.

[39] M. B. Panish, I. Hayashi, and S. Sumski, "Double-heterostructure injection lasers with room-temperature thresholds as low as 2300 A/cm2," Appl. Phys. Lett., 1970, doi: 10.1063/1.1653213.

[40] Y. Liu, M. D. Deal, and J. D. Plummer, "High-quality single-crystal Ge on insulator by liquid-phase epitaxy on Si substrates," Appl. Phys. Lett., 2004, doi: 10.1063/1.1691175.

[41] D. Feng, W. Qian, H. Liang, Ch. Kung, Z. Zhou, Z. Li, J. Levy, R. Shafiiha, J. Fong, B. Luff,
 M. Asghari Khiavi, "High-speed GeSi electroabsorption modulator on the SOI waveguide platform," IEEE J. Sel. Top. Quantum Electron., 2013, doi: 10.1109/JSTQE.2013.2278881.

[42] X. Bai, C. Y. Chen, P. B. Griffin, and J. D. Plummer, "Si incorporation from the seed into Ge stripes crystallized using rapid melt growth," Appl. Phys. Lett., 2014, doi: 10.1063/1.4863976.

[43] Y. Tojo, R. Matsumura, H. Yokoyama, M. Kurosawa, K. Toko, T. Sadoh, M. Miyao "High-quality formation of multiply stacked SiGe-on-insulator structures by temperaturemodulated successive rapid-melting-growth," Appl. Phys. Lett., 2013, doi: 10.1063/1.4794409.

[44] T. A. Langdo, C. W. Leitz, M. T. Currie, E. A. Fitzgerald, A. Lochtefeld, and D. A. Antoniadis, "High quality Ge on Si by epitaxial necking," Appl. Phys. Lett., 2000, doi: 10.1063/1.126754.

[45] C. G. Littlejohns, T. Dominguez Bucio, M. Nedeljkovic, H. Wang, G. Z. Mashanovich,
G. T. Reed and F. Y. Gardes, "Towards a fully functional integrated photonic-electronic platform via a single SiGe growth step," Sci. Rep., 2016, doi: 10.1038/srep19425.

[46] M. Miyao, T. Tanaka, K. Toko, and M. Tanaka, "Giant Ge-on-insulator formation by Si-Ge mixing-triggered liquid-phase epitaxy," Appl. Phys. Express, 2009, doi: 10.1143/APEX.2.045503.

[47] M. Miyao, K. Toko, T. Tanaka, and T. Sadoh, "High-quality single-crystal Ge stripes on quartz substrate by rapid-melting-growth," Appl. Phys. Lett., 2009, doi: 10.1063/1.3182795.

[48] T. Hashimoto, C. Yoshimoto, T. Hosoi, T. Shimura., and H. Watanabe, "Fabrication of local Ge-on-insulator structures by lateral liquid-phase epitaxy: Effect of controlling
interface energy between Ge and insulators on lateral epitaxial growth," Appl. Phys. Express, 2009, doi: 10.1143/APEX.2.066502.

[49] H. Y. S. Koh, "Rapid melt growth of silicon germanium for heterogeneous integration on silicon," Stanford University, 2011.

[50] R. Matsumura, Y. Tojo, M. Kurosawa, T. Sadoh, I. Mizushima, and M. Miyao, "Growthrate-dependent laterally graded SiGe profiles on insulator by cooling-rate controlled rapidmelting-growth," Appl. Phys. Lett., 2012, doi: 10.1063/1.4769998.

[51] H. Von Känel, D. Chrastina, B. Rössner, G. Isella, J. P. Hague, and M. Bollani, "High mobility SiGe heterostructures fabricated by low-energy plasma-enhanced chemical vapor deposition," in Microelectronic Engineering, 2004, doi: 10.1016/j.mee.2004.07.029.

[52] C. V. Falub, H. von Känel, F. Isa, R. Bergamaschini, A. Marzegalli, D. Chrastina, G. Isella,
E. Müller, P. Niedermann, L. Miglio, "Scaling hetero-epitaxy from layers to threedimensional crystals," Science (80)., 2012, doi: 10.1126/science.1217666.

Chapter 3 Methods

[1] "https://nanolab.berkeley.edu" accessed on 03/04/2020.

J. Weber, M. Hiller, and E. V. Lavrov, "Hydrogen in germanium," Mater. Sci. Semicond.
 Process., vol. 9, no. 4–5, pp. 564–570, Aug. 2006, doi: 10.1016/J.MSSP.2006.08.007.

[3] K. Wada and L. C. Kimerling, Photonics and electronics with germanium. 2015.

[4] C. G. Littlejohns, "Silicon - germanium for photonics applications," University of Southampton, 2015.

[5] C. G. Littlejohns, A. Khokhar, D.J. Thomson, Y. Hu, L. Basset, S. Reynolds, G. Mashanovich, G. Reed and F. Gardes, "Ge-on-Si Plasma-Enhanced Chemical Vapor Deposition for Low-Cost Photodetectors," IEEE Photonics J., 2015, doi: 10.1109/JPHOT.2015.2456069.

[6] S. Mukhopadhyay, A. Chowdhury, and S. Ray, "Substrate temperature dependence of microcrystalline silicon growth by PECVD technique," J. Non. Cryst. Solids, 2006, doi: 10.1016/j.jnoncrysol.2005.11.095.

[7] E. Gusev, J. Jityaeva, S. Avdeev, and O. Ageev, "Effect of substrate temperature on the properties of plasma deposited silicon oxide thin films," J. Phys. Conf. Ser., vol. 1124, p. 22034, Dec. 2018, doi: 10.1088/1742-6596/1124/2/022034.

[8] R. Amrani, P. Abboud, L. Chahed, and Y. Cuminal, "Low Temperature Growth of Hydrogenated Silicon Prepared by PECVD from Argon Diluted Silane Plasma," Cryst. Struct. Theory Appl., 2012, doi: 10.4236/csta.2012.13012.

[9] "http://www.plasma-therm.com" accessed on 03/04/2020.

[10] "Iness.como.polimi.it/" accessed on 03/04/2020.

[11] G. Isella, D.Chrastina, B.Rössner, T.Hackbarth, H.J. Herzog, U.König, H.von Känel, "Low-energy plasma-enhanced chemical vapor deposition for strained Si and Ge heterostructures and devices," Solid. State. Electron., 2004, doi: 10.1016/j.sse.2004.01.013.

[12] S. Marchionna, A. Virtuani, M. Acciarri, G. Isella, and H. von Kaenel, "Defect imaging of SiGe strain relaxed buffers grown by LEPECVD," Mater. Sci. Semicond. Process., 2006, doi: 10.1016/j.mssp.2006.09.003.

[13] H. Von Känel, D. Chrastina, B. Rössner, G. Isella, J. P. Hague, and M. Bollani, "High mobility SiGe heterostructures fabricated by low-energy plasma-enhanced chemical vapor deposition," in Microelectronic Engineering, 2004, doi: 10.1016/j.mee.2004.07.029.

[14] R. Bergamaschini et al., "Self-aligned Ge and SiGe three-dimensional epitaxy on dense

Si pillar arrays," Surface Science Reports. 2013, doi: 10.1016/j.surfrep.2013.10.002.

[15] S. Franssila, "Introduction to Microfabrication," 2010.

[16] S. Platias, K. I. Vatalis, and G. Charalampides, "Suitability of Quartz Sands for Different Industrial Applications," Procedia Econ. Financ., vol. 14, pp. 491–498, Jan. 2014, doi: 10.1016/S2212-5671(14)00738-2.

[17] "http://nanosiliconinc.com/thermal-oxide/" accessed on 03/04/2020.

[18] M. Totzeck, W. Ulrich, A. Göhnermeier, and W. Kaiser, "Semiconductor fabrication: Pushing deep ultraviolet lithography to its limits," Nat. Photonics, 2007, doi: 10.1038/nphoton.2007.218.

[19] H. H. Radamson, E. Simoen, J. Luo, C. Zhao, "Scaling and evolution of device architecture," C. Past, Present Futur., pp. 19–40, Jan. 2018, doi: 10.1016/B978-0-08-102139-2.00002-1.

 Y. Vladimirsky, A. Bourdillon, O. Vladimirsky, W. Jiang, and Q. Leonard, "Demagnification in proximity X-ray lithography and extensibility to 25 nm by optimizing Fresnel diffraction," J. Phys. D. Appl. Phys., 1999, doi: 10.1088/0022-3727/32/22/102.

[21] "www.compugraphics-photomasks.com" accessed on 03/04/2020.

[22] "https://www.microchemicals.com/technical_information/photoresist_baking.pdf."

[23] G. C. Tesoro, "Polymers for microelectronics resists and dielectrics, L. F. Thompson,
C. G. Wilson and S. Tagawa, eds. ACS Symposium Series 537, Washington, DC (1994),"
Polym. Adv. Technol., 1995, doi: 10.1002/pat.1995.220060318.

[24] "https://www.southampton-nanofab.com/fabrication/opticalLithography.php" accessed on 03/04/2020.

[25] "https://www.asml.com/en/technology" accessed on 03/04/2020.

[26] "https://cmi.epfl.ch" accessed on 03/04/2020.

[27] "https://www.nikon.com/products/semi/history/index.htm" accessed on 03/04/2020.

[28] C. Vieu, F.Carcenac, A.Pépin, Y.Chen, M.Mejias, A.Lebib, L.Manin-Ferlazzo, L.Couraud,
H.Launois, "Electron beam lithography: Resolution limits and applications," Appl. Surf. Sci.,
2000, doi: 10.1016/S0169-4332(00)00352-4.

[29] "www.jeol.co.jp" accessed on 03/04/2020.

[30] S. Prakash, J. Yeom, S. Prakash, and J. Yeom, "Advanced Fabrication Methods and Techniques," Nanofluidics Microfluid., pp. 87–170, Jan. 2014, doi: 10.1016/B978-1-4377-4469-9.00004-4.

[31] S. Prakash, J. Yeom, S. Prakash, and J. Yeom, "Advanced Fabrication Methods and Techniques," Nanofluidics Microfluid., pp. 87–170, Jan. 2014, doi: 10.1016/B978-1-4377-4469-9.00004-4.

[32] S. Yasin, D. G. Hasko, and H. Ahmed, "Fabrication of <5 nm width lines in poly(methylmethacrylate) resist using a water:isopropyl alcohol developer and ultrasonically-assisted development," Appl. Phys. Lett., 2001, doi: 10.1063/1.1369615.

[33] Y. Chen, "Nanofabrication by electron beam lithography and its applications: A review," Microelectronic Engineering. 2015, doi: 10.1016/j.mee.2015.02.042.

[34] "Manipulation and Patterning of Surfaces (Nanolithography)," Fundam. Appl. Nano
Silicon Plasmon. Fullerines, pp. 89–137, Jan. 2018, doi: 10.1016/B978-0-323-480574.00005-0.

[35] Y. Xia, J. A. Rogers, K. E. Paul, and G. M. Whitesides, "Unconventional Methods for Fabricating and Patterning Nanostructures," Chem. Rev., 1999, doi: 10.1021/cr980002q.
[36] S. Hönl, H. Hahn, Y. Baumgartner, L. Czornomaz, and P. Seidler, "Highly selective dry etching of GaP in the presence of AlxGa1-xP with a SiCl4/SF6 plasma," J. Phys. D. Appl. Phys., 2018, doi: 10.1088/1361-6463/aab8b7.

[37] D. Burt, A. Al-Attili, Z. Li, F. Gardès, M. Sotto, N. Higashitarumizu, Y. Ishikawa, K. Oda,
O. M. Querin, S. Saito, and R.Kelsall, "Enhanced light emission from improved homogeneity in biaxially suspended Germanium membranes from curvature optimization," Opt. Express, 2017, doi: 10.1364/oe.25.022911.

[38] J. T. L. Thong, W. K. Choi, and C. W. Chong, "TMAH etching of silicon and the interaction of etching parameters," Sensors Actuators, A Phys., 1997, doi: 10.1016/S0924-4247(97)80511-0.

[39] "www.memsnet.org" accessed on 03/04/2020.

[40] A. B. Kahng and K. Samadi, "CMP fill synthesis: A survey of recent studies," IEEE Trans.Comput. Des. Integr. Circuits Syst., 2008, doi: 10.1109/TCAD.2007.907061.

[41] J. M. Steigerwald, S. P. Murarka, and R. J. Gutmann, "Chemical Mechanical Planarization of Microelectronic Materials," 1997.

[42] "www.fujifilmusa.com cmp" accessed on 03/04/2020.

[43] D. Zhao and X. Lu, "Chemical mechanical polishing: Theory and experiment," Friction, 2013, doi: 10.1007/s40544-013-0035-x.

[44] G. Zwicker, "Applications of chemical mechanical planarization (CMP) to More than Moore devices," Adv. Chem. Mech. Planarization, pp. 463–485, Jan. 2016, doi: 10.1016/B978-0-08-100165-3.00018-8.

[45] Y. M. Sub, B. Y. T. Hian, L. I. Fong, P. M. Anak, A.B. Minhar, T.K. Wui, M. P. T. Kim, L.
H. Jin, F. T. Min, "The study on the effect of pattern density distribution on the STI CMP process," 2017, p. 030023, doi: 10.1063/1.4998394.

[46] C. Shan Xu, J. Liu, and Y. Xia, "Quantitative study of chemical mechanical planarization process affected by bare silicon wafer front surface topography," J. Vac. Sci. Technol. B Microelectron. Nanom. Struct., vol. 17, no. 5, p. 2210, 1999, doi: 10.1116/1.590895.

[47] A. Gupta, S. Noyel Victoria, and R. Manivannan, "Chemical Mechanical Planarization of Germanium using Polymorphs of Titania as Abrasive and Hydrogen Peroxide," Mater. Today Proc., vol. 18, pp. 861–868, Jan. 2019, doi: 10.1016/J.MATPR.2019.06.515.

[48] S. Peddeti, P. Ong, L. H. A. Leunissen, and S. V. Babu, "Chemical mechanical polishing of Ge using colloidal silica particles and H2O2," Electrochem. Solid-State Lett., 2011, doi: 10.1149/1.3575166.

[49] Y. Cai, W. Yu, L. C. Kimerling, and J. Michel, "Chemical Mechanical Polishing of Selective Epitaxial Grown Germanium on Silicon," ECS J. Solid State Sci. Technol., vol. 3, no. 2, pp. P5–P9, Nov. 2014, doi: 10.1149/2.003401jss.

[50] M. T. Currie, S. B. Samavedam, T. A. Langdo, C. W. Leitz, and E. A. Fitzgerald, "Controlling threading dislocation densities in Ge on Si using graded SiGe layers and chemical-mechanical polishing," Appl. Phys. Lett., 1998, doi: 10.1063/1.121162.

[51] "https://www.thermocoupleinfo.com/" accessed on 03/04/2020.

[52] X. Zhang, K.-S. Chen, and S. Mark Spearing, "Thermo-mechanical behaviour of thick PECVD oxide films for power MEMS applications," Sensors Actuators A Phys., vol. 103, no. 1–2, pp. 263–270, Jan. 2003, doi: 10.1016/S0924-4247(02)00343-6.

[53] "https://www.jawoollam.com" accessed on 03/04/2020.

[54] A. Bandyopadhyay and S. Bose, Characterization of Biomaterials. 2013.

[55] N. John and S. George, "Raman Spectroscopy," Spectrosc. Methods Nanomater.
 Charact., pp. 95–127, Jan. 2017, doi: 10.1016/B978-0-323-46140-5.00005-4.

[56] "https://www.horiba.com/en_en/raman-imaging-and-spectroscopy/" accessed on 03/04/2020.

[57] A. A. Bunaciu, V. D. Hoang, and H. Y. Aboul-Enein, "Vibrational Micro-Spectroscopy of Human Tissues Analysis: Review," Critical Reviews in Analytical Chemistry. 2017, doi: 10.1080/10408347.2016.1253454.

[58] J. S. Lupoi, E. Gjersing, and M. F. Davis, "Evaluating lignocellulosic biomass, its derivatives, and downstream products with Raman spectroscopy," Frontiers in Bioengineering and Biotechnology. 2015, doi: 10.3389/fbioe.2015.00050.

[59] E. D. Palik, "Handbook of optical constants of solids," 2012.

[60] P. M. Mooney, F. H. Dacol, J. C. Tsang, and J. O. Chu, "Raman scattering analysis of relaxed Ge_xSi_{1-x} alloy layers," Appl. Phys. Lett., 1993, doi: 10.1063/1.109481.

[61] H. Zhou, Z. Luo, and X. Li, "Using focus ion beam to prepare crystal lamella for electron diffraction," J. Struct. Biol., 2019, doi: 10.1016/j.jsb.2019.02.004.

[62] U. Villanueva-Amadoz, A. Benedetti, J. Méndez, L. M. Sender, and J. B. Diez, "Focused ion beam nano-sectioning and imaging: A new method in characterisation of palaeopalynological remains," Grana, 2012, doi: 10.1080/00173134.2011.641579.

[63] "www.sites.ualberta.ca" accessed on 03/04/2020.

Chapter 4 Design of mask and experiments

[1] J. Soole, Y. Silberberg, A. Scherer, H. P. LeBlanc, N. C. Andreadakis, C. Caneau, H. Schumacher and U. Erben, "Fast high-efficiency integrated waveguide photodetectors using novel hybrid vertical/butt coupling geometry," Appl. Phys. Lett., vol. 61, pp. 13–15, Aug. 1992, doi: 10.1063/1.107645.

[2] C. G. Littlejohns, "Silicon - germanium for photonics applications," University of Southampton, 2015.

[3] D. M.-M. D. V. T. J. Brouckaert, L. Vivien, "Report on electrical and optical integration designs of Ge and InGaAs photodetectors."

[4] "https://www.silvaco.com" accessed on 03/04/2020.

[5] C. G. Littlejohns, T. Dominguez Bucio, M. Nedeljkovic, H. Wang, G. Z. Mashanovich,
 G. T. Reed and F. Y. Gardes., "Towards a fully functional integrated photonic-electronic platform via a single SiGe growth step," Sci. Rep., 2016, doi: 10.1038/srep19425.

[6] D. Feng, S. Liao, H. Liang, J. Fong, B. Bijlani, R. Shafiiha, B. J. Luff, Y. Luo, J. Cunningham, A. V. Krishnamoorthy and M. Asghari, "High-speed GeSi electroabsorption modulator on the SOI waveguide platform," IEEE J. Sel. Top. Quantum Electron., 2013, doi: 10.1109/JSTQE.2013.2278881.

[7] L. Mastronardi, M. Banakar, A.Z. Khokhar, N. Hattasan, T. Rutirawut, T. Domínguez Bucio, K. M. Grabska, C. Littlejohns, A. Bazin, G. Mashanovich and F.Y. Gardes, "Highspeed Si/GeSi hetero-structure Electro Absorption Modulator," Opt. Express, 2018, doi: 10.1364/oe.26.006663.

[8] C. G. Littlejohns, M. Nedeljkovic, C. F. Mallinson, J. F. Watts, G. Z. Mashanovich, G. T. Reed and F. Y. Gardes, "Next Generation Device Grade Silicon-Germanium on Insulator," Sci. Rep., 2015, doi: 10.1038/srep08288.

Chapter 5 Fabrication of SiGe wires on SOI platform

[1] Y. Cai, W. Yu, L. C. Kimerling, and J. Michel, "Chemical Mechanical Polishing of Selective Epitaxial Grown Germanium on Silicon," ECS J. Solid State Sci. Technol., vol. 3, no. 2, pp. P5–P9, Nov. 2014, doi: 10.1149/2.003401jss.

[2] J. T. L. Thong, W. K. Choi and C.W. Chong, "TMAH etching of silicon and the interaction of etching parameters. Sensors and Actuators" A: Physical, 63(3), 243-249, 1997.

[3] Y. Cai, "Materials Science and Design for Germanium Monolithic Light Source on Silicon," Massachusetts Institute of Technology, 2014.

 Y. Cai, W. Yu, L. C. Kimerling and J. Michel, "Chemical Mechanical Polishing of Selective Epitaxial Grown Germanium on Silicon," ECS J. Solid State Sci. Technol. 2014 volume 3, issue 2, P5-P9 2014.

[5] S. Peddeti, P. Ong, L.H.A. Leunissen, S.V. Babu, "Chemical mechanical planarization of germanium shallow trench isolation structures using silica-based dispersions," Microelectronic Engineering Volume 93, 2012.

[6] S. Babu, "Advances in Chemical Mechanical Planarization (CMP)," Woodhead Publishing, 2016.

[7] E. lannone, "Labs on Chip: Principles, Design and Technology," CRC Press 2014.

[8] B. G. Ko, H. C. Yoo, J. G. Park and U. Paik, "Effects of pattern density on CMP removal rate and uniformity," Journal of the Korean Physical Society, 2001. 39: p. S318-S321.

[9] B.-H. Kwon, J.-H. Lee, H.- J. Kim, S. S. Kweon, Y-G. Ryu, J.- G. Lee, "Dishing and erosion in STI CMP, in VLSI and CAD", ICVC '99, 6th International Conference, 1999.

[10] S. Dhumal, S.K., A Theoretical and Experimental Study of Stresses Responsible for the SOI Wafer Warpage. ECS Trans., 2008.

[11] X. Bai, C. Y. Chen, P. B. Griffin, and J. D. Plummer, "Si incorporation from the seed into Ge stripes crystallized using rapid melt growth," Appl. Phys. Lett., 2014, doi: 10.1063/1.4863976.

[12] C. G. Littlejohns, "Silicon - germanium for photonics applications," University of Southampton, 2015.

[13] C. G. Littlejohns, M. Nedeljkovic, C. F. Mallinson, J. F. Watts, G. Z. Mashanovich, G. T. Reed and F. Y. Gardes, "Next Generation Device Grade Silicon-Germanium on Insulator," Sci. Rep., 2015, doi: 10.1038/srep08288.

[14] L. Mastronardi, M. Banakar, A.Z. Khokhar, N. Hattasan, T. Rutirawut, T. Domínguez Bucio, K. M. Grabska, C. Littlejohns, A. Bazin, G. Mashanovich and F.Y. Gardes, "High-speed Si/GeSi hetero-structure Electro Absorption Modulator," Opt. Express, 2018, doi: 10.1364/oe.26.006663.

[15] C. G. Littlejohns, T. Dominguez Bucio, M. Nedeljkovic, H. Wang, G. Z. Mashanovich,
G. T. Reed and F. Y. Gardes, "Towards a fully functional integrated photonic-electronic platform via a single SiGe growth step," Sci. Rep., 2016, doi: 10.1038/srep19425.

Chapter 6 Conclusions and Future Work

[1] L. Mastronardi, M. Banakar, A.Z. Khokhar, N. Hattasan, T. Rutirawut, T. Domínguez Bucio, K. M. Grabska, C. Littlejohns, A. Bazin, G. Mashanovich and F.Y. Gardes, "High-speed Si/GeSi hetero-structure Electro Absorption Modulator," Opt. Express, 2018, doi: 10.1364/oe.26.006663.

C. G. Littlejohns, T. Dominguez Bucio, M. Nedeljkovic, H. Wang, G. Z. Mashanovich,
 G. T. Reed and F. Y. Gardes, "Towards a fully functional integrated photonic-electronic platform via a single SiGe growth step," Sci. Rep., 2016, doi: 10.1038/srep19425.

[3] J. Fujikata, M. Noguchi, Y. Kim, S. Takahashi, T. Nakamura, and M. Takenaka, "High speed and highly efficient Si optical modulator with strained SiGe layer," in IEEE International Conference on Group IV Photonics GFP, 2015, doi: 10.1109/Group4.2015.7305930.

[4] Y. Kim, M. Takenaka, T. Osada, M. Hata, and S. Takagi, "Strain-induced enhancement of plasma dispersion effect and free-carrier absorption in SiGe optical modulators," Sci. Rep., 2014, doi: 10.1038/srep04683.

[5] C. G. Littlejohns A. Khokhar, D.J. Thomson, Y. Hu, L. Basset, S. Reynolds, G. Mashanovich, G. Reed and F. Gardes, "Ge-on-Si Plasma-Enhanced Chemical Vapor Deposition for Low-Cost Photodetectors," IEEE Photonics J., 2015, doi: 10.1109/JPHOT.2015.245606